Development of Molecularly Imprinted Polymers for Corticosteroids

By Laurence Fitzhenry



A dissertation submitted to WIT for the degree of Doctor of Philosophy

October 2011

Based on research carried out under the supervision of

Dr. Patrick Duggan and Dr. Peter McLoughlin

Pharmaceutical and Molecular Biotechnology Research Centre Department of Chemical and Life Sciences Waterford Institute of Technology Waterford Ireland

Declaration

I hereby declare that this thesis is my own work, in fulfilment of the requirements of Doctor of Philosophy degree. It is based on research carried out in the Department of Chemical and Life Sciences, Waterford Institute of Technology, Ireland.

Laurence Fitzhenry

Date

INTO MY OWN

One of my wishes is that those dark trees, So old and firm they scarcely show the breeze, Were not, as 'twere, the merest mask of gloom, But stretched away unto the edge of doom.

I should not be withheld but that some day Into their vastness I should steal away, Fearless of ever finding open land, Or highway where the slow wheel pours the sand.

I do not see why I should e'er turn back, Or those should not set forth upon my track To overtake me, who should miss me here And long to know if still I held them dear.

They would not find me changed from him they knew— Only more sure of all I thought was true.

Robert Frost

For My Children

Acknowledgements

I would like to thank my supervisors, Pat and Peter, for all of their help and support over the last four years. I am grateful to have had the freedom to try the things I wanted to try, and to have had their experience and ability to help me get it right when it went wrong. I feel privileged to have learned from you both.

I would like to extend my sincere thanks to Panagiotis for his input into the project and for his help with all things scientific. This would not have been the thesis it is without your help, and for that thank you.

I would like to thank Jimmy for his help with the NMR studies and the technicians for helping me with my many requests for assistance and materials. I must also extend a big thank you to everyone that I have worked with over the last fours years, in the PMBRC and beyond. Thanks for all of the tea, the poker, the hill walks, the film and book clubs, the conversation and most importantly, the laughter. I wish you all the best in the future. I would like to say a particular thanks to Mike who over the last years has provided not just help and advice but also friendship.

Thank you to all of my friends, and a special thank you to my family. Tabatha, thank you for everything you have given to me over the last nine years. I feel lucky to have spent this part of the journey together. Most importantly, to my three amazing children, Rebecca, Adam and Thomas, thank you, for all that you are and for all that you have given to me.

Abstract

The aim of this research was to prepare a series of molecularly imprinted polymers (MIPs) for corticosteroid templates using bulk and precipitation polymerisation strategies. Initially, a systematic study was undertaken, investigating the factors affecting the synthesis of MIPs by precipitation polymerisation. Careful optimisation of polymerisation parameters, including the monomer concentration, initiator type and solvent, allowed for a high degree of control over particle size and morphology. Pre-equilibrium interactions between the primary target budesonide (BDN) and functional monomers were investigated using NMR studies and showed weak interactions with dissociation constants, k_{diss} , ranging from 0.534-1.47 M

Due to this weak functional monomer-template interaction, polymers were prepared in bulk format using a target analogue approach utilising hydrocortisone-17-butyrate (HCB) as a low-cost alternative to BDN. The HCB imprinted ground bulk polymers demonstrated imprinting factors (*IF*) as high as 7.5 for the template. Selectivity studies on six steroidal analogues showed that recognition was driven by shape and size selectivity as well as hydrophobic interactions, with *IF* values ranging from 1.2-9.5 in both organic and aqueous based systems.

Applications of the imprinted polymers were demonstrated by solid-phase extraction (SPE) and analytical column chromatography. SPE studies utilising HCB imprinted polymers displayed a seven-fold increase in the specific retention of BDN. Precipitated prednisolone imprinted polymers achieved chromatographic separation between the template and hydrocortisone, a close structural analogue. Selectivity factors as high as 1.4 and a resolution value of 0.75 were achieved.

Two custom functional monomer strategies were utilised towards enhancing MIP performance. The synthesis of tailor-made monomers based on derivatised cyclodextrins (CD) achieved a substitution of 2.5 polymerisable vinyl groups per CD. This derivatised CD was used to prepare precipitated polymers (0.3-7.2 μ m) in a 40% aqueous acetonitrile porogen. A polymerisable hydrocortisone methacrylate monomer was successfully synthesised for use in semi-covalent imprinting which produced discrete microspherical beads (1.2-5.1 μ m) using divinylbenzene as the crosslinker.

List of abbreviations

MIP: Molecularly imprinted polymer NIP: Non-imprinted polymer MAA: Methacrylic acid HEMA: 2-(hydroxyethyl)methacrylate EGDMA: Ethylene glycol dimethacrylate DVB: Divinylbenzene TRIM: Trimethylolpropane trimethacrylate AIBN: Azobisisobutyronitrile ACCN: 1,1'-azobis(cyclohexanecarbonitrile) CVA: 4,4'-azobis(4-cyanovaleric acid) HCB: hydrocortisone-17-butyrate **BDN:** budesonide PRD: prednisolone HYD: hydrocortisone HAC: hydrocortisone-21-acetate PAC: prednisolone-21-acetate AcOD: acetic acid- d_6 CD: cyclodextrin α -CD: α -cyclodextrin β-CD: β-cyclodextrin γ -CD: γ -cyclodextrin MAC: methacryloyl chloride NMA: N-(hydroxymethyl) acrylamide CDM: cyclodextrin methacrylate β -CDA: acrylamidomethyl β -cyclodextrin γ -CDA: acrylamidomethyl γ -cyclodextrin CDMP: cyclodextrin molecularly imprinted polymer HMAMP: hydrocortisone methacrylate molecularly imprinted polymer

Table of Contents

1.	Introduction	1
	1.1 The History of Molecular Imprinting	2
	1.2 Principle of molecular imprinting	4
	1.2.1 Polymers	
	1.2.2. Molecular recognition	6
	1.2.3 Molecular imprinting	
	1.3 Methodologies for the creation of imprinted sites	
	1.3.1 Covalent imprinting	
	1.3.2 Non-covalent imprinting	
	1.3.3 Semi-covalent imprinting	
	1.3.4 Stoichiometric non-covalent imprinting	
	1.4 Optimising the performance of molecularly imprinted polymers	
	1.4.1 The functional monomer	
	1.4.2 The role of the diluent	
	1.4.3 The crosslinking monomer	
	1.4.4 Polymer initiation	
	1.4.5 A rational design approach to molecular imprinting	
	1.5 Methods of polymerisation	
	1.5.1 Bulk polymerisation	
	1.5.2 Molecularly imprinted beads	
	1.5.3 Film and lens type MIPs	
	1.5.4 Advantages and disadvantages of selected polymerisation strate	gies 50
	1.6 Characterisation of molecularly imprinted polymers	51
	1.6.1 Chemical characterisation	
	1.6.2 Morphological characterisation	
	1.6.3 Binding site characterisation	
	1.7 Applications of molecularly imprinted polymers	
	1.7.1 Solid phase extraction	

1.7.2 MIPs in catalysis	
1.7.3 MIPs in drug delivery	58
1.8 Scope of the thesis	62
References	63
2. Towards a budesonide specific molecularly imprinted polymer	
2.1 Introduction	
2.1.1 Steroids	80
2.1.2 Steroids in molecular imprinting	
2.1.3 Strategies for the synthesis of steroid imprinted polymers	
2.1.4 Precipitation polymerisation	86
2.2 Research aims and objectives	
2.3 Experimental	90
2.3.1 Materials	
2.3.2 Instrumentation	91
2.3.3 Preparation of polymers	91
2.3.4 Scanning electron and optical microscopy	94
2.3.5 Equlibrium binding studies	94
2.3.6 Particle size analysis	
2.3.7 NMR analysis	96
2.4 Results and discussion	97
2.4.1 Initial polymer formulation	97
2.4.2 Systematic synthesis and testing of polymers M1-M7	106
2.4.3 Factors affecting polymerisation	109
2.4.4 NMR Spectroscopy	117
2.4.5 Initial studies towards the use of a template analogue approach	
2.5 Conclusions	
References	
3. The synthesis of a molecularly imprinted polymer using the tar	get analogue
approach	

	3.1 Introduction	142
	3.1.1 Target analogue imprinting	
	3.1.2 MIP Chromatography	
	3.1.3 Nitrogen sorption porosimetry	
	3.2 Research aims and objectives	148
	3.3 Experimental	149
	3.3.1 Materials	
	3.3.2 Instrumentation	
	3.3.3 Preparation of polymers	
	3.3.4 Chromatographic polymer evaluation	151
	3.3.5 Nitrogen sorption porosimetry	153
	3.3.6 Particle size and scanning electron microscopy analysis	153
	3.4 Results and discussion	154
	3.4.1 Target analogue selection	154
	3.4.2 Preparation and analysis of HCB imprinted polymers	
	3.4.3 Polymers prepared by bulk methods	
	3.4.4 Polymers prepared by precipitation	177
	3.5 Conclusion	187
	References	188
4.	4. Applications of corticosteroid imprinted polymers	
	4.1 Introduction	194
	4.2 Research aims and objectives	196
	4.3 Experimental	196
	4.3.1 Materials	
	4.3.2 Molecularly imprinted solid phase extraction	
	4.3.3 Preparation and evaluation of prednisolone imprinted polymers	(M25-M27)
		197
	4.4 Results and discussion	198
	4.4.1 Molecularly imprinted solid phase extraction of budesonide	

4.4.2 MIP chromatography for the separation of closely related steroidal
compounds
4.5 Conclusions
References
5. The synthesis of tailor-made functional monomers for corticosteroid molecular
imprinting
5.1 Introduction
5.2 Research aims and objectives
5.3 Experimental
5.3.1 Materials
5.3.2 Instrumentation
5.3.3 Monomer synthesis
5.3.4 Polymer synthesis
5.3.5 NMR
5.3.6 LC-MS
5.3.7 SEM and nitrogen sorption porosimetry
5.3.8 Equilibrium binding studies
5.3.9 Column packing and chromatographic evaluation
5.4 Results and discussion
5.4.1 Cyclodextrin compounds as complex-forming agents for molecular
imprinting
5.4.2 The synthesis of a polymerisable template-monomer for use in semi-covalent
corticosteroid MIPs
5.5 Conclusion
References
6. Conclusions and Future work

Chapter 1

Introduction

1. Introduction

The field of molecular imprinting in synthetic polymers is one of the most exciting areas in chemistry and material science today. The imprinting of polymers on the molecular level may be described as the formation of sites or cavities within the polymer that are receptive to a specific molecule or group of related molecules. The formation of these cavities is achieved by the synthesis of the polymer in the presence of a specific molecule. The molecule, around which the polymer is synthesised, is known as the template and the subsequent removal of the template, after polymer formation, results in the creation of the selective sites. These sites are then available for rebinding of the template based on shape, size and perhaps most importantly, functionality, all combining to promote a high degree of molecular recognition. The polymer thus formed is known as a molecularly imprinted polymer (MIP).

1.1 The History of Molecular Imprinting

The history and development of the field of molecularly imprinted polymers has been well documented in a number of excellent and comprehensive reviews [1, 2]. With the aid of such reviews and the MIP database [3] it is possible to present a brief account of the most notable stages in this, by now, rapidly growing field.

It is now generally accepted [1, 4] that the first recorded instance of molecular imprinting was carried out in the 1930s by the Soviet chemist M.V. Polyakov who reported interesting adsorption properties while investigating silica gel matrices [5]. In this work benzene, toluene or xylene was added to the silica matrix prior to drying. Adsorption studies were carried out on dried silica, from which the additives had been removed by extensive washing. These studies showed selective adsorption for the additive present during the formation of the gel. Despite the fact that Polyakov published a number of other papers [6, 7] in that decade his work appeared to go largely unnoticed within the wider scientific community [1].

It was not until the work of a graduate student of Linus Pauling, Frank Dickey, in the late forties, detailing the then termed, 'specific adsorption' of methyl orange dye in silica gels, that this area of research received more attention [8]. Dickey stated that the

investigation itself was not so much a quantitative one as a qualitative one and he attributed the adsorption to the formation of "pockets that fit closely enough to the foreign molecule to hold it in by Van der Waal's forces, hydrogen bonds, ionic attraction and other types of intermolecular interaction". Dickey published a more detailed paper six years later investigating such topics as the amount of attractive surface, the formation of the cavities and problems surrounding what in contemporary scientific literature is now termed the 'un-extracted template' [9]. Around that time there were a number of other works published correlating the work of Dickey as well as an early use of 'specific adsorption' for the separation of stereoisomers in the work of Curti and Colombo [10-12]. It is worth noting that many of the uses investigated and perhaps more importantly, the problems encountered, are much the same as the uses and problems being investigated today.

An article by Morrison *et al.* published in 1959, saw a challenge to Dickey's suggestions for the cause of selective adsorption, with the likely cause ascribed to association between un-extracted dye and dye present in the rebinding solution [13]. This fact notwithstanding, the 1960's saw a continuation in the research of "stereoselective adsorbents" where the range of applications began to become more diverse, with papers outlining their use in, for example, the 'imprinting' of bacteria to investigate the optical activity of receptors in dissymmetric organisms by Patrikeev *et al.* and early work on their use as drug receptor and enzyme active site models by Beckett and Youssef [14, 15].

It is generally stated, that it was the advent of molecular imprinting in organic polymers by Wulff and Sarhan in the seventies, and later the group of Arshady and Mosbach in the eighties [16, 17] that ushered in, as one commentator has termed it: "the new era of molecular imprinting" [18] and has led to the abundance of literature pertaining to molecular imprinting that is present in contemporary scientific research. Before discussion of this wealth of literature, however, it is necessary to first discuss the underlying principles of molecular imprinting.

1.2 Principle of molecular imprinting

Molecular imprinting has two major scientific areas at its core: molecular recognition and polymer chemistry. While molecular recognition often takes precedence in the design of imprinted polymers, the fact that this recognition must take place in a synthetic polymer can not be overlooked. With this in mind it is considered necessary to give a brief overview of the basics of polymer chemistry.

1.2.1 Polymers

Polymer molecules are large molecules with high molecular weights made of simple repeating units [19], known as monomers. The number of monomeric units per polymer is the degree of polymerisation [20]. Polymers prepared from a single monomer type are known as homopolymers, while copolymer is the term used for those prepared with more than one monomer. The incorporation of these different monomers into the polymeric matrix can occur in a number of ways [19]. Alternating copolymers occur when the monomeric units alternate in a linear fashion, random copolymers, as the name suggests, involve randomly incorporated monomers, while block copolymers involve blocks of one monomer connected to blocks of the other monomer. Some examples of monomers and their corresponding polymers can be seen in Figure 1.1 [21].



Figure 1.1: I Methacrylic acid and its corresponding polymer, polymethacrylic acid; II ethylene glycol and its corresponding polymer, polyethylene glycol [21].

Polymers possess characteristic chain structures and the monomers of which they are composed are related chemically *via* a primary bond, usually covalent in nature [21]. When these chemically bonded monomers form polymers possessing no side chains the

result is termed a *linear polymer*. Where side chains are present a *branched polymer* is formed while *network polymers* possess a three-dimensional structure arising from the linking together of polymer chains in a process known as *crosslinking* [19, 21]. The crosslinking of polymer networks can effect major changes in the physical properties of polymers. Crosslinked polymers, for example, despite swelling considerably in appropriate solvents, are not soluble in any solvents [22].

Whether polymers are prepared as linear, branched or networked polymers, two general mechanisms are responsible: Step growth (condensation) and chain growth (addition) polymerisations [23]. These were first classified by Carothers in 1929, using the terms condensation and addition, and this theory was expanded by Flory in the 1930s [24, 25].

1.2.1.1 Step Growth Polymerisations (Condensation).

Step growth polymers are generally formed by a reaction between two different functional groups with the loss of some small molecule, such as water, in the course of the formation of the new bond. Examples of polymers formed by condensation reactions are nylon and polyester. Different types of condensation reactions include:

- Activated polycondensations
- Polyadditions
- Branching polyadditions

1.2.1.2 Chain growth polymerisations (addition).

Chain growth polymerisations are usually formed by the addition of one monomer unit to another in a repetitive manner. Here the addition of the monomer molecule will result in transference of the reactive site to the new chain end [23]. Addition polymerisation is a straight or linear chain process, which continues until the process is terminated in some way. The typical steps in the reaction are as follows:

- Initiation
- Propagation
- Termination

These polymerisation reactions can be initiated by a variety of methods and the resultant polymerisation reaction types are named in accordance with the mechanism of initiation. These include [19, 20]:

- Anionic polymerisations: In anionic polymerisation the propagating chain is a carbanion.
- Cationic polymerisations: The propagating chain is a carbocation in cationic polymerisations.
- Free radical polymerisations: Free radical polymerisation involves the addition of monomers to a free-radical bearing reactive site.

As the majority of molecular imprinting reactions are free radical initiated polymeric reactions, this topic will be discussed in more detail in Section 1.4.4.

1.2.2. Molecular recognition

Molecular recognition may be described as a set of phenomena under the control of specific non-covalent interactions [26]. These interactions, and the subsequent molecular level 'recognition' events, are often necessary in order to perform or induce specific functions or effects. Nobel laureate, J.M. Lehn states that binding alone is not recognition but that recognition can be considered to be "binding with a purpose" [27].

1.2.2.1 Molecular recognition in biological systems

These molecular recognition processes are of paramount importance in nature and thus, life itself. This can be seen in DNA where molecular recognition is responsible not only for the structure of DNA molecules but also in ligand-DNA or protein-DNA interactions [28]. The formation of base pairs occurs by means of specific hydrogen bonds, for example the adenine:thymine base pair has two hydrogen bonds while the guanine:cytosine base pair has three. This is represented in Figure 1.2.



Figure 1.2: I Adenine:thymine base pair and II guanine:cytosine base pairs showing hydrogen bonding

In double-helical DNA it is the occurrence of these hydrogen bonds between bases on separate strands that hold the double helix together. For ligand-DNA recognition, such as DNA-drug recognition, the incoming ligand must have a hydrogen bonding (H-bonding) capability complementary to the H-bonding potential of the base pairs, which remains after base pairing itself takes place. There are a number of categories of DNA-binding proteins, a particular example being regulatory proteins that generally bind to highly specific regions of DNA and control the processing of a gene [28].

The specificity of enzymes is well documented and the stereochemical specificity of the active site is effected by the presence of at least three different points of interaction between enzyme and substrate [29]. Binding sites interact with specific groups on the substrate which determines the specific orientation, in which the substrate and enzyme are held with respect to each other. The binding site is three-dimensional and bonds formed are generally non-covalent. In 1894 Emil Fischer suggested that enzyme specificity was a result of sites bearing functionality complementary to that of the substrate and the substrate would fit into the active site in a manner analogous to a key and a lock [30]. While this lock-and-key model was later determined to be insufficient to account for the flexibility of proteins it is often cited as the inspiration for synthetic replication of biological specificity.

Antibodies too, are proteins that exhibit a high degree of specificity. In this case, the proteins are antigen-specific immunoglobins (antibodies) that are part of the immune system's response to attack by an antigen [31]. The aim of this response is the specific elimination of the antigen utilising the antigen-specific antibodies. These antibodies

have affinity and binding complementarities for the antigen. The response is polyclonal, where a mixture of antibodies for the antigen is made and upon specific binding to the antigen, it is destroyed. Again, these interactions are based mainly on non-covalent interactions in three-dimensional domains where non-polar interactions can also play a role [31].

1.2.2.2 Molecular recognition within synthetic materials

Given the high degree of specificity involved in the molecular recognition that takes place in biological systems, coupled with the myriad functions of this recognition, it is little wonder that the incorporation of these properties into synthetic materials has been the focus of much scientific and technological research. This research has developed into the wide-ranging areas of supramolecular and host-guest chemistry.

Early work was carried out in this field by the groups of Nobel Laureates: Pedersen, Lehn and Cram. The crown ether was first observed 'by chance' by Pedersen in 1967 as a side product due to an impurity in one of the starting materials [32-34]. Once the metal-ion binding properties were observed a wide variety of compounds were developed, crown ethers being followed by cryptands, initially developed by Lehn. These have chemical bridges and also involved N atoms, giving a three-dimensional cavity [35] and spherands, developed by Cram where the donor centres point to the interior of a rigid ring [36].

These synthetic molecules have found many uses in synthetic chemistry where ions or ionic intermediates are involved. The synthesis can often be modified or improved [33] and they may also be used in chemical analysis or selective sensing. Examples of these are shown in Figure 1.3.



Figure 1.3: I Crown ether of the type developed by Pederson with K^+ ion complexed within the molecule; II cryptand of the type developed by Lehn; III spherand of the type developed by Cram.

Binding between target molecules and compounds of the type seen in Figure 1.3, as with those discussed in biological systems, is typically based on "simultaneous non-covalent interactions between single binding sites" [37]. The presence of a number of binding sites is necessitated by the inherently weak nature of non-covalent interactions and recognition is achieved by concurrent interaction between the individual binding sites. Lehn [27] states a number of factors necessary to induce the large difference in binding energies between those of a given substrate and other substrates that are required to achieve the desired level of recognition and they include the following:

- Steric complementarity between substrate and site
- Interactional complementarity within three-dimensional sites
- Large contact areas
- Multiple interaction sites
- Strong overall binding
- Solvent effects

With these considerations firmly in mind, plus the advantages of readily synthesised artificial receptors, the appeal and potential for utilising polymeric material in which these receptor sites could be created and maintained is readily seen.

1.2.3 Molecular imprinting

The process of molecular imprinting is often represented in a schematic fashion similar to that shown in Figure 1.4.



Figure 1.4: Schematic representation of the imprinting process involving the following steps: (1) interaction of the template (blue) and functional monomers to form a template-monomer complex; (2) Reaction of template-monomer complex and crosslinking monomer to form a crosslinked polymer containing a template specific binding site; (3) Removal of the template; (4) Rebinding of the template.

It must be stressed that Figure 1.4 is a highly schematic representation of the molecular imprinting process and does not adequately represent the three-dimensional nature of the site, the cavity nor the polymeric matrix. Although highly schematic, Figure 1.4 readily allows for a description of the fundamental steps that are considered necessary for molecular imprinting, as outlined in the following sections.

1.2.3.1 Formation of the template-monomer complex

The template molecule and the functional monomer are added to a solvent (or porogen) in conditions and at ratios that are designed to promote interaction between the two species. This interaction leads to the formation of the template-monomer complex and

may proceed via a number of processes, e.g. covalent or non-covalent, that will be discussed in more detail in Section 1.3.

1.2.3.2 Polymerisation

Polymerisation is carried out in the porogen, selection of which is carried out to facilitate the template-functional monomer interaction. This polymerisation of the functional monomers involved in the template-monomer complex is now carried out in order to 'lock' the complex within a porous, polymeric material. This is achieved by the addition of crosslinking monomers, which contain at least two polymerisable moieties. An organic initiator is typically used, which causes the polymerisation reaction to proceed. Upon completion, the result should be a rigid, porous polymer, containing sites in which the template-monomer complex is locked within the matrix of the polymer. The rigidity of the polymer is achieved using a high degree of crosslinking monomer, typically between 10 and 80 mol% [38]. Appropriate selection of the crosslinking monomer and the porogen imparts adequate porosity to the polymer, thus allowing the analyte access to and from the cavities [39].

1.2.3.3 Extraction of the template

This involves breaking the bonds between the functional moieties of the polymer and the template with subsequent removal of the template from the polymeric material. How this is achieved is dependent on the type of bond formed in the template-monomer complex. For example, if covalent bonds were formed then a step such as hydrolysis would be necessary, whereas if non-covalent bonds were formed then disruption of the bonds could be brought about by washing in an acidified or basified solvent. Whatever the method, the desired result is the creation of a site within the polymeric matrix that is complementary in shape, size and functionality to the template molecule. This is an important step in the preparation of molecularly imprinted polymers, as un-extracted template can result in a reduction of the number of binding sites available for rebinding or errors in analytical measurements due to template bleed [40].

1.2.3.4 Rebinding of the template

At this stage, it should be possible to rebind the template with the polymer using the empty imprinted sites or cavities. This is typically achieved with the greatest success under conditions similar to those used for the formation of the polymer, i.e. solvent, etc. [41, 42].

The overall success of molecularly imprinted polymers is, to varying degrees, dependent on how each of the four steps listed above is designed and implemented and a greater understanding of the design of molecularly imprinted polymers, requires a deeper understanding of the factors involved and the extent to which they affect this process. The following sections will attempt to outline some of these factors and discuss how they may be best adapted for the myriad tasks to which MIPs may be applied.

1.3 Methodologies for the creation of imprinted sites

The interaction between the functional monomer and the template is responsible for the preparation of the template-monomer complex and thus is ultimately responsible for the complementary functionality of the binding site in the prepared polymer. This interaction is therefore of paramount importance to the performance of the resultant polymer and proper selection of the type of interaction must be tailored to suit not only the template but also the final application of the polymer. While there is an ever growing number of processes by which the functional or binding site monomer may interact with the template molecule, these can be separated into two main strategies: covalent and non-covalent. These strategies and a number of their variants will be discussed in the following sections.

1.3.1 Covalent imprinting

The covalent imprinting strategy involves a reversible covalent linkage between the template and the functional monomer. The template-monomer complex is then incorporated into the polymer matrix and requires chemical treatment, such as hydrolysis, to break the chemical bond between the template and the functional moieties

of the binding site monomers. Upon rebinding, the chemical bond is then re-formed between the monomers and the template.

As mentioned in Section 1.1 the first such imprinted organic polymers were prepared by Wulff and Sarhan [16] and were utilised in the resolution of D,L-glyceric acid. The template-monomer complex in this instance was a synthesised derivative of glyceric acid, a 2,3-*O*-*p*-vinylphenylboronic ester of *p*-vinylanilide. This method was also used for D-mannitol [43]. The polymers were synthesised using divinylbenzene (DVB) as the crosslinking monomer. The template, template-monomer complex and diagrammatic representation of the binding pocket for glyceric acid are shown in Figure 1.5.



Figure 1.5: Schematic representation of the imprinting protocol for D-glyceric acid developed by Wulff and Sarhan whereby a covalently bonded template-monomer complex was used to effect selectivity for D- over L-glyceric acid. (Adapted from references [16] and [43])

Figure 1.5 demonstrates how the template-monomer complex is incorporated into the resultant polymer and how upon hydrolysis, the template is removed and the functional groups are maintained in the desired spatial arrangement for rebinding of D-glyceric acid. As such, the binding moieties would not be as suitably aligned for L-glyceric acid recognition. It was possible to remove 50% of the template and the polymer had the capacity to rebind this amount. In subsequent racemate resolution the polymer

preferentially bound D-glyceric acid allowing for enrichment of the L-glyceric form in solution. The α -value, the ratio of the distribution between the solution and the two forms of glyceric acid, was determined to be 1.034. Similar values were also obtained with this polymer for structural analogues of glyceric acid.

This boronate ester approach has often been utilised for the imprinting of carbohydrates, with recent use for the preparation of electrochemical sensors for monosaccharides by Willner *et al.* [44]. Here, 3-hydroxyphenylboronic acid was used to form a template-monomer complex incorporated into an electropolymerised polyphenol matrix. Ferrocene functionalised derivatives of D-mannose and D-glucose were prepared in order to calculate association constants, $K_{ass.}$, and values of 870 ± 40 M⁻¹ and 170 ± 30 M⁻¹ were found for D-mannose and D-glucose, respectively and enantioselectivity was also observed in these polymers.

More recently, Schumacher *et al.* have used a polymerisable benzoboroxole to improve the complexation of fructose at neutral pH [45]. Here two template-monomers were synthesised for fructose using both 5-methacrylamido-2-hydroxymethylphenylboronic acid and 3-vinylphenylboronic acid. At a basic pH of 11.4, where the high concentration of hydroxide ions would promote the complexation of fructose, the binding capacities of the polymers prepared from both boronic acids were comparable, attaining values of 135.6 nmol g⁻¹ for the benzoboroxole and 114.7 nmol g⁻¹ for the boronic acid. However, at pH 7.4, where the structure of the boron monomer would be expected to play a more significant role, the polymer prepared from the benzoboroxole had a three-fold increase in binding capacity over the polymers containing pendant vinylphenylboronic acid moieties, with values of 60 nmol g⁻¹ and 18.2 nmol g⁻¹ for the former and the latter, respectively.

Schiff's bases have also been used for the imprinting of amino acids by Wulff and Vietmeier [46] while ketal and acetal formation between diols and carbonyl compounds have been employed for imprinting of aromatic ketones by Shea and Dougherty [47] during investigations into the influence of functional group positioning on molecular recognition. Covalent imprinting has been applied to a number of different templates, such as bisphenol A, widely used in the production of epoxy resins [48], the tricyclic

antidepressant, nortryptiline [49] and steroids such as cholesterol [50] and castasterone [51].

1.3.1.1 Advantages and disadvantages of covalent imprinting

Despite the ability to prepare binding sites using the exact stoichiometric ratio of functional monomer to template, thereby reducing non-specific binding, as well as the increased stability of the template-monomer complex during polymerisation, a factor that lends itself to increased homogeneity of the binding sites, a number of problems arise in the case of covalent molecular imprinting. These problems include the relatively low number of templates and corresponding functional monomers that can be used to synthesise appropriate template-monomer complexes. Template removal has also been cited as a problem [2] as well as the slow kinetics of template release or rebinding, which would make this approach less desirable for applications requiring fast binding kinetics, such as chromatography. The impact of some of these negative factors can often be overcome, or at the very least decreased, by the use of non-covalent or semi-covalent binding strategies.

1.3.2 Non-covalent imprinting

The non-covalent imprinting strategy involves the self-assembly of functional monomers about the template molecule by way of non-covalent interactions. This self-assembly may be driven by forces similar to those in nature such as hydrogen bonds, hydrophobic interaction, the formation of ion-pairs, dipole-dipole forces or π - π interactions. While the template-monomer complexes, or adducts, formed by non-covalent interactions are not as inherently stable as those formed by covalent linkages, the range of possible interactions (and the monomers that can utilise them) allows for greater choice of template. This factor, plus the reduced need for synthetic chemistry when commercially available functional monomers are employed, has led to non-covalent imprinting becoming the most widely used strategy in molecular imprinting today [1, 2, 52].

The non-covalent strategy was pioneered by Mosbach and Arshady who prepared the then termed 'substrate-selective polymers' for two organic dyes, rhodanile blue and safranine O [17]. These dyes and a number of the monomers used to provide complementary functionality are shown in Figure 1.6.



Figure 1.6: Templates and monomers used by Mosbach and Arshady during initial non-covalent imprinting studies: I methyl methacrylate; II *N*,*N*'-methylenediacrylamide; III rhodanile blue; IV safranine O.

The polymers prepared from the templates and monomers shown in Figure 1.6, were prepared in H₂O:DMF mixtures at 40 °C using ammonium persulfate as the initiator. The rebinding and selectivity capabilities of the resultant polymers were investigated by comparison to polymers prepared without the presence of the templates, which is now commonly referred to as a non-imprinted polymer or 'NIP'. The binding capacity was measured to be as high as 72 μ mol/g of polymer with specific binding as high as 100%.

Since the advent of this early work in the 1980s, involving the generation of such promising results, the range of templates, the number of functional monomers and the applications for non-covalent molecular imprinting has grown exponentially. Some template examples include the widely used pharmaceutical actives propranolol, morphine and benzodiazepines [53, 54], proteins such as myoglobin [55], herbicides and fungicides [56, 57], chemical warfare agents [58], amino acids [59, 60] and steroids [42, 61-64], as demonstrated by the work of Ramstrom *et al.* for which the imprinting strategy is depicted in Figure 1.7.



Figure 1.7: Schematic for the non-covalent imprinting of hydrocortisone using methacrylic acid (MAA) as functional monomer and ethylene glycol dimethacrylate (EGDMA) as the crosslinker. Adapted from reference [63].

Figure 1.7 illustrates the non-covalent approach used by Ramström *et al.* for the preparation of polymers imprinted with hydrocortisone. At 1, the functional monomers are shown assembling about the template at the proposed sites of interaction, based on their hydrogen bonding capabilities. Once this complex has been given sufficient time to form, the crosslinking monomer and initiator were added and the polymer was formed, as in 2. The template was extracted and upon rebinding, the non-covalent interactions employed to form the initial template-monomer complex should then act to rebind the template within the template specific cavities, 3.

In this study the NIP bound only 10-16% of the quantity of template that was successfully bound to the MIP. The cross-reactivity or selectivity of the polymers for the template over close structural analogues was shown to be comparable to monoclonal antibodies used for corticosteroids with values as low as 4% in the MIP, where the corresponding antibody value was 0.1%. There were, however, a number of problems cited, such as the heterogeneity of the binding sites. Two types of binding sites were observed: high affinity sites with a dissociation constant, $K_{diss.}$, of $0.57 \pm 0.16 \times 10^{-6}$ M and a binding site density of $0.21 \pm 0.05 \mu \text{mol g}^{-1}$ and low affinity binding sites with $K_{diss.}$ of $1.59 \pm 0.73 \times 10^{-3}$ M and a binding site density of $280 \pm 120 \mu \text{mol g}^{-1}$. This demonstrates that the quantity of low affinity sites was considerably larger than high affinity sites. This heterogeneity of the binding sites is one of a number of problems encountered within non-covalent imprinting.

Despite these problems, the ability to non-covalently interact with the template in a biomimetic fashion is an attractive attribute of this method. An example of this approach was carried out by Jiang *et al.* on the separation of racemic ketoprofen [65]. Here a tailor-made, chiral monomer was prepared from cinchonine and used as both functional and crosslinking monomer. The monomer was designed to have a proposed three-point interaction with the chiral target, (*S*)-ketoprofen, employing H-bonding, ionic, and π - π interactions. The use of a number of binding interaction types such as these, demonstrates the biomimetic nature of the molecularly imprinting approach. When the resultant polymer was used as packing material for a HPLC column, (*R*)- and (*S*)-ketoprofen were separated with a resolution of 1.25.

1.3.2.1 Advantages and disadvantages of non-covalent imprinting

The advantages of the non-covalent strategies are many and include the myriad templates that may be imprinted, the vast array of functional monomers that may be employed and the fast rebinding kinetics possible. The potential to use a wide range of solvents and a number of initiation methods as well as the low risk of damage to the template during the polymerisation have also been cited as advantages of the non-covalent method [66]. Despite these obvious advantages, there are also a number of disadvantages inherent to this strategy, including, as stated in the preceding section, a lack of homogeneity of the binding sites. This is due in part because of the need to use

an excess of functional monomer to drive template-monomer complex formation as the interactions between monomer and template are generally weak [67]. This leads to a relatively large number of complexes that may be formed between the template and the functional monomers and can result in a relatively low yield of high-affinity receptor sites in comparison to low-affinity sites [2]. For these reasons, a number of strategies have been developed in order to combine the most desirable properties of both covalent and non-covalent methodologies.

1.3.3 Semi-covalent imprinting

Semi-covalent molecular imprinting attempts to couple the homogenous binding sites created by covalent imprinting with the fast rebinding kinetics achieved using non-covalent methods by utilising covalent linkages to create the binding site but employing non-covalent interactions to effect rebinding. Interestingly, the first covalent imprinting protocol, discussed earlier, involved additional rebinding capability due to electrostatic interaction (Figure 1.5) [16].

The first truly semi-covalent approach has been cited [1] as that used by Andersson and Sellergren, for the imprinting of *p*-aminophenylalanine ethyl ester [68]. In this instance, a structural analogue that had two covalently attached polymerisable groups containing ester linkages was used. Upon completion of polymerisation and removal of the imprint molecule two carboxylic acid groups remained in the binding sites, each having the capacity to rebind the amino acid via non-covalent interactions. This method is exemplified by Cacho *et al.* during the preparation of an MIP prepared for the herbicide, propazine, by semi-covalent methods [69]. An outline of the procedure used is given in Figure 1.8.



Free binding site Binding site containing template-monomer complex Figure 1.8: Schematic for the procedure used by Cacho *et al.* for the semi-covalent imprinting of propazine. 1: propazine; 2: Methacryloyl chloride; 3: propazine methacrylate.

Figure 1.8 depicts the procedure used for the semi-covalent imprinting of propazine. The template (1) was reacted with methacryloyl chloride (2) in dichloromethane, to prepare the vinyl amide template-monomer (3). After polymerisation, basic hydrolysis was used to remove the template, leaving the binding site free to non-covalently rebind propazine.

The method shown in Figure 1.8 was compared to earlier work carried out by Cacho *et al.* where propazine was imprinted using a non-covalent strategy [70]. It is interesting to note that the semi-covalent method produced a polymer containing 52% more binding sites (though this was still considerably lower than that expected by such methods) and binding isotherms revealed a heterogeneity index almost 20% higher than that seen in the non-covalent approach. However, the binding constant obtained for the semi-covalent polymer using isotherm data, at 151.7 mM⁻¹, was considerably lower than that observed for the non-covalent method, which had a value of 1484 mM⁻¹. This was attributed to the high affinity binding sites prepared in non-covalent polymers due to the greater molar ratio of monomer to template compared to the 1:1 stoichiometry used for this method.

In 2009, Curcio *et al.* applied the semi-covalent method in conjunction with one-stage mini-emulsion polymerisation to prepare molecularly imprinted polymers for glucopyranoside [71]. This method employed a polymerisable surfactant template to enhance the surface imprinting effect and improve binding kinetics. This was achieved using a one-step mini-emulsion polymerisation and resulted in a polymer that was designed to have four carboxylic acid residues per binding site after hydrolytic removal of the template from the polymer, as illustrated in Figure 1.9.



Figure 1.9: I The polymerisable surfactant template monomer used for the preparation of II, the surface imprinted polymer bearing a proposed four carboxylic acid residues per binding site [72].

As well as herbicides and amino acids, the semi-covalent approach has also been used for the imprinting of resin production intermediates such as bisphenol-A [73], testosterone [74] and 4-nitrophenol [75]. Interestingly, for bisphenol-A and 4nitrophenol, comparisons were made to non-covalent polymers prepared in tandem and these outperformed the semi-covalent polymers in terms of selectivity. A variation of this method has been developed, which also uses covalently bound template-monomer complexes and rebinds the target *via* non-covalent means, and is known as the sacrificial spacer method.

1.3.3.1 The sacrificial spacer method

The sacrificial spacer method was introduced by Whitcombe *et al.* as a method to effectively imprint cholesterol, a molecule considered to have poor functionality for imprinting by non-covalent means [76]. Here the template used was a 4-vinylphenyl carbonate ester of cholesterol and upon removal of the template the 'sacrificial spacer', CO_2 was lost and a recognition site bearing a phenolic residue remained, as shown in Figure 1.10.



Figure 1.10: Sacrificial spacer method of imprinting for cholesterol. 1: Polymerisation of template monomer, cholesteryl (4-vinyl) phenyl carbonate to form a polymer containing the template; 2: Removal of template leaving phenolic residue in the binding site; 3: Rebinding of template via non-covalent interaction. Adapted from [76].

After polymerisation of the template-monomer with EGDMA, a polymer is formed containing the template-monomer within the binding sites. Upon removal of the 'sacrificial' CO_2 group by hydrolysis, the site is free to non-covalently bind the target. The loss of this spacer reduces problems caused by steric crowding during the non-covalent rebinding step. The binding affinities of these polymers were studied using Scatchard analysis, a method for quantifying the binding interaction between ligand and

receptor, and revealed affinities as high as $K_{diss} = 0.59 \pm 0.12$ mM at a capacity of 114 ± 6 µmol/g [76].

Cholesterol has also been the target for chitin-based molecularly imprinted polymers prepared using the semi-covalent method [77]. Here, Tong *et al.* prepared cholesteryl chitin carbonate, a novel chitin derivative, and cross-linked the polymer using toluene-2,4-diisocyanate. This resulted in a polymer, which was able to bind up to 30% more cholesterol than either the non-imprinted polymer or an imprinted polymer prepared *via* non-covalent means.

Other templates imprinted in this manner have included *N*-heterocycles such as pyridine, where the size selectivity of the imprinted sites was greater than for the conventional non-covalent approach [78], testosterone [79], methyl paraben [80] and paracetamol [81].

1.3.4 Stoichiometric non-covalent imprinting

Stoichiometric non-covalent imprinting involves the use of monomers that have such high affinity for the template species that the vast majority of the functional monomer molecules interact with the template prior to, and during polymerisation [2, 67]. This situation therefore negates the use of excess functional monomer to drive association, thereby reducing the occurrence of low affinity binding sites and increasing in strength and number, the high affinity sites. To achieve this, the association constants, K_{ass} , between the functional monomer and the template must be in the order of 10^3 M^{-1} . To date, this has been achieved using specially designed functional monomers, a strategy pioneered by Wulff *et al.*, which utilises such monomers as shown in Figure 1.11



Figure 1.11: I Amidinium monomers synthesised for use in stoichiometric non-covalent imprinting and II typical template for which these monomers have been used [82, 83].

Figure 1.11 above presents an example of the type of monomers used in stoichiometric non-covalent imprinting. This type of monomer was first used to imprint a phosphonic monoester transition state analogue [82] but has since been used to prepare molecularly imprinted artificial models of cholesterol esterase [84] and carboxypeptidase A [83]. In the case of the latter example catalytic enhancement values, k_{cat}/k_{uncat} , as high as 410 000 were reported. Pasetto *et al.* have also utilised the stoichiometric non-covalent approach for the preparation of molecularly imprinted nanogels for catalysis of a carbonate hydrolysis reaction [85]. Here, apparent affinity constants between the phosphate template and the polymerisable arginine monomer were cited to range as high as 10⁴ to 10⁵ M⁻¹.

As can be deduced from the preceding examples, the stoichiometric non-covalent strategy for molecular imprinting is particularly well suited to catalytic applications, though it has also been used for polymers designed for use in the solid-phase extraction of the alkaloid huperzine A [86]. The advantages of this method have been ascribed to the combined attributes of non-covalent and covalent imprinting, i.e. the high splitting yields, fast rebinding kinetics and precise location of the functional groups.

1.4 Optimising the performance of molecularly imprinted polymers

One of the most attractive features of molecular imprinting, and non-covalent molecular imprinting in particular, is the potential to adapt this technology to a wide range of target analytes. Such templates may vary greatly from each other in terms of polarity, functionality, solubility, availability, toxicity and potency, etc. As well as the wide range of templates, the applications and requirements for the resultant polymer may also vary greatly, with respect to factors such as the environment in which it will be used and the desired lifetime of the polymer. With such a degree of variation in both templates and polymerisation strategies, it is obvious that to achieve the greatest degree of success, it is necessary to optimise the composition of the polymers as well as their method of synthesis for each application.

1.4.1 The functional monomer

This brief introduction to the methodologies used for site creation within MIPs shows that the functional monomer is largely responsible for the formation of template specific binding sites within the polymer matrix, as well as subsequent rebinding of the template within these cavities. In both instances this is due to the presence of functional moieties on the monomer, which are complementary to the functionality of the target analyte in conjunction with the subsequent promotion of interaction between these functionalities. For this reason careful consideration of the functional monomer is of paramount importance when setting about the task of developing and optimising an MIP.

Karim *et al.* state that over 4000 polymerisable compounds with potential use as functional monomers have been reported in the literature [87]. Despite this, however, a relatively small number are regularly utilised for imprinting purposes. While, as discussed in section 1.3, imprinting strategies other than conventional non-covalent methods require tailor-made or custom functional monomers, the vast majority of functional monomers put to use are commercially available. These monomers can be broken down into the following general groups: acidic, basic, neutral and electrostatically charged [2, 88]. An example of each type of monomer and some applications are presented in Table 1.1 below.

Table 1.1: Examples	of functional monomer st	ructure and applications	of the different ty	pes of monomer.
		The second		F

Monomer	Structure	Functionality	Applications	Reference
Methacrylic acid (MAA)	HOHO	Acidic	17β-estradiol, Aspirin, melamine, clozapine	[89-92]
4-vinyl pyridine (4-VP)	N	Basic	Chlormequat, 17β- estradiol, quercitin, fancy red	[93-96]
2-Hydroxyethyl methacrylate (HEMA)	H ₃ C O OH	Neutral	Cholesterol, hydrocortisone, glycyrrhizic acid	[97-99]
2-acrylamido-2- methylpropane sulfonic acid (AMPSA)	O H SO ₃ H	Charged	Bovine haemoglobin, creatinine	[100, 101]
While the monomers presented in Table 1.1 represent a mere fraction of those that are regularly used for the synthesis of MIPs, they serve to show the variety of possible monomers that exist. The examples in Table 1.1 also serve to show that different types of monomers may be used for the same template, as shown by the reported use of both the acidic MAA and the basic 4-vinyl pyridine (4-VP) for the steroid 17β -estradiol by Gadzala-Kopciuch *et al.* and Watabe *et al.* [90, 93]. It is worth noting that Gadzala-Kopciuch *et al.* studied the use of MAA and 4-VP but in this particular example MAA based MIPs had improved binding ability over the 4-VP based MIPs. This was attributed to increased hydrophilicity of the polymers containing MAA. This example serves to demonstrate that for a particular template, careful choice will have to be made regarding the required functionality of the monomer.

Researchers have often found it beneficial to use a combination of two or more monomers, such as the use of the neutral 2-hydroxyethyl methacrylate (HEMA) combined with the basic dimethylaminoethyl methacrylate for the nitroimidazole drug, (*S*)-ornidazole [102]. Here, a monolithic polymer was prepared for use in capillary electrophoresis and was used to separate the template from its enantiomer and a range of structural analogues. Bovine haemoglobin has also been imprinted using a combination of two functional monomers [103]. In this work, Janiak *et al.* observed that the use of the negatively charged 2-acrylamido-2-methylpropane sulfonic acid combined with the positively charged 3-methacrylamidopropyl trimethylammonium chloride gave increased specificity due to the improved swelling properties brought about by the combination of monomers but decreased selectivity due to the presence of the negative charge, which interacted strongly with a competing protein.

When selecting a functional monomer, as well as selecting the appropriate functionality, it is also necessary to determine the possibility of monomer-monomer self-association and to match the reactivity ratios of the functional monomer with either a second functional monomer or a crosslinking monomer or both, to ensure that each monomer will be incorporated into the polymer matrix to the desired extent [88]. In order to promote the interactions utilised by non-covalent molecular imprinting, e.g. H-bonding, the formation of ion-pairs, dipole-dipole forces or π - π interactions, it is necessary to select polymerisation conditions conducive to such interactions, whether in the prepolymerisation complex or in subsequent rebinding of the template. In both instances, the role of the solvent must not be underestimated.

1.4.2 The role of the diluent

The role of the diluent, or solvent, is central to the synthesis, subsequent characteristics and performance of molecularly imprinted polymers. It is the solvent that brings all of the components, i.e. the monomers, the template and the initiator, into a single phase [88]. In addition to bringing the polymer components together at the pre-polymerisation stage it also plays a fundamental role in promoting interaction between the functional monomer and the template [104]. The solvent is also responsible for dictating the degree of porosity by affecting the timing of phase separation [88, 104, 105] and by occupying space growing polymer matrix. For these reasons, the solvent is commonly referred to as the porogen.

1.4.2.1 Solvent promotion of interaction between the template and functional monomer

It is generally considered that to impart maximum affinity in the resultant polymer, it is necessary to promote maximum interaction between the template and the functional monomer in both the pre-polymerisation complex and the rebinding solution [106]. These are interactions that the solvent may promote or interfere with. Of the non-covalent interactions that are commonly used for imprinting, H-bonding is considered to be one of the most important in non-aqueous solutions [107]. In order to promote H-bonding it is therefore necessary to select a less polar solvent, such as chloroform, which will facilitate such interactions rather than a polar, or worse, a protic polar solvent, which can cause maximum disruption and compete with H-bonding interactions [104].

This was shown in a comprehensive study by Sellergren and Shea which investigated, amongst other factors, the importance of the H-bonding capability of the solvent on subsequent rebinding of the template L-phenylalanine anilide, depicted in Figure 1.12 [108].



Figure 1.12: L-phenylalanine anilide, the template molecule used by Sellergren and Shea for the investigation on the effect of solvent on rebinding [108].

This work, carried out using the model template L-phenylalanine anilide showed, that in going from a solvent with very strong H-bonding capability, such as acetic acid, to one with weak H-bonding capability, such as benzene, the enantioseparation factor in subsequent chromatographic experiments increased from 1.9 to 6.8.

This was further corroborated by the work of Nomachi *et al.* where a pseudo-template, N-propionyl-5-methoxytryptamine was used to prepare polymers imprinted for melatonin in chloroform (CHCl₃), acetonitrile (MeCN) and tetrahydrofuran (THF) [109]. The chemical structures of the template and pseudo-template used in this study are given in Figure 1.13.



Figure 1.13: I target molecule, melatonin and II, the pseudo-template, *N*-propionyl-5methoxytryptamine used by Nomachi *et al.* during investigations on the effect of solvent on molecular imprinting [109].

Nomachi *et al.*, using NMR spectroscopy, found that the H-bonding capability of the solvent was the most important factor in determining the degree of affinity in the final polymer. They found that in CHCl₃, the solvent of the three that showed the lowest degree of H-bonding with the template, the MIP proved to have the highest degree of selectivity over the corresponding non-imprinted polymer. It is worth noting that when aqueous solutions are required for rebinding, it is hydrophobic interactions that will promote rebinding, such as in the work of Puoci *et al.* where cholesterol, a relatively hydrophobic molecule, was rebound to the greatest extent on the most hydrophobic polymer in MeCN:H₂O mixtures [98].

1.4.2.2 The solvent and physical characteristics of MIPs

Another important contribution made by the solvent relates to the morphology of the resulting polymer, which ultimately affects how well the target molecule can access the binding sites within the polymer matrix [88, 104, 105]. The diluent achieves this by affecting the timing of the phase separation based on whether it is a thermodynamically good solvent or a thermodynamically poor solvent for the growing polymer [110]. The use of a good solvent, typically MeCN, where phase separation occurs later, leads to the formation of well developed pore structures and high surface areas while the use of a poor solvent, typically CHCl₃, where phase separation starts earlier, leads to the opposite [104].

While the use of a pore-forming agent, and the resultant porosity of the polymer, does affect accessibility to the binding sites, two related studies carried out by Spivak *et al.* have shown that this is not essential for selective and efficient rebinding [111, 112]. In this work, 9-ethyladenine was used as the template for a range of polymers prepared with different crosslinking ratios and in two solvents, CHCl₃ and MeCN. While the former is not considered a pore-forming agent and the latter is, successful rebinding was achieved in both cases. This suggested that for non-porous polymers the binding sites were not situated on the surface of the pores or the particle but rather in the non-porous gel of the polymer and that this non-porous section of the gel is swollen in the presence of the solvent, thereby allowing the template to diffuse in and out of the gel to the binding sites [112].

The most important finding from this work was that effective rebinding of the template was most successfully achieved when carried out in the solvent in which the polymers were prepared. This was the case for the polymers prepared in both CHCl₃ and MeCN and was attributed to the swelling effects of the solvent and that the exact arrangement and proximity of the functional groups responsible for binding the template was only achieved when the same or similar solvent conditions were observed during rebinding. This was due to the so-called 'solvent memory effect'. While the solvation of the template is often overlooked, the interaction between the template and the solvent both during the formation of cavities and rebinding to those cavities should be taken into consideration.

Studies by Haginaka et al. [113] and Liang et al. [114] have looked at the effect of the porogen on the resultant binding properties and enantioselectivity of polymers prepared using a range of porogens. In the former, Haginaka et al. imprinted d-chlorpheniramine using toluene, phenylacetonitrile, benzylacetonitrile or chloroform as the porogen. These imprinted polymers were prepared *via* a multi-step swelling procedure and the resultant uniform beads were tested as chromatographic stationary phases [113]. The morphologies and surface areas in the dry state of eight different polymers were seen to vary greatly. The highest surface area obtained was $323 \text{ m}^2/\text{g}$ for the polymer prepared using 2-(triflouromethyl)acrylic acid (TFMAA) as the functional monomer and phenylacetonitrile as the porogen and the lowest was $10 \text{ m}^2/\text{g}$ for the same monomer, using chloroform as the porogen. Despite these vast differences in the dry state, however, enantioselectivity of chlorpheniramine was 1.86 for both polymers. This result suggested that despite the low surface area of the latter polymer, when in the solventswollen state, the enantioselective sites were still present and readily accessed by the template. It was suggested, however, that the polymer prepared in chloroform, due to its limited pore structure, could have a lower number of binding sites for the template than the other polymers tested.

Liang *et al.* investigated the use of a range of liquid and solid porogens for their effects on the binding properties and enantioseparation of the enantiomers of ursodeoxycholic acid [114]. Porous core-shell molecularly imprinted polymers were prepared using acrylamide and ethylene glycol dimethacrylate as the functional and crosslinking monomers, respectively. Toluene, acetone and dodecyl alcohol/cyclohexanol mixtures were used as liquid porogens while solid granules of CaCO₃ and Na₂SO₄ were used as solid particle porogens. Polymers without the use of porogens were also studied. As expected, the polymers using toluene and CaCO₃ as porogens, either separately or together, produced the polymers with the highest surface area, 158.83, 130.28 and 138.88 m²/g, for toluene, CaCO₃ and the mixture, respectively. The latter polymer had the largest average pore diameter at 19.39 nm. Not surprisingly, the polymer prepared in the absence of any porogen had the lowest surface area, at 55.41 m²/g. When the enantioselectivity of the polymers was tested, the three polymers with the largest surface area had the highest selectivity factor (α) values, ranging from 2.88 to 3.24, while that which was prepared with no porogen attained the considerably lower value of 1.73.

These results again suggest that the greater the defined pore structure of the polymer, the better the imprinting performance. Once a rigid pore structure has been achieved, appropriate selection of the rebinding solvent should swell the polymer to an extent that will allow for access to, at least, a number of the binding sites. Despite playing this defining role in the ultimate morphology of the polymer, the solvent alone does not dictate this characteristic. Indeed, the crosslinking monomer also has a vital part to play in this area and it is by careful optimisation of the two components that the most suitable morphology may be achieved. Characterisation of such physical parameters is an important aspect of molecular imprinting and will be discussed in more detail in Section 1.6.

1.4.3 The crosslinking monomer

The crosslinking monomer or 'crosslinker' can comprise between 70 and 98% of the total polymer composition and as such has a number of roles in the synthesis of molecularly imprinted polymers [88]. One of the major aspects of the polymer for which the crosslinker is partly or wholly responsible, is the morphology of the polymer, i.e. whether the polymer formed is gel-type, macroporous or microgel powder. It is also the crosslinking monomer which imparts rigidity to the imprinted polymer, thereby enhancing the fidelity of the binding site by 'locking' the functional moieties into the correct spatial and functional configuration for rebinding of the analyte. The final aspect in which the crosslinker plays a role is in the ultimate mechanical stability of the

polymer [2]. While far fewer crosslinking than functional monomers are put to regular use, there is still quite a considerable number and some examples of the most widely used are given in Figure 1.14.



Figure 1.14: Crosslinking monomers reported in the literature: I divinylbenzene (DVB); II ethylene glycol dimethacrylate (EGDMA); III trimethylolpropane trimethacrylate (TRIM); IV methylenebis-acrylamide.

The crosslinkers in Figure 1.14 are examples of a number of different classes of crosslinking monomers that have been put to use during the development of the molecular imprinting technology. Divinylbenzene (DVB), I above, was the first crosslinker employed for the purpose of modern molecular imprinting and this was in the work of Wulff and Sarhan discussed in Section 1.3.1 [16]. A later study by Wulff *et al.* compared the use of DVB with ethylene glycol dimethacrylate (EGDMA) (II) and three other crosslinkers, two other methacrylates and a novel optically active crosslinking monomer [115]. In that work it was found that EGDMA was the most effective crosslinker and polymers prepared with 70-95% crosslinking monomer were the most successful for racemic resolution.

EGDMA has since become the most widely used crosslinker [2] and a number of studies have been carried out that correlate the effect of the crosslinking ratio. In one such investigation, Sellergren observed that the separation factor of the polymer during chromatographic investigation reached a maximum at 75% EGDMA but below this

selectivity started to drop considerably and at 50%, almost all selectivity was lost [116]. More recent investigations have employed novel methodologies, which have also corroborated the importance of a high degree of crosslinker. Notably, the work of Holland *et al.* where GC-MS based thermal desorption studies and UV spectroscopy demonstrated that as the concentration of EGDMA in the polymer composition decreased, so too did the specificity of the polymer [117].

While a high proportion of crosslinking monomer is deemed necessary for selectivity in MIPs, a number of studies have shown that for some applications this may not be necessary. These include the work of Alvarez-Lorenzo *et al.* where soft contact lenses were prepared using molecular imprinting technology to impart selectivity for timolol and ratios of crosslinking (EGDMA) to functional monomer (MAA) as low as 1:10 demonstrated selectivity [118]. Work by Hillberg *et al.* succeeded in preparing theophylline selective flexible membranes with as little as 12.5% (molar) crosslinking monomer (tri-ethylene glycol dimethacrylate) [119].

As well as using divinyl crosslinkers such as DVB and EDGMA a number of trivinyl and tetravinyl monomers, such as the trivinyl monomer trimethylolpropane trimethacrylate (TRIM) (Figure 1.14 **III**) have been used, and, in a number of cases have proven to be more effective than EGDMA. This is exemplified by the work of Mosbach *et al.* for templates such as dipeptides [120] and steroids [121]. Water soluble monomers such as methylene-*bis*-acrylamide (Figure 1.14 **IV**) have also been shown to serve as effective crosslinking monomers for use in aqueous or aqueous based solutions, as seen in the work of Suedee *et al.* where dopamine imprinted polymers were prepared for use in the analysis of urine for adrenergic drugs [122]. Due to the polarity of dopamine it was not soluble in any organic solvents and synthesis of the polymer had to be carried out in aqueous methanol.

Similarly, Cirillo *et al.* employed ethylene-*bis*-acrylamide for imprinting the antioxidant, phytic acid, using a 3:1 water to ethanol mixture as the porogen [123]. The use of the relatively hydrophilic crosslinker was important due to the hydrophilic nature of the template, and thus the necessity to use aqueous solutions as the porogen, as well as the intended end use of the polymer as a controlled release platform. The release profiles of the polymers were studied in simulated gastrointestinal fluids and at acidic pH the drug was completely released from the NIP after 2.5 hours, while only 40% had been released from the MIP at this stage.

An interesting study by Ates and Guven used positron annihilation lifetime (PAL) experiments to study the cavity sizes within radiation induced glucose imprinted polymers prepared with crosslinking monomers with increasing chain lengths [124]. The crosslinking monomers used were diethyleneglycol diacrylate (DEGDA), polypropylene glycol dimethacrylate (PPGDMA) and triethylene glycol dimethacrylate (TEGDMA), and these were studied over increasing molar concentrations relative to the functional monomer, HEMA. The PAL experiments showed that the free-volume radius, R, was only slightly affected by the increasing chain length when concentrations of the crosslinker were low, i.e. 10-30% but at higher concentrations, i.e. 70%, the increased chain length of the PPGDMA contributed to an increase in R of the order of 13%. Swelling studies corroborated the difference in the pore volumes and in terms of molecular recognition, DEGDA showed the highest affinity for glucose. This was attributed to the optimisation of the cavity size for glucose recognition.

Papaionnou *et al.* studied the use of three functional and three crosslinking monomers for imprinting the tripeptide, Arg-Gly-Asp, denoted as RGD [125]. They found that not just the functional monomer but also the crosslinking agent had a profound effect on the percentage net binding of the template, as well as the overall imprinting factor. They observed that for a poly(MAA-co-TRIM) imprinted polymer an imprinting factor of almost 2.5 was observed compared to approximately 1.5 for MAA and bisacrylamide or EGDMA. When acrylamide was used as the functional monomer, however, the polymer using EGDMA as a crosslinker outperformed the others.

Occasionally, researchers have found it useful to employ tailor made crosslinkers, as mentioned in section 1.3.2, where Jiang *et al.* synthesised a custom monomer from chinchonine and was used as both functional and crosslinking monomer for the separation of racemic ketoprofen [65]. Gavrilovic *et al.* synthesised a steroidal crosslinking monomer, which allowed for the separation of testosterone and its epimer, epitestosterone [126]. The use of this crosslinker allowed for postulated Van der Waal's stacking of the steroid and baseline resolution was achieved for the two epimers, which

co-eluted perfectly using the non-imprinted polymer. The steroidal crosslinker used is shown in Figure 1.15.



Figure 1.15: Steroidal crosslinker used for the separation of testosterone and epitestosterone [126].

As is the case with the functional monomer and the porogen, careful selection of the crosslinking monomer is also necessary and while it is generally stated that the crosslinker does not interact to any great extent with the template, it is necessary to match reactivity ratios with the functional monomer and take factors such as solubility, etc. into consideration.

1.4.4 Polymer initiation

While the methods of initiation applied to molecular imprinting technologies have been many and varied, including γ -irradiation [127], an ever-increasing use of iniferters [128, 129] and the production of radical species using peroxide based compounds [130], the most widely used strategy remains the use of azo-compounds and their decomposition to produce radical species [1]. Compounds such as 2,2'-azobis(2-methylpropionitrile) (or azobisisobutyronitrile - AIBN) [131], 1,1-azobis(cyclohexanecarbonitrile) [132] and 4, 4-azobis(4-cyanovaleric acid) [133] have all been utilised for MIP preparation.

These azo-compounds can undergo decomposition to form radicals using both thermal and photochemical means. The use of photochemical means allows for initiation at low temperatures, which could better promote interaction between template and functional monomer. Studies published in 2011 by Zhang *et al.* [134] and Dopico-Garcia *et al.* [135] have compared the two methods of initiation.

The former study used AIBN as the initiator and demonstrated that while the polymers prepared by both means had an almost identical molecular structure (as observed by FT-IR), transmission electron microscopy (TEM) demonstrated that differences in the pore structure of the MIP and NIP were more pronounced for photo-initiated than thermally initiated polymers. This difference between the imprinted and non-imprinted polymers for photo-initiated polymers was considered to be important for the performance of the polymers and results showed that imprinted polymers prepared in this fashion had higher adsorption capacities and selectivity than their thermally initiated counterparts.

Dopico-Garcia *et al.* also found that photo-initiated polymers outperformed thermally initiated polymers, with the former having the capability to rebind the template while the latter did not. It is worth noting, however, that Lu *et al.* [136] state that while low temperatures allow for the formation of a more stable template-monomer complex in the pre-polymerisation solution, higher temperatures can improve the quantity and quality of template recognition sites within the polymer by achieving more complete polymerisation reactions.

1.4.5 A rational design approach to molecular imprinting

Judicious selection of functional and crosslinking monomers, as well as solvent and initiating strategy is most often determined by empirical means and this method of determining the most appropriate composition is perhaps still the most popular and successful route for MIP synthesis. Due to the very often time consuming nature of this method, various approaches have been taken in an effort to reduce the time necessary to synthesise successfully imprinted polymers. This ever-increasing number of methods aimed at improving the speed and success with which MIPs are developed is often collectively termed as a 'rational design approach'.

It is widely accepted that the success of non-covalent molecular imprinting is due in no small part to suitably strong complexation of the template and functional monomer in the pre-polymerisation complex [106, 137, 138]. While it is recognised that these adducts are exposed to a much different environment during polymerisation (due to factors such as high temperatures and growing polymer chains), it has been shown that

polymers prepared in such a way as to maximise complex formation have achieved good success in terms of ultimate performance [139-141].

A variety of computational and spectroscopic methods have been developed (for use alone or in tandem with each other) to fully illuminate the processes taking place in the pre-polymerisation complex, thereby allowing for the most suitable selection of the components for use in the preparation of MIPs. The first of these to be discussed here are the computational approaches.

1.4.5.1 Computational approaches to the rational design of MIPs

One of the earliest MIPs prepared using computational methods to predict interactions in the pre-polymerisation complex was a biotin imprinted polymer synthesised by Takeuchi *et al.* [142]. The molecular modelling studies carried out here were based on Monte Carlo simulations, a computational algorithm. These simulations predicted that the most stable complex structure for the functional monomer, MAA, and the biotin derivative being used as a template, involved the formation of bidentate hydrogen bonds between the two compounds. This was supported by NMR studies and the polymer prepared was capable of binding 6.81 μ mol/g of polymer and Scatchard analysis gave association constants of 2.5 × 10⁵ M⁻¹ and 6.3 × 10³ M⁻¹ for the high and low affinity binding sites, respectively.

A range of molecular modelling software packages have been employed in the preparation of MIPs, including the Cerius² software utilised by Pavel and Lagowski [143], GaussView by Isarankura-Na-Ayudhya *et al.* [144] and Gaussian03 software by Diaz-Diaz *et al.*, density functional theory [145], and the Hartree-Flock model by Li *et al.* [146] and Mojica *et al.* [147]. A number of packages have stood out in the literature, having demonstrated success for a variety of templates. One such software package is that developed and used over the last decade by the group of Piletsky *et al.*, i.e. The Leapfrog algorithm.

The Leapfrog algorithm is used to screen a library of the most commonly used functional monomers against a variety of templates, in order to determine which monomer has the highest binding score, a value incorporating Van der Waal's forces, hydrogen bonds, ionic interactions and spatial considerations [87]. Once the most suitable monomers have been selected, a simulated annealing process is employed whereby the template is placed inside a virtual box with the largest number of the chosen functional monomer possible. Energy is then added to the system in a series of energy minimisation and heat addition steps and the optimal template to monomer ratio is determined. This system has been used for the imprinting of abacavir, a HIV-1 reverse transcriptase inhibitor [148], the mycotoxin aflatoxin B1 [149], cholic acid [150] and atrazine [151]. In the last example, three monomers were identified as the most appropriate, MAA, itaconic acid (ITA) and acrylamide (AA). The software generated complexes for the three monomers and the template are shown in Figure 1.16.



Figure 1.16: Software images for the complexes between, atrazine and I MAA; II ITA; III AA. Yellow dotted lines represent H-bonding interaction [151].

Figure 1.16 **I** shows that two molecules of MAA were predicted to interact with the template while only one of each of the other two monomers was seen to take part in complex formation. This proved to play a significant role in the resultant binding properties of the imprinted polymers and the MAA based polymers were shown to have the highest specificity (~100% analyte rebound in the MIP and <10% in the NIP) and capacity (12.5 mg/g of polymer) of the three monomers tested [151].

This software uses the dielectric constant of the solvent to account for the effect of the solvent and allows for the input of the crosslinking monomer into the software. However, as can be seen from Figure 1.16, this is generally not included, as the interaction between the crosslinker and the template is considered to be negligible.

Another notable software package is the AMBER programme, developed by Pearlman *et al.* to "simulate the structural and energetic properties of molecules" [152], which has been adapted by Mizaikoff *et al.* for use in the study of the pre-polymerisation complex in the preparation of molecularly imprinted polymers for 17 β -estradiol [107, 153]. This software package places more focus on the hydrogen bonds formed as this is deemed to give more direct information regarding the strength of the bonds in the template-monomer complex in the pre-polymerisation solution. This package also allows the input of solvent molecules into a virtual box, analogous to the one used by Piletsky *et al.* and incorporates factors, such as the density of the solvent, in subsequent calculations.

This molecular dynamics simulation approach was retrospectively used to provide a more detailed insight into the pre-polymerisation solution of a successfully prepared 17 β -estradiol imprinted polymer [107]. The polymer had been prepared in an earlier work using MAA and DVB as the functional and crosslinking monomers, respectively and yielded polymers with a mean affinity constant of 2.45 × 10⁻¹¹ M⁻¹ (obtained from Freundlich isotherm) [154]. An image of the pre-polymerisation complex is portrayed in Figure 1.17.



Figure 1.17: Image of the pre-polymerisation complex for 17β-estradiol MIP, generated by the Amber software package. Colour scheme: 17β-estradiol: orange; MAA: blue; DVB: grey. (From reference [107]

In later work on 17 β -estradiol, results from modelling experiments suggested that the monomers MAA, 2-(diethylamino)ethyl methacrylate and methacrylamide were the most suitable monomers [153]. Since MAA had been shown to be successfully used for the preparation of 17 β -estradiol imprinted polymers [154], this was chosen for further NMR investigations where the optimum 6:4 MAA:17 β -estradiol complex formation predicted by the modelling study was confirmed by NMR analysis.

The more recent use of the Amber software suite, as applied by the group of Nicholls *et al.* for computational analysis that is stated to be "the simulation of a complete molecularly imprinted polymer pre-polymerisation system" [155] could, at present, be deemed the state of the art in computational modelling. The first study by this group used molecular dynamics software to look at the interactions of the template, bupivicaine, and the other components in the pre-polymerisation solution. In this work, two pre-polymerisation solutions were studied: the solution containing the template, functional monomer, crosslinker, initiator and solvent, and the solution with the crosslinker and initiator omitted. The presence of the initiator and the crosslinker was seen to have a clear impact on the interactions between the crosslinker, EGDMA, and the template, though weak, were present and indeed contributed to the presence of shape selective sites in the resultant polymer.

While interaction between the template and the initiator were not observed, it was noted that dimerisation of MAA was less favoured when both the crosslinker and the initiator were present. The porogen, chloroform, was seen to be mainly located in the non-polar regions of the template. This use of the Amber software has since been successfully applied to targets such as the anticoagulant warfarin [156] and the vitamin biotin [157].

Interestingly, in the preceding six examples spectroscopic analysis was used in conjunction with the computational systems in a bid to further elucidate the interaction events taking place in the pre-polymerisation mixture with good agreement between the two methods being demonstrated. The use of spectroscopic analysis either alone or in conjunction with molecular modelling software, has become a powerful tool in the rational design of molecularly imprinted polymers.

1.4.5.2 Spectroscopic methods for the rational design of MIPs

Spectroscopic methods, be they NMR, IR or UV, allow for a property of either the template or the functional monomer to be monitored as it changes due to complexation with the other in an environment chosen to mimic the pre-polymerisation solution [106, 158, 159]. Here Job plots, where the total molar concentration of two complex forming species is held constant while varying their respective mole fractions [160], can be used for determination of the complex ratio. The construction of titration curves by changing the concentration of one component while holding the other constant and monitoring the change in signal of either component [161], can be used to calculate association constants.

Early work carried out by Sellergren *et al.* employed NMR spectroscopy to fully elucidate the enantioselectivity and substrate selectivity of L-phenylalanine imprinted polymers [162]. In this study a number of polymers were prepared with the L-form of phenylalanine derivatives and their selectivity was demonstrated using chromatographic methods. For the L-phenylalanine anilide imprinted polymer a capacity factor, k'_L , of 4.3 and a separation factor, α , of 3.2 for the D- and L- forms were achieved. NMR studies were used to rationalise the behaviour of the imprinted polymers. This spectroscopic technique was used to monitor changes in the line widths of the protons of the MAA and the exchangeable protons of the amino and carboxyl groups of the template and revealed that this was due in part to the presence of 1:1, 1:2 and 1:3 template:monomer complexes in the pre-polymerisation solution.

Later work by Nicholls and Andersson used UV spectroscopy to investigate the prepolymerisation complexes of MAA based polymers previously imprinted by Nicholls *et al.*, namely cinchonidine, yohimbine and *N*-acetyl-L-phenylalaninyl-L-tryptophanyl methyl ester imprinted polymers [163]. This work was presented as a suitable method for optimising the pre-polymerisation composition of new imprinted polymers and since then spectroscopic methods have been put to routine use in the rational design of molecularly imprinted polymers.

A notable example of the use of spectroscopic methods in the evaluation of prepolymerisation complexes was a three part NMR study by Ansell *et al.* where the most suitable monomers and solvents were chosen for the preparation of ephedrine MIPs by first fully probing the interactions between MAA and the template [164]. This was followed by determining the most suitable solvent for use with this monomer [165] and finally investigating the use of three other functional monomers [141]. This study yielded a polymer capable of achieving an α value of 2.46 for the separation of (±)-ephedrine. The effect of the crosslinker in the pre-polymerisation solution was also investigated in each case, demonstrating the extent to which the pre-polymerisation conditions may be examined.

The work of Osmani *et al.* has involved the use of two spectroscopic methods, where both NMR and IR were combined to elucidate the interactions between the template, 2-aminopyridine (2-apy), and the functional monomer, MAA [106]. The complementary correlation between the findings of the NMR and FT-IR studies is here exemplified by agreement of the experimentally derived complex stoichiometry represented in Figure 1.18.



Figure 1.18: Job plots obtained by (a) FT-IR and (b) NMR experiments both depicting a 1:1 stoichiometry for the 2-apy-MAA complex in the pre-polymerisation solution [106].

Zhang *et al.* have quite cleverly used ¹H NMR studies to study their theories on the effect of functional monomer aggregation on the presence of background sites in imprinted and non-imprinted systems [166, 167]. In their work on the use of a urea monomer, they have looked at monomer-monomer and monomer template aggregation [167]. Solution state NMR studies demonstrated that the monomer-monomer aggregation had a constant, K_{agg} , of 3.5 M⁻¹ in chloroform (the porogen) alone, but when DMSO was added, no aggregation took place. When the interaction between the template, tetrabutlyammonium diphenyl phosphate, and the monomer was studied in the presence of the porogen (chloroform), a value of 176 M⁻¹ was observed for the association constant.

It was postulated that when the template was present in the pre-polymerisation complex, the interaction between template and monomer would predominate, forming templatemonomer complexes. When no template was present, however, the monomers would self-associate and limit the number of "background binding sites" and thus, significantly reduce the non-specific binding due to binding moieties, which would typically be indiscriminately spread throughout the polymer matrix. Indeed, the imprinted polymer had a capacity of 23 μ mol g⁻¹, which was a four-fold increase over that of the non-imprinted polymer at only 5.4 μ mol g⁻¹, demonstrating a strong imprinting effect.

As previously mentioned, studies were carried out on the aggregation of the functional monomers in increasingly polar solvents (by addition of DMSO) and when the resultant polymers were formed, the MIP had a modest decrease in its uptake capacity. What was more significant, however, was the large increase in the uptake capacity of the NIP, which rose from 5.4 to 15 μ mol g⁻¹ upon addition of the DMSO to the prepolymerisation mixture. This was attributed to the disruption of the self-association of the monomer in the pre-polymerisation of the binding sites by the presence of the competing polar solvent.

What can be seen from this and the preceding sections is that the use of molecular modelling and spectroscopic methods can aid in the careful selection of the best components for the preparation of MIPs. These methods are typically aimed towards the rational design of molecularly imprinted polymers and while there are still arguments against this, what is important about these methods is that they contribute towards a deeper understanding of the processes that are involved in the synthesis of MIPs.

Whether using this approach, using tried and tested monomers or empirically testing a range of compositions, it is important that the most appropriate polymer format and method for polymer preparation be chosen, taking into account factors such as the final application of the polymer.

1.5 Methods of polymerisation

As with all polymers, molecularly imprinted polymers can be prepared *via* a range of polymerisation methods and, since the earliest work on imprinted polymers a number of general polymerisation strategies have been adapted for use in molecular imprinting. Presented here is a brief outline of some of the common polymerisation methods that have been applied to the synthesis of MIPs.

1.5.1 Bulk polymerisation

The preparation of a polymer in bulk monolith format is achieved by mixing together all of the components, i.e. the template, the monomers and the initiator, in the solvent, where polymerisation takes place. All components must be soluble in the solvent and the monomer concentration in terms of total volume of pre-polymerisation solution is typically 40-60% [121]. The result is a polymeric block comprised of densely fused and interlinked microgels. The polymeric block is then crushed, ground and sieved to obtain particles of the required size.

Since the first organic imprinted polymer by Wulff and Sarhan [16], this has historically been the most widely used strategy for the preparation of MIPs [133]. The widespread use of this method can be ascribed to its synthetic simplicity and general applicability, although it also has a number of disadvantages, including the need for labour intensive grinding and sieving, the loss of a high percentage of polymer (up to 70%) during the sieving process and the possible destruction of binding sites during the grinding process. While this method is still regularly used [45, 141, 145, 168], these disadvantages, coupled with the possible advantages (and often times necessity), of using different

polymer formats appears to have brought about a shift in focus in terms of polymerisation strategies.

1.5.2 Molecularly imprinted beads

A number of techniques have been applied to the preparation of MIPs in bead form, as good control over particle size may be achieved and spherical particles are considered advantageous for a number of applications, such as chromatography [169] and drug delivery [170]. A range of approaches have been used, most notably suspension and precipitation techniques.

1.5.2.1 Suspension polymerisation of MIPs

Suspension polymerisation involves the dispersion of, typically water-insoluble, monomers as small droplets in the chosen solvent with each of the monomeric droplets behaving as a "mini-bulk reactor" [23, 133, 171]. Early work utilising suspension polymerisation for non-covalently imprinted polymers was carried out by Mayes and Mosbach, where a liquid perfluorocarbon was employed as the dispersing phase [172]. The reaction was UV initiated and depending on the amount of the stabilising polymer used, the bead size could be controlled to between 5 and 50 μ m. While the fluorocarbon has the advantage of not disrupting the interactions between the template and the monomers, as would be the case with water, it has the disadvantage of being quite expensive.

An alternative to this method was presented by Kempe and Kempe, where mineral oil and acetonitrile were used as the dispersant and solvent, respectively, for the preparation of spherical propranolol imprinted beads prepared in roughly the same size range as those of Mayes and Mosbach [173]. This made the process cheaper and acetonitrile's lack of solubility in mineral oil negated the need for added stabiliser. Acetonitrile can, however, interfere with H-bonding interactions and the use of other porogens would not be suitable with mineral oil. Where mineral oil is not used, stabilisers are generally employed and can be responsible for unwanted functionality on the surface of the polymer. While these novel methods of suspension polymerisation have been put to use, conventional suspension methods currently enjoy a large degree of popularity and have been used for templates such as the triazine, melamine [92], the peptide hepcidin [174] and Sudan dyes [175]. As discussed in section 1.3.3, Curcio *et al.* synthesised a surfactant template based on glucopyranoside (see figure 1.9) [71]. This surfactant template had the dual role of allowing for an emulsion to be formed, as well as ensuring that due to its surfactant nature, the imprinted sites would be at the surface of the imprinted bead, i.e. the interface of the two phases. In this case, a mini-emulsion was formed by sonication allowing for the synthesis of beads in the range of 150 nm.

1.5.2.2 Precipitation techniques for MIPs

Precipitation polymerisation is a technique that will be discussed in more detail in Section 2.1.4 but a brief introduction to the technique will be presented here. This strategy is based on the use of diluents in which the monomers (typically <5% of the total volume) are soluble but the growing polymer chain becomes increasingly insoluble, thus causing the polymeric beads to precipitate out of solution [20, 176, 177]. Beads are formed in the range of 0.3-10 μ m, no added stabiliser is necessary and careful optimisation of the solvent composition allows good control over particle size and morphology.

Early work by Mosbach *et al.* employed this method for the synthesis of theophylline and 17β -estradiol imprinted beads, where the microspherical particles exhibited superior rebinding ability over the bulk monolith particles [121, 178]. This method has since been used for the preparation of beads imprinted for chemical warfare agents [58], as well as anticonvulsant [179], antimalarial [180] and antipsychotic drugs [89].

Recent work by Long *et al.* has provided a greater understanding of the processes taking place in MIPs prepared by precipitation polymerisation [181]. For this insightful study, NMR and dynamic light scattering techniques were utilised to investigate a range of factors including the amount of monomer incorporated into the polymer relative to time as well as the size and growth rate of the nanoparticles. Two model templates were used, propranolol and testosterone, and NMR studies were carried out on the polymerisation mixture at different time points. For the propranolol system,

approximately 60% of the free template was bound to the polymer after twenty four hours, with 70% of this bound template associated with the particles after the first three hours. This suggested that the majority of binding sites were already formed during the first three hours. When testosterone was studied, it was observed that only 20% of the template was associated with the polymer after the full twenty-four hours. This disparity in the amount of material associated with the polymer was predicted by prepolymerisation binding studies where the only noticeable complex induced shift observed was a change in chemical shift of 0.041 ppm, related to a weak hydrogen bond to the 17 β -OH of the steroid. This was in stark contrast to the changes in chemical shift observed for propranolol, where the change in shift values ranged from 0.224 to 0.505 ppm.

Rebinding studies were also carried out on the growing polymer particles at time points throughout the study and it was seen that after three hours, the amount of propranolol bound to the imprinted polymer remained relatively static. This was in good agreement with the suggestion that the majority of the binding sites were formed by this time. Dynamic light scattering studies showed that the hydrodynamic volume of the polymer particles changed to a small extent after only thirty minutes but the apparent swelling of the polymers suggested that the particles became more heavily crosslinked as reaction time increased.

This particular study, while being informative about the processes involved in precipitation polymerisation for MIPs, also serves to demonstrate an inherent problem in the method: when the degree of interaction between the template and the monomer is low, as was the case with testosterone, the final amount of template associated with the polymer prepared in such dilute solutions will also be considerably lower than is typically desired for molecularly imprinted polymers.

1.5.2.3 Core-shell MIPs

A method which has also been used for the preparation of spherical MIP beads, and which may to some extent counter the dilution problems of precipitated beads, is the use of core-shell polymerisation techniques. Carter *et al.* [182] have attributed the origins of this method for use in imprinted polymers to Yu *et al.* during their work on the selective

extraction of copper (II) and nickel (II) [183]. During work on the use of core-shell imprinting for cholesterol, Perez *et al.* [184] have outlined the process as having two distinct stages; firstly a seed (or core) is formed, often by emulsion polymerisation of any of a number of monomers. This seed is then added to another batch of monomers (or the monomers are fed to the seed solution) and the thin imprinted 'shell' is then formed on the surface of the core. This has the obvious advantage that the template should be more readily removed from this thin layer of polymer and conversely, binding kinetics should be faster. Other advantages include the degree of control over bead size and the ability to prepare larger beads.

The term 'core-shell imprinting' has come to encapsulate a number of different strategies. Perez *et al.* [184, 185], Carter *et al.* [186] and more recently Liang *et al.* [114, 187] have utilised monomers such as styrene, DVB and EGDMA to synthesise the core *via* conventional emulsion methods, while Ma *et al.* [188] and Jiang *et al.* [65] have employed silica nanoparticles as the core for beads selective for estradiol and ketoprofen, respectively.

Barahona *et al.* have applied to molecularly imprinted polymers, the method developed by Stover and Li [189] for the synthesis of core-shell beads *via* two-step precipitation [56, 190]. Here, the core-shell MIP beads are prepared by first synthesising the core using a simple precipitation method for crosslinked DVB beads. These DVB microsphere cores were then coated with the 'shell', which was composed of a mixture of DVB, MAA and the template, the fungicide thiabendazole. This resulted in the preparation of an inline solid-phase extraction method where it was possible to inject samples of citrus fruits and orange juices, from which accurate determination of the target as low as 0.8 mg kg^{-1} was achieved.

1.5.3 Film and lens type MIPs

Applications such as sensors require the preparation of imprinted polymers in formats suitable for such use. In a number of cases the preparation of polymers as thin films has been used to prepare sensors for targeting analytes such as myoglobin [55], billirubin [191] and melamine [192]. Methods such as spin-coating [193, 194], surface-grafting polymerisation [195] and electropolymerisation [196] have been employed to

incorporate molecularly imprinted films of the requisite thickness onto the surface of sensors such as quartz crystal microbalances and potentiostats.

The preparation of lens type materials for application in drug delivery also requires the synthesis of polymers in formats suitable for such use. Hiratani *et al.* [197-199] have employed polypropylene and Teflon to construct moulds capable of forming gels of a given thickness to develop the use of molecularly imprinted contact lenses. Careful consideration must be given to the biocompatibility of the monomers and polymer when used for such applications.

Disadvantages of these methods can include low density of imprinted sites and slow diffusion of the analyte, as well as thickness control [44] [200].

1.5.4 Advantages and disadvantages of selected polymerisation strategies

There are a number of advantages and disadvantages associated with each polymerisation method and these are summarised in Table 1.2.

Technique	Advantages	Disadvantages
Bulk monolith	Easily optimised	Irregular particle size and shape
	Suitable for most polymer	Low yield of usable polymer
	formulations	Time and labour intensive
Suspension	Good control over particle	Need for surfactant or stabiliser
	size	Issues surrounding component
	Spherical particles obtained	solubility in continuous phase
Precipitation	Good control over particle	Needs optimisation for each set of
	size and morphology	functional monomers
	Spherical particles obtained	
	No stabiliser necessary	
Core-shell	Good control over particle	Multiple steps in synthesis
	size and morphology	Often stabiliser necessary
	Spherical particles obtained	
Thin film	Well suited for specific	Slow diffusion of analyte
	applications, e.g. sensors	Poor thickness control

Table 1.2: Advantages and disadvantages of some common methodologies employed for the preparation of MIPs

While the polymerisation strategies listed should not be considered exhaustive, the applications, as well as the advantages and disadvantages associated with each method serve to demonstrate that careful consideration of the final application is necessary before deciding which polymerisation methodology to pursue. Whatever the strategy used, physical characterisation of the resulting polymer is necessary to gain a full understanding of the ultimate behaviour of the polymers.

1.6 Characterisation of molecularly imprinted polymers

The characterisation of molecularly imprinted polymers can be roughly divided into three areas: chemical characterisation, morphological characterisation and characterisation of binding site or molecular recognition properties [88].

1.6.1 Chemical characterisation

The inherent insolubility of crosslinked molecularly imprinted polymers means that chemical characterisation is most readily achieved using spectroscopic methods. Fourier-transform infrared spectroscopy (FTIR) is one such method. This has been applied to determine the incorporation of functional groups, as well as the extent of polymerisation. Wang *et al.* have employed this technique for surface-imprinted polymers for 3-chlorophenol where they monitored the successful modification of the core silica particle [201] while Zhang *et al.* have used FTIR to determine the differences in polymers initiated by either photochemical or thermal means [134].

Solid-state NMR is another technique that provides information on the chemical composition of molecularly imprinted polymers [104]. Analysis of fenitrothion imprinted polymers has been carried out by Pereira and Rath using this technique [202], while Weng *et al.* have characterised MIPs with catalytic activity towards *o*-fluoroacetophenone in the same manner [203].

1.6.2 Morphological characterisation

The morphological properties of molecularly imprinted polymers may be investigated by using any of a number of techniques. Solvent uptake studies are one example, where the amount of solvent taken up by the polymer, can be used to give an estimation of the specific pore volume of the polymer [88]. Solvent swell ratios were employed by Holland *et al.* as part of a study on the influence of morphology on the performance of 2-amino-pyridine imprinted polymers prepared using varying amounts of crosslinker and polymers with higher surface areas were shown to have lower swell ratios [204].

Microscopy has also been used to study the morphology of imprinted polymers with simple optical microscopy being employed by Wei *et al.* to obtain a rough estimation of the particle size and shape of estradiol imprinted polymers prepared by precipitation polymerisation [154]. This group then used scanning electron microscopy (SEM) to get a more detailed analysis of the surface morphology of the polymers, a technique also employed by Esfandyari-Manesh *et al.* when investigating the effect of the porogen on precipitated polymers [179]. Transmission electron microscopy (TEM) has also been employed for morphological analysis in imprinted polymer technologies [205].

Nitrogen physisorption experiments can reveal a great deal of information relating to the surface area, pore size and pore size distribution of polymers in the dry state [88, 104]. This technique involves the construction of adsorption-desorption isotherms by exposing a fixed mass of dry material to a gas at a constant temperature and monitoring the equilibrium pressures, as a series of known quantities of gas are added to the sealed chamber [88, 110, 206]. As it is relevant to this study, this topic will be dealt with in more detail in Section 3.1.3.

1.6.3 Binding site characterisation

Quantitation of the strength and number of binding sites present within the matrices of molecularly imprinted polymers is essential to a full understanding of the effect templating processes have on polymer synthesis. This is typically achieved by obtaining adsorption data for a polymer over a specified concentration range, where concentrations of bound (B) and free (F) analyte are measured. This data is then treated using a suitable mathematical model capable of calculating physical parameters such as the number (N) and affinity (K) of the sites present.

The construction of Scatchard plots has been employed to achieve this throughout the development of imprinting technology [76, 207, 208]. This is a method used to analyse the data for reversible ligand-receptor interactions and the plot should provide a straight line, the slope of which gives the binding constant, K_a [104]. This method has typically produced a curve when applied to MIP binding data, which in turn has been used to construct two straight lines, suggesting the existence of two types of binding site [207]. In fact, Umpleby *et al.* discuss how the binding site heterogeneity that is practically inherent to MIPs can be problematic when applying the largely homogenous model of Scatchard analysis [207].

In the early 2000s a series of studies published by Shimizu and co-workers, investigated the application to imprinted polymers of binding models that may be more readily used for heterogeneous materials [207, 209-213]. These studies have focussed on application of the Freundlich and the Langmuir-Freundlich isotherms. Both of these models are power functions, which take into account the heterogeneity index, m, and a, a variable related to the median binding affinity as well as the concentrations B and F [211, 212].

This group applied the Freundlich isotherm to data obtained for twelve polymers previously reported in the literature and found that a "consistent mathematical relationship" was observed for the data [211]. This suggested that the Freundlich isotherm is generally appropriate for the calculation of binding parameters for imprinted polymers. They did note, however, that the Freundlich isotherm can only be reliably used for sub-saturation concentrations and its applicability for a particular polymer must be shown before the data obtained is to be considered valid.

The Langmuir-Freundlich, however, is free from the limitation of studying subsaturation concentrations [212] and while it is considered to be more universally applicable, the Freundlich isotherm is considered to be more readily applied and sufficient in the majority of cases [213]. A more detailed discussion of the use and choice of these models will be presented in Chapter 3 but it is worth noting that these models are continuous distribution models that incorporate a distribution of the totality of binding site types.

1.7 Applications of molecularly imprinted polymers

The high degree of selectivity and specificity reported for MIPs, coupled with their inherent robustness and stability under harsh conditions and in a variety of environments, make them attractive for use in any number of applications. The following sections demonstrate how the ability to prepare molecularly imprinted polymers for a wide range of templates has made them the focus of intense research in a number of areas where the specific recognition of an analyte is necessary. This is not an exhaustive list of the myriad applications to which MIPs have been put, rather a set of examples selected to demonstrate their wide-ranging applicability.

1.7.1 Solid phase extraction

Solid phase extraction (SPE) is a sample preparation technique used in complex matrices or where the target analyte is present in only trace amounts. Here, either the analyte of interest or the interfering components are adsorbed onto the sorbent allowing for subsequent elution and enrichment of the target analyte.

Since Sellergren first used molecularly imprinted polymers for the selective enrichment of pentamidine [214] SPE has become one of the most successful applications of MIPs where the technology, now known as molecularly imprinted solid phase extraction or MISPE, has found commercial success with examples such as SupelMIP [215] and POLYIntell [216]. Templates with a wide range of functionality have been the focus of research aimed at the preparation of molecularly imprinted sorbents for SPE and some examples are presented in Figure 1.19.



Figure 1.19: Templates reported in the literature for use in the preparation of MISPE techniques: I Protocatechuic acid [217]; II Bensulfron-methyl (BSM) [57]; III Berberine chloride [218]; IV β -sitosterol [219].

As discussed as far back as the fifties by Dickey [9], an inherent problem pervasive in the technology of molecular imprinting is that of un-extracted template. In few applications is this problem as relevant as it is in solid-phase extraction, where the predominant objective is to enrich analytes that are present in trace quantities. A method that has been used to counter this problem is the use of a template analogue, i.e., using a template that is similar in shape, size and functionality to the target analyte, but which will not interfere with analysis of the target analytes.

The use of this method for MISPE is exemplified by the work of Baggiani *et al.* for solid-phase extraction of a class of azo-dyes, known as Sudan dyes, in food samples

[175]. The template analogue and two of the nine analytes studied are shown in Figure 1.20.



Figure 1.20: I 1-(4-chlorphenyl)azonaphthalen-2-ol (chloro-Sudan) the template analogue used for analysis of Sudan dyes, including II Sudan I and III Sudan II [175].

Preliminary investigations in this work showed that there was significant bleed when Sudan I was used. For this reason, I in Figure 1.20 was synthesised for use as a 'template mimic'. This approach allowed for preparation of an imprinted polymer capable of the selective extraction of several banned Sudan dyes from five different food samples. While template bleed was still observed at a low level, approximately $0.01-0.05 \ \mu g/mL$, the retention time in the analytical method was different than all of the other analytes allowing for quantitation of the dyes to as low a level as $0.75 \ \mu g/mL$ in spiked food samples.

1.7.2 MIPs in catalysis

Catalysis is a process whereby the rate of a chemical reaction is altered by the use of another substance, which itself, is not used up in the reaction. The ability to use MIPs as catalysts in chemical reactions has considerable appeal due to their ability to selectively bind a target analyte under a variety of conditions. While many reports of using substrates for a particular catalytic or enzymatic reaction as templates exist, such as *p*-nitrophenyl palmitate as a lipase substrate [220] and glutathione as a glutathione peroxidase substrate [221], a good deal of the work involving catalytic MIPs has involved the preparation of an imprinted polymer, specific for a particular transition state analogue, TSA. The MIP stabilises the transition state of a particular reaction to a

greater extent than the ground state [222]. This is analogous to the use of catalytic enzymes and results in an overall reduction of the activation energy for the process, thus enhancing the rate of reaction [1]. The ability of MIPs to selectively bind a target analyte in the presence of structurally similar molecules is essential, as the TSA may be similar in structure to either the reactant or the product or both.

The TSA approach has been employed for ester hydrolysis [223], class I and class II aldolase reactions [224, 225], transamination [226] and Diels-Alder reactions [227]. A recently reported example of the use of the TSA approach by Li *et al.* has employed the use of 4-nitrophenyl phosphate imprinted sites to create a catalytically active and thermosensitive polymer for the hydrolysis of 4-nitrophenyl acetate [228]. This positively thermosensitive polymer used the interaction between poly(2-trifluoromethlyacrylic acid) and poly(1-vinylimidazole) to prepare a polymer that possessed catalytic capability at 40 °C but had a significantly reduced capability at 20 °C.

As discussed in section 1.3.4, stoichiometric non-covalent strategies are particularly well suited to catalytic applications due to the use of strong 1:1 template to functional monomer interactions, reducing the need for excess functional monomer thus allowing for precise positioning of the functional groups and reducing non-specific interactions. An example of the transition state analogues and substrates employed by Wulff *et al.* for the preparation of catalytic MIPs using stoichiometric non-covalent methods is shown in Figure 1.21.



Figure 1.21: Transition state analogues, I and II, and substrates, III and IV, used by Wulff *et al.* in the preparation of MIPs with carboxypeptidase A activity [83, 229].

The tetrahedral templates presented in Figure 1.21 were used to mimic the tetrahedral transition state intermediate in the hydrolysis reactions of the substrates. This enabled the preparation of molecularly imprinted polymers that were capable of enhancing the rate of carbonate hydrolysis by a factor of up to 410 000 [83]. While this rate enhancement is atypically high for MIPs, with rate enhancements of 20-50 commonly reported [225, 226], it serves to show the capabilities of molecularly imprinted catalysts.

1.7.3 MIPs in drug delivery

The use of molecularly imprinted polymers in drug delivery has received considerable attention since early work by Sreenivasan proposed the use of MIPs as drug retaining matrices with controllable release properties [97] and Norell *et al.* investigated their use as sustained release platforms [230]. The drugs used as templates in both studies were, respectively, the corticosteroid, hydrocortisone and the methyl xanthine, theophylline.

Norell *et al.* showed that at pH 7.0, the rate of release of theophylline was 4×10^{-5} mmol g⁻¹ min⁻¹ for the NIP and 3×10^{-5} mmol g⁻¹ min⁻¹ for the MIP [230]. While it was noted that the difference in behaviour between the imprinted and non-imprinted system was relatively small, it served to demonstrate the potential use of MIPs in sustained release drug delivery systems (DDS).

Sreenivasan showed that 95% of the rebound template, hydrocortisone, was retained in the imprinted polymer matrix after thirty days in a methanol/water mixture compared to almost 80% released after 1 day in the NIP [97]. Subsequent work on hydrocortisone imprinted polymers demonstrated the potential of MIPs as template responsive release systems [231]. In this study testosterone was loaded onto the hydrocortisone MIP and its rate of release was monitored both with and without the presence of hydrocortisone in the release solution. To 100 mg of MIP, 175 μ g of testosterone was rebound and during release studies where no template was present only 79 μ g was released after 24 hours. When hydrocortisone, the template, was present in the release solution, 171 μ g of testosterone were released in 4 hours.

These early demonstrations of the potential of MIPs in drug delivery appear to have coincided with a change in focus in the development of DDS towards 'intelligent therapeutics', where drug delivery systems are now required to respond to the individual patient's needs by releasing drugs in response to a particular location, environment or the presence of biological triggers [232]. It has been proposed that it is in meeting these demands that the technology of molecular imprinting may have its greatest potential [233]. A number of strategies and approaches have been investigated in an effort to meet these demands.

One such strategy is the use of molecularly imprinted polymers as excipients in sustained delivery platforms, exemplified by the work of Wang *et al.* where polymers imprinted with dexamethasone-21 phosphate disodium (DXP) were used to release the anti-inflammatory drug, dexamethasone, for reduction of inflammation at the site of an implantable biosensor for glucose [234]. Indeed, the imprinted polymer was seen to release the template at lower pH, which would be typical of the site of inflammation. A composite was formed using a hydrogel, prepared from HEMA, and the imprinted nanospheres prepared by precipitation techniques. The use of the hydrogel/nanospheres composite prolonged the release of the template by about 50 days longer than the hydrogel on its own, as shown in Figure 1.22 (b).



Figure 1.22: (a) Structure of dexamethasone-21 phosphate disodium (DXP); (b) release profile of DXP from (i) hydrogel alone and (ii) hydrogel/imprinted nanosphere composite [234].

The use of hydrogels in molecularly imprinting is an approach which allows for the direct synthesis of molecularly imprinted drug delivery platforms. Hydrogels are insoluble, cross-linked polymer network structures composed of hydrophilic co-polymers that have the ability to absorb significant amounts of water [235]. This ability to absorb significant quantities of water makes them suitable for use in biological

applications. The groups of Hiratani and Alvarez-Lorenzo have used hydrogels in the successful synthesis of soft contact lenses for the sustained delivery of drugs such as timolol and norfloxacin [118, 236, 237].

Biomimetic strategies, whereby an understanding of biological processes at the molecular level is exploited for use in synthetic systems [238], have been employed in the preparation of hydrogel based lenses and membranes [239, 240]. An example of this approach can be seen in the work of Ali *et al.* where hyaluronic acid, a polysaccharide used in the treatment of ocular discomfort and dry eye syndrome, was imprinted in soft contact lenses [241]. CD44 is a hyaluronic acid binding protein where the amino acids, asparagine, tyrosine and arginine are important for binding of the acid. The FDA approved monomers acrylamide, N-vinyl pyrrolidine and 2-(diethlyamino)ethyl methacrylate bear functional groups analogous to those present on the amino acids, as illustrated in Figure 1.23.



Figure 1.23: Comparison of the functional groups present in both FDA approved functional monomers and amino acids important for the binding of hyaluronic acid (adapted from reference [241]).

The use of a combination of the monomers shown in Figure 1.23 and careful optimisation of the polymer composition enabled the preparation of a contact lens capable of releasing hyaluronic acid at a therapeutic rate of 6 μ g/h for 24 hours. The same group has also investigated the use of 'living/controlled' polymerisation (LCP) strategies to enhance the loading and delay the release of drugs from typically weakly crosslinked gels and showed that the release could be extended over a two-fold time period by using LCP compared to conventional free-radical methods [242].

The use of molecularly imprinted microspherical hydrogel beads has also been an important step in the development of MIP based drug delivery systems. Micro- and nanospheres are particularly suited to drug delivery as they typically have a large surface area to volume ratio and an interior network capable of incorporating drugs and biomolecules [243]. They can have a high degree of uptake into cells as well as causing minimal irritation if administered subcutaneously [244, 245]. These attributes make them likely candidates for the targeted drug delivery proposed by Sellergren and Allender whereby the MIP releases its drug load when it binds to its target on the cell wall or is internalised into the cell [233]. Precipitation polymerisation strategies are well suited for use in drug delivery applications due to the bead size obtained and the fact that, due to incompatibility between the final polymer and the solvent, the amount of residual organic solvent is drastically decreased [246].

Sustained release of drug molecules under biological conditions has been achieved by the group of Puoci *et al.* for the prodrug sulfasalizine [170], and the antineoplastic agent, 5-fluorouracil [72] while Javanbakht *et al.* have demonstrated the release of diypyridamole, which acts to inhibit thrombus formation [247]. The structures of these examples are presented in Figure 1.24.



Figure 1.24: Drug templates used in the preparation of beads by precipitation polymerisation. I 5-fluorouracil; II sulfasalizine; III dipyridamole [72, 170, 247]

MAA-co-EGDMA polymers were used for *in vitro* release studies for 5-fluorouracil, where the same quantity of drug was reloaded onto both the MIP and the NIP. While all the drug was released from the NIP after 5 hours, the release of the drug from the MIP had still not been completed after 50 hours [72].

1.8 Scope of the thesis

This study will investigate the preparation of molecularly imprinted polymers selective for corticosteroids, with a particular focus on budesonide and prednisolone as target compounds for recognition. The preparation of corticosteroid specific polymers in bulk format, as well as novel precipitated microspheres and microparticles will be outlined. A systematic study that investigated the factors affecting precipitation polymerisation will be described and the importance of these results in the achievement of highly controllable polymerisation strategies will be examined and discussed. An NMR investigation into the interaction between commercially available functional monomers and the weakly functionalised corticosteroids will be discussed. These studies will show that such interactions were weak and the significance of these findings and correlations between pre-polymerisation studies and chromatographic evaluation of the resultant polymer will be critically examined.

The ability to optimise these polymeric systems will be shown by their demonstrated applicability for solid-phase extraction and chromatographic separation of close structural analogues. The application of knowledge related to particle optimisation will be highlighted for the latter example, where control over particle size will be shown to be crucial for improving chromatographic resolution.

Results from this study will highlight the need for strategies to enhance the strength and number of corticosteroid selective binding sites within the matrix of molecularly imprinted polymers. Two approaches to achieve this will be investigated. The successful preparation of tailor-made monomers capable of increased interaction with the template in aqueous systems will be outlined and the potential to use these polymers in the preparation of corticosteroid MIPs prepared in aqueous based porogens will be shown. A semi-covalent strategy, with the potential to increase the number of steroid specific binding sites within the polymeric matrix by direct polymerisation of a template-monomer will be discussed and the application of morphology control to prepare uniform microspherical polymers with the successfully derivatised steroidal template-monomer, will be demonstrated.
References

- 1. Alexander C, Andersson HS, Andersson LI, Ansell RJ, Kirsch N, Nicholls IA, O'Mahony J, Whitcombe MJ: Molecular imprinting science and technology: a survey of the literature for the years up to and including 2003. *Journal Of Molecular Recognition* 2006, **19**:106-180.
- 2. Mayes AG, Whitcombe MJ: Synthetic strategies for the generation of molecularly imprinted organic polymers. Advanced Drug Delivery Reviews 2005, 57:1742-1778.
- 3. Whitcombe MJ: **MIPDATABASE.COM.** 2011.
- 4. Andersson HS, Nicholls IA: **Chapter 1.** In *Molecularly Imprinted Polymers: Man made mimics of anitibodies and their applications*. Edited by Sellergren B. Amsterdam: Elsevier; 2001: *Techniques and Instrumentation in Analytical Chemistry*].
- 5. Polyakov MV: Adsorption Properties and structure of silica gel. *Zhurnal Fizieskoj Khimii* 1931, **2:**799-805.
- 6. Polyakov MV, Kuleshina L, Neimark I: On the dependence of silica gel adsorption properties on the character of its porosity. *Zhurnal Fizieskoj Khimii* 1937, **10**:100-112.
- 7. Polyakov MV, Stadnik P, Paryckij M, Malkin I, Duchina F: **On the structure of silica.** *Zhurnal Fizieskoj Khimii* 1933, **4:**454-456.
- 8. Dickey FH: **The preparation of specific adsorbents.** *Proceedings of the National Academy of Sciences* 1949, **35**:227-229.
- 9. Dickey FH: **Specific adsorption.** *Journal of Physical Chemistry* 1955, **59:**695-707.
- 10. Bernhard SA: The preparation of specific adsorbents. *Journal Of the American Chemical Society* 1952, **74:**4946-4947.
- 11. Haldeman RG, Emmet PH: **Specific adsorption of alkyl orange dyes on silica gel.** *Journal of Physical Chemistry* 1955, **59**:1039-1043.
- 12. Curti R, Colombo U: Chromatography of stereoisomers with "tailor made" compounds. *Journal Of the American Chemical Society* 1952, 74:3961-3961.
- 13. Morrison JL, Worsley M, Shaw DR, Hodgson GW: **The nature of the specificity of adsorption of alkyl orange dyes on silica gel.** *Canadian Journal of Chemistry-Revue Canadienne De Chimie* 1959, **37**:1986-1995.
- 14. Patrikeev VV, Balandin AA, Klabunovskii EI, Mardashev YS, Maksimova GI: Selectivity of an adsorbent produced in the presence of bacteria with respect to optical isomers. *Dokl Akad Nauk SSSR* 1960, **132:**850-852.
- 15. Beckett AH, Youssef HZ: Active sites in stereoselective adsorbents as models of drug receptors and enzyme active sites. *Journal of Pharmacy and Pharmacology* 1963, 15:253T-266T.
- 16. Wulff G, Sarhan A: The use of polymers with enzyme-analogous structures for the resolution of racemates. *Angewandte Chemie-International Edition* 1972, **11**:364-364.
- 17. Mosbach K, Arshady R: Synthesis of substrate-selective polymers by hostguest polymerization. *Macromolecular Chemistry and Physics* 1981, 182:687-692.
- 18. Mayes AG: Chapter 2. In *Molecularly imprinted materials: Science and Technology*. Edited by Ramstrom O, Yan M. New York: Marcel Dekker; 2005
- 19. Stevens MP: *Polymer Chemistry: An introduction.* 2nd edn. Oxford: Oxford University press; 1990.

- 20. Elias HG: An introduction to polymer science. Weinhein: VCH; 1997.
- 21. Lovell PA, Young RJ: *Introduction to polymers*. 2nd edn. London: Chapman and Hall; 1991.
- 22. Odian G: Principles of polymerisation. New York: McGraw-Hill; 1970.
- 23. Campbell IM: *Introduction to synthetic polymers*. Oxford: Oxford University press; 1994.
- 24. Carothers WH: *J Am Chem Soc* 1929, **51**:2548.
- 25. Flory PJ: *Principles of polymer chemistry*. New York: Cornell University press; 1953.
- Gellman SH: Introduction: Molecular Recognition. Chemical Reviews 1997, 97:1231-1232.
- 27. Lehn J-M: Supramolecular chemistry: Concepts and perspectives. Weinheim: VCH; 1995.
- 28. Neidle S: *DNA Structure and Recognition*. New York: Oxford University Press; 1994.
- 29. Palmer T: Understanding Enzymes. 3rd edn. Chichester: Ellis Horwood; 1991.
- 30. Fischer E: Ber Dtsch Chem Ges 1894, 27:2985-2993.
- 31. Birch JRL, E.S., Editors: *Monoclonal Antibodies: Principles and Applications*. New York: Wiley-Liss; 1995.
- 32. Pedersen CJ: Cyclic polyethers and their complexes with metal salts. *Journal* of the American Chemical Society 1967, **89:**7017-7036.
- 33. Vogtle F: Supramolecular Chemistry. Chichester: Wiley; 1991.
- 34. Pedersen CJ: Cyclic polyethers and their complexes with metal salts. *Journal* of the American Chemical Society 1967, **89:**2495-2496.
- 35. Dietrich B, Lehn JM, Sauvage JP: Diazapolyoxamacrocycles And Macrobicycles. *Tetrahedron Letters* 1969:2885-&.
- Cram DJ, Cram JM: Host guest chemistry. Complexes between organic compounds simultate the substrate selectivity of enzymes. *Science* 1974, 183:803-809.
- 37. Schneider H-J, Yatsimirsky A: *Principles and Methods in Supramolecular Chemistry*. Chichester: Wiley; 2000.
- 38. Kryscio DR, Peppas NA: Mimicking Biological Delivery Through Feed back-Controlled Drug Release Systems Based on Molecular Imprinting. *Aiche Journal* 2009, **55:**1311-1324.
- 39. Advincula RC: Engineering molecularly imprinted polymer (MIP) materials: Developments and challenges for sensing and separation technologies. *Korean Journal of Chemical Engineering* 2011, **28**:1313-1321.
- 40. Lorenzo RA, Carro AM, Alvarez-Lorenzo C, Concheiro A: **To Remove or Not to Remove? The Challenge of Extracting the Template to Make the Cavities Available in Molecularly Imprinted Polymers (MIPs).** *International Journal Of Molecular Sciences* 2011, **12:**4327-4347.
- 41. Yu C, Mosbach K: Influence of mobile phase composition and cross-linking density on the enantiomeric recognition properties of molecularly imprinted polymers. *Journal of Chromatography A* 2000, **888**:63-72.
- 42. Baggiani C, Baravalle P, Giovannoli C, Anfossi L, Giraudi G: Molecularly imprinted polymers for corticosteroids: Analysis of binding selectivity. *Biosensors and Bioelectronics* 2010, **26**:590-595.
- 43. Wulff G, Sarhan A, Zabrocki K: Enzyme-analogue built polymers and their use for the resolution of racemates. *Tetrahedron Letters* 1973, **14**:4329-4322.

- 44. Willner I, Granot E, Tel-Vered R, Lioubashevski O: Stereoselective and Enantioselective Electrochemical Sensing of Monosaccharides Using Imprinted Boronic Acid-Functionalised Polyphenol Films. Advanced Functional Materials 2008, 18:478-484.
- 45. Schumacher S, Gruneberger F, Katterle M, Hettrich C, Hall DG, Scheller FW, Gajovic-Eichelmann N: Molecular imprinting of fructose using a polymerizable benzoboroxole: Effective complexation at pH 7.4. *Polymer* 2011, 52:2485-2491.
- 46. Wulff G, Vietmeier J: Enzyme-analogue built polymer. 26. Enantioselective synthesis of amino-acids using polymers possessing chiral cavities obtained by an imprinting procedure with template molecules. *Macromolecular Chemistry and Physics* 1989, 190:1727-1735.
- 47. Shea KJ, Dougherty TK: Molecular recognition on synthetic amorphous surfaces the influence of functional-group positioning on the effectiveness of molecular recognition. *Journal Of the American Chemical Society* 1986, 108:1091-1093.
- 48. Sasaki S, Ooya T, Takeuchi T: Highly selective bisphenol A-imprinted polymers prepared by atom transfer radical polymerization. *Polymer Chemistry* 2011, 1:1684-1688.
- 49. Remcho VT, Vallano PT, Khasawneh MA: Affinity screening by packed capillary high performance liquid chromatography using molecular imprinted sorbents II. Covalent imprinted polymers. *Journal of Chromatography A* 2001, **922**:97-97.
- 50. Hwang CC, Lee WC: Chromatographic characteristics of cholesterolimprinted polymers prepared by covalent and non-covalent imprinting methods. *Journal of Chromatography A* 2002, **962**:69-78.
- 51. Kugimiya A, Matsui J, Abe H, Aburatani M, Takeuchi T: Synthesis of castasterone selective polymers prepared by molecular imprinting. *Analytica Chimica Acta* 1998, **365:**75-79.
- 52. Sellergren B: Chapter 5: The non-covalent approach to molecular imprinting. In *Molecularly Imprinted Polymers: Man-Made Mimics of Antibodies and their Applications in Analytical Chemistry. Volume* 23. Edited by Sellergren B. Amsterdam: Elsevier; 2001
- 53. Miller EI, Anderson RA, Ariffin MM, Cormack PAG: Comparison of molecularly imprinted solid-phase extraction (MISPE) with classical solid-phase extraction (SPE) for the detection of benzodiazepines in post-mortem hair samples. *Forensic Science International* 2008, 174:40-46.
- 54. Mayes AG, Pérez-Moral N: Direct rapid synthesis of MIP beads in SPE cartridges. *Biosensors and Bioelectronics* 2006, **21**:1798-1803.
- 55. Lin HY, Rick J, Chou TC: **Optimizing the formulation of a myoglobin** molecularly imprinted thin-film polymer-formed using a micro-contact imprinting method. *Biosensors & Bioelectronics* 2007, **22**:3293-3301.
- 56. Barahona F, Turiel E, Cormack PAG, Martin-Esteban A: Synthesis of coreshell molecularly imprinted polymer microspheres by precipitation polymerization for the inline molecularly imprinted solid-phase extraction of thiabendazole from citrus fruits and orange juice samples. *Journal of Separation Science* 2011, 34:217-224.
- 57. Tang K, Chen S, Gu X, Wang H, Dai J, Tang J: Preparation of molecularly imprinted solid phase extraction using bensulfuron-methyl imprinted

polymer and clean-up for the sulfonylurea-herbicides in soybean. *Analytica Chimica Acta* 2008, **614:**112.

- 58. Malosse L, Buvat P, Ades D, Siove A: Detection of degradation products of chemical warfare agents by highly porous molecularly imprinted microspheres. *Analyst* 2008, **133:**588-595.
- 59. Mosbach K, O'Shannesy DJ, Ekberg B: Molecular imprinting of amino-acid derivatives at low-temperature (0 °C) using photolytic homolysis of azobisnitriles. *Analytical Biochemistry* 1989, 177:144-149.
- 60. Monier M, Ayad DM, Wei Y, Sarhan AA: **Preparation of cross-linked** chitosan/glyoxal molecularly imprinted resin for efficient chiral resolution of aspartic acid isomers. *Biochemical Engineering Journal* 2011, **51**:140-146.
- 61. Zhu Q, Wang L, Wu S, Joseph W, Gu X, Tang J: Selectivity of molecularly imprinted solid phase extraction for sterol compounds. *Food Chemistry* 2009, 113:608.
- 62. Wei ST, Molinelli A, Mizaikoff B: Molecularly imprinted micro and nanospheres for the selective recognition of 17 beta-estradiol. *Biosensors & Bioelectronics* 2006, **21**:1943-1951.
- 63. Ramstrom O, Ye L, Mosbach K: Artificial antibodies to corticosteroids prepared by molecular imprinting. *Chemistry & Biology* 1996, **3**:471-477.
- 64. Baggiani C, Giraudi G, Trotta F, Giovannoli C, Vanni A: Chromatographic characterization of a molecular imprinted polymer binding cortisol. *Talanta* 2000, **51**:71-75.
- 65. Jiang JB, Song KS, Chen Z, Zhou Q, Tang YW, Gu FL, Zuo XJ, Xu ZG: Novel molecularly imprinted microsphere using a single chiral monomer and chirality-matching (S)-ketoprofen template. *Journal of Chromatography A* 2011, **1218**:3763-3770.
- 66. Mosbach K, Haupt K, Xiao-Chuan Liu, Cormack PAG, Ramstrom O: Chapter 3: Molecular Imprinting Status Artis et Quo Vadere? In Molecular and Ionic Recognition with Imprinted Polymers. Edited by Bartsch RA, Maeda M. New York: Oxford University Press; 1998: ACS symposium series].
- 67. Wulff G, Knorr K: Stoichiometric noncovalent interaction in molecular imprinting. *Bioseparation* 2001, **10**:257-276.
- 68. Sellergren B, Andersson L: Molecular recognition in macroporous polymers prepared by a substrate analog imprinting strategy. *The Journal of Organic Chemistry* 1990, **55**:3381-3383.
- 69. Cacho C, Turiel E, Martín-Esteban A, Ayala D, Pérez-Conde C: Semi-covalent imprinted polymer using propazine methacrylate as template molecule for the clean-up of triazines in soil and vegetable samples. *Journal of Chromatography A* 2006, 1114:255.
- 70. Cacho C, Turiel E, Martin-Esteban A, Pérez-Conde C, Cámara C: Characterisation and quality assessment of binding sites on a propazineimprinted polymer prepared by precipitation polymerisation. *Journal of Chromatography B* 2004, **802:**347.
- 71. Curcio P, Zandanel C, Wagner A, Mioskowski C, Baati R: Semi-Covalent Surface Molecular Imprinting of Polymers by One-Stage Mini-emulsion Polymerization: Glucopyranoside as a Model Analyte. Macromolecular Bioscience 2009, 9:596-604.
- 72. Cirillo G, Iemma F, Puoci F, Parisi OI, Curcio M, Spizzirri UG, Picci N: Imprinted hydrophilic nanospheres as drug delivery systems for 5fluorouracil sustained release. *Journal of Drug Targeting* 2009, **17:**72 - 77.

- 73. Alexiadou DK, Maragou NC, Thomaidis NS, Theodoridis GA, Koupparis MA: Molecularly imprinted polymers for bisphenol A for HPLC and SPE from water and milk. *Journal Of Separation Science* 2008, **31**:2272-2282.
- 74. Cheong SH, McNiven S, Rachkov A, Levi R, Yano K, Karube I: **Testosterone Receptor Binding Mimic Constructed Using Molecular Imprinting.** *Macromolecules* 1997, **30**:1317-1322.
- 75. Caro E, Masque M, Marce RM, Borrull F, Cormack PAG, Sherrington DC: Non-covalent and semi-covalent molecularly imprinted polymers for selective on-line solid-phase extraction of 4-nitrophenol from water samples. *Journal Of Chromatography A* 2002, 963:169-178.
- 76. Whitcombe MJ, Rodriguez ME, Villar P, Vulfson EN: A New Method for the Introduction of Recognition Site Functionality into Polymers Prepared by Molecular Imprinting: Synthesis and Characterization of Polymeric Receptors for Cholesterol. Journal of the American Chemical Society 1995, 117:7105-7111.
- 77. Tong YJ, Guan HM, Wang SF, Xu JW, He CB: Syntheses of chitin-based imprinting polymers and their binding properties for cholesterol. *Carbohydrate Research* 2011, **346**:495-500.
- 78. Kirsch N, Alexander C, Davies S, Whitcombe MJ: Sacrificial spacer and noncovalent routes toward the molecular imprinting of "poorly-functionalized" N-heterocycles. Analytica Chimica Acta 2004, 504:63.
- 79. He CY, Long YY, Pan JL, Li K, Liu F: Molecularly imprinted silica prepared with immiscible ionic liquid as solvent and porogen for selective recognition of testosterone. *Talanta* 2008, **74:**1126-1131.
- 80. Beltran A, Marce RM, Cormack PAG, Borrull F: Synthetic approaches to parabens molecularly imprinted polymers and their applications to the solid-phase extraction of river water samples. *Analytica Chimica Acta* 2011, 677:72-78.
- 81. Rosengren-Holmberg JP, Karlsson JG, Svenson J, Andersson HS, Nicholls IA: Synthesis and ligand recognition of paracetamol selective polymers: semicovalent versus non-covalent molecular imprinting. Organic & Biomolecular Chemistry 2009, 7:3148-3155.
- 82. Wulff G, Schonfeld R: Polymerizable amidines Adhesion mediators and binding sites for molecular imprinting. *Advanced Materials* 1998, **10**:957-+.
- 83. Liu J-q, Wulff Gn: Functional Mimicry of Carboxypeptidase A by a Combination of Transition State Stabilization and a Defined Orientation of Catalytic Moieties in Molecularly Imprinted Polymers. *Journal of the American Chemical Society* 2008, **130**:8044-8054.
- 84. Kim JM, Ahn KD, Wulff G: Cholesterol esterase activity of a molecularly imprinted polymer. *Macromolecular Chemistry And Physics* 2001, 202:1105-1108.
- 85. Pasetto P, Flavin K, Resmini M: Simple spectroscopic method for titration of binding sites in molecularly imprinted nanogels with hydrolase activity. *Biosensors and Bioelectronics* 2009, **25**:572-578.
- 86. Wang GS, Cao Q, Zhu XF, Yang XQ, Yang MH, Ding ZT: Molecular Imprinted Solid-Phase Extraction of Huperzine A from Huperzia serrata. Journal of Applied Polymer Science 2009, 113:3049-3058.
- 87. Karim K, Breton F, Rouillon R, Piletska EV, Guerreiro A, Chianella I, Piletsky SA: How to find effective functional monomers for effective molecularly imprinted polymers? *Advanced Drug Delivery Reviews* 2005, **57**:1795.

- 88. Cormack PAG, Elorza AZ: Molecularly imprinted polymers: synthesis and characterisation. *Journal of Chromatography B-Analytical Technologies in the Biomedical and Life Sciences* 2004, **804**:173-182.
- 89. Mohajeri SA, Karimi G, Aghamohammadian J, Khansari MR: Clozapine Recognition via Molecularly Imprinted Polymers; Bulk Polymerization Versus Precipitation Method. *Journal of Applied Polymer Science* 2011, 121:3590-3595.
- 90. Gadzala-Kopciuch R, Ricanyova J, Buszewskia B: Isolation and detection of steroids from human urine by molecularly imprinted solid-phase extraction and liquid chromatography. *Journal Of Chromatography B-Analytical Technologies In The Biomedical And Life Sciences* 2009, **877:**1177-1184.
- 91. Kan XW, Geng ZR, Zhao Y, Wang ZL, Zhu JJ: Magnetic molecularly imprinted polymer for aspirin recognition and controlled release. *Nanotechnology* 2009, **20**.
- 92. Wang B, Wang YZ, Yang H, Wang JQ, Deng AP: **Preparation and** characterization of molecularly imprinted microspheres for selective extraction of trace melamine from milk samples. *Microchimica Acta* 2011, 174:191-199.
- 93. Watabe Y, Kubo T, Nishikawa T, Fujita T, Kaya K, Hosoya K: Fully automated liquid chromatography-mass spectrometry determination of 17 beta-estradiol in river water. *Journal of Chromatography A* 2006, 1120:252-259.
- 94. Kamel AH, Moreira FTC, Almeida SAA, Sales MGF: Novel potentiometric sensors of molecular imprinted polymers for specific binding of chlormequat. *Electroanalysis* 2008, **20**:194-202.
- 95. Gam-Derouich S, Nguyen MN, Madani A, Maouche N, Lang P, Perruchot C, Chehimi MM: Aryl diazonium salt surface chemistry and ATRP for the preparation of molecularly imprinted polymer grafts on gold substrates. Surface and Interface Analysis 2011, 42:1050-1056.
- 96. Long C, Mai Z, Yang Y, Zhu B, Xu X, Lu L, Zou X: Synthesis and characterization of a novel molecularly imprinted polymer for simultaneous extraction and determination of water-soluble and fat-soluble synthetic colorants in chilli products by solid phase extraction and high performance liquid chromatography. *Journal of Chromatography A* 2009, **1216**:8379-8385.
- 97. Sreenivasan K: Application of molecularly imprinted polymer as a drug retaining matrix. *Angewandte makromolekulare Chemie* 1997, **246:**65-69.
- 98. Puoci F, Iemma F, Cirillo G, Trombino S, Cassano R, Picci N: Molecularly imprinted polymers for selective adsorption of cholesterol from aqueous environment. *E-Polymers* 2007:-.
- 99. Cirillo G, Curcio M, Parisi OI, Puoci F, Iemma F, Spizzirri UG, Restuccia D, Picci N: Molecularly imprinted polymers for the selective extraction of glycyrrhizic acid from liquorice roots. *Food Chemistry* 2011, 125:1058-1063.
- 100. Janiak DS, Ayyub OB, Kofinas P: Effects of Charge Density on the Recognition Properties of Molecularly Imprinted Polymeric Hydrogels. *Macromolecules* 2009, **42**:1703-1709.
- 101. Chang YS, Ko TH, Hsu TJ, Syu MJ: Synthesis of an Imprinted Hybrid Organic-Inorganic Polymeric Sol-Gel Matrix Toward the Specific Binding and Isotherm Kinetics Investigation of Creatinine. *Analytical Chemistry* 2009, 81:2098-2105.

- 102. Liao SL, Wang XC, Lin XC, Xie ZH: Preparation and characterization of a molecularly imprinted monolithic column for pressure-assisted CEC separation of nitroimidazole drugs. *Electrophoresis* 2010, **31**:2822-2830.
- 103. Janiak DS, Ayyub OB, Kofinas P: Effects of charge density on the recognition properties of molecularly imprinted polyampholyte hydrogels. *Polymer* 2010, **51:**665-670.
- 104. Spivak DA: **Optimization, evaluation, and characterization of molecularly imprinted polymers.** *Advanced Drug Delivery Reviews* 2005, **57**:1779.
- 105. Batra D, Shea KJ: Combinatorial methods in molecular imprinting. *Current Opinion in Chemical Biology* 2003, 7:434.
- 106. Osmani Q, Hughes H, Flavin K, Hedin-Dahlstrom J, Allender CJ, Frisby J, McLoughlin P: The use of FTIR and NMR spectroscopies to study prepolymerisation interactions in nitrogen heterocycles. *Analytical and Bioanalytical Chemistry* 2008, **391**:1229-1236.
- 107. Mizaikoff B, Wei S, Jakusch M: Capturing molecules with templated materials Analysis and rational design of molecularly imprinted polymers. *Analytica Chimica Acta* 2006, **578:**50-58.
- 108. Sellergren B, Shea KJ: Influence Of Polymer Morphology On The Ability Of Imprinted Network Polymers To Resolve Enantiomers. *Journal Of Chromatography* 1993, 635:31-49.
- 109. Nomachi M, Kubo T, Hosoya K, Kaya K: Solvent effects in the preparation of molecularly imprinted polymers for melatonin using N-propionyl-5methoxytryptamine as the pseudo template. *Analytical & Bioanalytical Chemistry* 2006, **384**:1291.
- 110. Sellergren B, Hall AJ: Chapter 2. In Molecularly Imprinted Polymers: Man Made Mimics of Antibodies and their Applications in analytical chemistry. Volume 23. Edited by Sellergren B. Amsterdam: Elsevier; 2001: Techniques and Instrumentation in Analytical Chemistry].
- 111. Spivak D, Gilmore MA, Shea KJ: Evaluation of binding and origins of specificity of 9-ethyladenine imprinted polymers. *Journal Of the American Chemical Society* 1997, **119:**4388-4393.
- Spivak DA, Shea KJ: Investigation into the scope and limitations of molecular imprinting with DNA molecules. *Analytica Chimica Acta* 2001, 435:65-74.
- 113. Haginaka J: Monodispersed, molecularly imprinted polymers as affinitybased chromatography media. *Journal of Chromatography B* 2008, 866:3.
- 114. Liang S, Wan JF, Zhu JL, Cao XJ: Effects of porogens on the morphology and enantioselectivity of core-shell molecularly imprinted polymers with ursodeoxycholic acid. Separation and Purification Technology 2010, 72:208-216.
- 115. Wulff G, Vietmeier J, Poll HG: Enzyme-Analog Built Polymers .22. Influence Of The Nature Of The Cross-Linking Agent On The Performance Of Imprinted Polymers In Racemic-Resolution. Makromolekulare Chemie-Macromolecular Chemistry And Physics 1987, 188:731-740.
- 116. Sellergren B: Molecular Imprinting By Noncovalent Interactions -Enantioselectivity And Binding-Capacity Of Polymers Prepared Under Conditions Favoring The Formation Of Template Complexes. Makromolekulare Chemie-Macromolecular Chemistry And Physics 1989, 190:2703-2711.

- 117. Holland N, Duggan P, Owens E, Cummins W, Frisby J, Hughes H, McLoughlin P: Thermal desorption characterisation of molecularly imprinted polymers. Part II: Use of direct probe GC-MS analysis to study crosslinking effects. Analytical And Bioanalytical Chemistry 2008, 391:1245-1253.
- 118. Alvarez-Lorenzo C, Hiratani H, Gomez-Amoza JL, Martinez-Pacheco R, Souto C, Concheiro A: Soft contact lenses capable of sustained delivery of timolol. *Journal of Pharmaceutical Sciences* 2002, **91**:2182-2192.
- 119. Hillberg AL, Brain KR, Allender CJ: Design and evaluation of thin and flexible theophylline imprinted polymer membrane materials. *Journal Of Molecular Recognition* 2009, **22**:223-231.
- 120. Kempe M, Mosbach K: Receptor binding mimetics: A novel molecularly imprinted polymer. *Tetrahedron Letters* 1995, **36**:3563.
- 121. Ye L, Weiss R, Mosbach K: Synthesis and characterization of molecularly imprinted microspheres. *Macromolecules* 2000, 33:8239-8245.
- 122. Suedee R, Seechamnanturakit V, Canyuk B, Ovatiarnporn C, Martin GP: Temperature sensitive dopamine-imprinted (N,N-methylene-bis-acrylamide cross-linked) polymer and its potential application to the selective extraction of adrenergic drugs from urine. *Journal Of Chromatography A* 2006, 1114:239-249.
- 123. Cirillo G, Curcio M, Parisi OI, Puoci F, Iemma F, Spizzirri UG, Picci N: Gastro-intestinal sustained release of phytic acid by molecularly imprinted microparticles. *Pharmaceutical Development And Technology* 2010, 15:526-531.
- 124. Ates Z, Güven O: Radiation-induced molecular imprinting of d-glucose onto poly(2-hydroxyethyl methacrylate) matrices using various crosslinking agents. *Radiation Physics and Chemistry* 2010, **79:**219-222.
- 125. Papaioannou E, Koutsas C, Liakopoulou-Kyriakides M: Molecularly imprinted polymers for RGD selective recognition and separation. *Amino Acids* 2009, **36:**563-569.
- 126. Gavrilovic I, Mitchell K, Brailsford AD, Cowan DA, Kicman AT, Ansell RJ: A molecularly imprinted receptor for separation of testosterone and epitestosterone, based on a steroidal cross-linker. *Steroids* 2011, **76**:478-483.
- 127. Sreenivasan K: The effect of polymerisation methods on the adsorption capacity of HEMA based molecularly imprinted polymers. *Journal of Polymer Research-Taiwan* 2001, **8**:197-200.
- 128. Qin L, He XW, Yuan X, Li WY, Zhang YK: Molecularly imprinted beads with double thermosensitive gates for selective recognition of proteins. *Analytical and Bioanalytical Chemistry* 2011, **399:**3375-3385.
- 129. Li JY, Zu BY, Zhang Y, Guo XZ, Zhang HQ: **One-Pot Synthesis of Surface-Functionalized Molecularly Imprinted Polymer Microspheres by Iniferter-Induced ''Living'' Radical Precipitation Polymerization.** Journal of Polymer Science Part a-Polymer Chemistry 2010, **48**:3217-3228.
- 130. Rebelo TnSCR, Almeida SAA, Guerreiro JRL, Montenegro MCÃoBSM, Sales MGF: Trimethoprim-selective electrodes with molecularly imprinted polymers acting as ionophores and potentiometric transduction on graphite solid-contact. *Microchemical Journal* 2011, **98**:21-28.
- 131. Gong X-Y, Cao X-J: Preparation of molecularly imprinted polymers for artemisinin based on the surfaces of silica gel. *Journal of Biotechnology* 2011, 153:8-14.

- 132. Pascale M, De Girolamo A, Visconti A, Magan N, Chianella I, Piletska EV, Piletsky SA: Use of itaconic acid-based polymers for solid-phase extraction of deoxynivalenol and application to pasta analysis. *Analytica Chimica Acta* 2008, 609:131-138.
- 133. Pérez-Moral N, Mayes AG: Comparative study of imprinted polymer particles prepared by different polymerisation methods. *Analytica Chimica Acta* 2004, **504:**15.
- 134. Zhang ZL, Wang B, Li JA: Effect of the Synthesis Initiation Mode on the Structure and Properties of Sulfadiazine Molecularly Imprinted Polymers. Journal of Applied Polymer Science 2011, 119:3189-3198.
- 135. Dopico-Garcia MS, Cela-Perez C, Lopez-Vilarino JM, Gonzalez-Rodriguez MV, Barral-Losada LF: An Approach to Imprint Irganox 1076: Potential Application to the Specific Migration Test in Olive Oil. Journal of Applied Polymer Science 2010, 119:2866-2874.
- 136. Lu Y, Li C, Wang X, Sun P, Xing X: Influence of polymerization temperature on the molecular recognition of imprinted polymers. *Journal of Chromatography B* 2004, **804:**53-59.
- 137. O'Mahony J, Molinelli A, Nolan K, Smyth MR, Mizaikoff B: Towards the rational development of molecularly imprinted polymers: 1H NMR studies on hydrophobicity and ion-pair interactions as driving forces for selectivity. *Biosensors and Bioelectronics* 2005, 20:1884-1893.
- 138. Nicholls IA: Thermodynamic Considerations for the Design of and Ligand Recognition by Molecularly Imprinted Polymers. *Chemistry Letters* 1995:1035-1036.
- 139. Idziak I, Benrebouh A, Deschamps F: Simple NMR experiments as a means to predict the performance of an anti-17[alpha]-ethynylestradiol molecularly imprinted polymer. *Analytica Chimica Acta* 2001, **435**:137.
- 140. Urraca JL, Carbajo MC, Torralvo MJ, González-Vázquez J, Orellana G, Moreno-Bondi MC: Effect of the template and functional monomer on the textural properties of molecularly imprinted polymers. *Biosensors and Bioelectronics* 2008, 24:155.
- 141. Ansell RJ, Wang DY: Imprinted polymers for chiral resolution of (+/-)ephedrine. Part 3: NMR predictions and HPLC results with alternative functional monomers. *Analyst* 2009, **134**:564-576.
- 142. Takeuchi T, Dobashi A, Kimura K: Molecular Imprinting of Biotin Derivatives and Its Application to Competitive Binding Assay Using Nonisotopic Labeled Ligands. Anal Chem 2000, 72:2418-2422.
- 143. Pavel D, Lagowski J: Computationally designed monomers and polymers for molecular imprinting of theophylline and its derivatives. Part I. *Polymer* 2005, **46**:7528.
- 144. Isarankura-Na-Ayudhya C, Nantasenamat C, Buraparuangsang P, Piacham T, Ye L, Bulow L, Prachayasittikul V: Computational Insights on Sulfonamide Imprinted Polymers. *Molecules* 2008, 13:3077-3091.
- 145. Díaz-Díaz G, Diñeiro Y, Menéndez MI, Blanco-López MC, Lobo-Castañón MJ, Miranda-Ordieres AJ, Tuñón-Blanco P: Molecularly imprinted catalytic polymers with biomimetic chloroperoxidase activity. *Polymer* 2010, 52:2468-2473.
- 146. Li XF, Zhong SA, Chen L, Whittaker A: Computer simulation and preparation of molecularly imprinted polymer membranes with chlorogenic acid as template. *Polymer International* 2011, **60**:592-598.

- 147. Mojica E-RE, Autschbach J, Bright FV, Aga DS: Synthesis and evaluation of tetracycline imprinted xerogels: Comparison of experiment and computational modeling. *Analytica Chimica Acta* 2011, 684:72-80.
- 148. Chianella I, Karim K, Piletska EV, Preston C, Piletsky SA: Computational design and synthesis of molecularly imprinted polymers with high binding capacity for pharmaceutical applications-model case: Adsorbent for abacavir. *Analytica Chimica Acta* 2006, **559**:73.
- 149. Piletska E, Karim K, Coker R, Piletsky S: **Development of the custom** polymeric materials specific for aflatoxin B1 and ochratoxin A for application with the ToxiQuant T1 sensor tool. *Journal of Chromatography A* 2010, **1217**:2543-2547.
- 150. Yañez F, Chianella I, Piletsky SA, Concheiro A, Alvarez-Lorenzo C: Computational modeling and molecular imprinting for the development of acrylic polymers with high affinity for bile salts. *Analytica Chimica Acta* 2010, **659**:178-185.
- 151. Sergeyeva TA, Piletska OV, Piletsky SA, Sergeeva LM, Brovko OO, El'ska GV: Data on the structure and recognition properties of the template-selective binding sites in semi-IPN-based molecularly imprinted polymer membranes. Materials Science and Engineering: C 2008, 28:1472.
- 152. Pearlman DA, Case DA, Caldwell JW, Ross WS, Cheatham TE, DeBolt S, Ferguson D, Seibel G, Kollman P: AMBER, a package of computer programs for applying molecular mechanics, normal mode analysis, molecular dynamics and free energy calculations to simulate the structural and energetic properties of molecules. *Computer Physics Communications* 1995, 91:1.
- 153. Shutting W, Jakusch M, Mizaikoff B: Investigating the mechanisms of 17 Î²estradiol imprinting by computational prediction and spectroscopic analysis. *Analytical & Bioanalytical Chemistry* 2007, **389**:423.
- 154. Wei S, Molinelli A, Mizaikoff B: Molecularly imprinted micro and nanospheres for the selective recognition of 17[beta]-estradiol. *Biosensors and Bioelectronics* 2006, 21:1943.
- 155. Karlsson BCG, O'Mahony J, Karlsson JG, Bengtsson H, Eriksson LA, Nicholls IA: Structure and Dynamics of Monomer-Template Complexation: An Explanation for Molecularly Imprinted Polymer Recognition Site Heterogeneity. Journal Of the American Chemical Society 2009, 131:13297-13304.
- 156. Karlsson BCG, Rosengren AM, Naslund I, Andersson PO, Nicholls IA: Synthetic Human Serum Albumin Sudlow I Binding Site Mimics. Journal of Medicinal Chemistry 2010, 53:7932-7937.
- 157. Wiklander J, Karlsson BCG, Aastrup T, Nicholls IA: **Towards a synthetic** avidin mimic. *Analytical and Bioanalytical Chemistry* 2011, **400**:1397-1404.
- 158. Connors KA: Binding Constants. New York: Wiley; 1987.
- 159. Atwood JL: Comprehensive Supramolecular Chemistry. 1st edn. Amsterdam: Elsevier; 1996.
- 160. Huang CY: Determination Of Binding Stoichiometry By The Continuous Variation Method The Job Plot. *Methods In Enzymology* 1982, 87:509-525.
- 161. Fielding L: Determination of Association Constants (Ka) from Solution NMR Data. *Tetrahedron* 2000, 56:6151.
- 162. Sellergren B, Lepistoe M, Mosbach K: Highly enantioselective and substrateselective polymers obtained by molecular imprinting utilizing noncovalent

interactions. NMR and chromatographic studies on the nature of recognition. J Am Chem Soc 1988, 110:5853-5860.

- 163. Andersson HS, Nicholls IA: Spectroscopic Evaluation of Molecular Imprinting Polymerization Systems. *Bioorganic Chemistry* 1997, 25:203.
- 164. Ansell RJ, Kuah KL: Imprinted polymers for chiral resolution of (+/-)ephedrine: understanding the pre-polymerisation equilibrium and the action of different mobile phase modifiers. *Analyst* 2005, **130**:179-187.
- 165. Ansell RJ, Wang DY, Kuah JKL: Imprinted polymers for chiral resolution of (+/-)-ephedrine. Part 2: probing pre-polymerisation equilibria in different solvents by NMR. *Analyst* 2008, **133**:1673-1683.
- 166. Zhang YG, Song D, Lanni LM, Shimizu KD: Importance of Functional Monomer Dimerization in the Molecular Imprinting Process. Macromolecules 2010, 43:6284-6294.
- 167. Zhang YG, Song D, Brown JC, Shimizu KD: Suppression of background sites in molecularly imprinted polymers via urea-urea monomer aggregation. *Organic & Biomolecular Chemistry* 2010, **9:**120-126.
- 168. Urraca JL, Aureliano CSA, Schillinger E, Esselmann H, Wiltfang J, Sellergren B: Polymeric Complements to the Alzheimer's Disease Biomarker beta-Amyloid Isoforms A beta 1-40 and A beta 1-42 for Blood Serum Analysis under Denaturing Conditions. *Journal Of the American Chemical Society* 2011, 133:9220-9223.
- 169. Yoshimatsu K, Reimhult K, Krozer A, Mosbach K, Sode K, Ye L: Uniform molecularly imprinted microspheres and nanoparticles prepared by precipitation polymerization: The control of particle size suitable for different analytical applications. *Analytica Chimica Acta* 2007, **584**:112.
- 170. Puoci F, Iemma E, Muzzalupo R, Spizzirri UG, Trombino S, Cassano R, Picci N: Spherical molecularly imprinted polymers (SMIPs) via a novel precipitation polymerization in the controlled delivery of sulfasalazine. *Macromolecular Bioscience* 2004, **4**:22-26.
- 171. Mayes AG: Chapter 12. In Molecularly Imprinted Polymers: Man Made Mimics of Antibodies and their Applications in Analytical Chemistry. Volume 23. Edited by Sellergren B. Amsterdam: Elsevier; 2001: Techniques and Instrumentation in Analytical Chemistry].
- 172. Mosbach K, Mayes AG: Molecularly Imprinted Polymer Beads: Suspension Polymerisation Using a Liquid Perflourcarbon as the Dispersing Phase Anal Chem 1996, 68:3769-3774.
- 173. Kempe M, Kempe H: Novel Method for the Synthesis of Molecularly Imprinted Polymer Bead Libraries. *Macromolecular Rapid Communications* 2004, 25:315-320.
- 174. Abbate V, Frascione N, Bansal SS: Preparation, Characterization, and Binding Profile of Molecularly Imprinted Hydrogels for the Peptide Hepcidin. Journal of Polymer Science Part a-Polymer Chemistry 2010, 48:1721-1731.
- 175. Baggiani C, Anfossi L, Baravalle P, Giovannoli C, Giraudi G, Barolo C, Viscardi G: Determination of banned Sudan dyes in food samples by molecularly imprinted solid phase extraction-high performance liquid chromatography. Journal of Separation Science 2009, **32**:3292-3300.
- 176. Li K, Stover HDH: Synthesis of Monodisperse Poly(Divinylbenzene) Microspheres. Journal of Polymer Science Part a-Polymer Chemistry 1993, 31:3257-3263.

- 177. Naka Y, Kaetsu I, Yamamoto Y, Hayashi K: Preparation of Microspheres by Radiation-Induced Polymerization .1. Mechanism for the Formation of Monodisperse Poly(Diethylene Glycol Dimethacrylate) Microspheres. Journal of Polymer Science Part a-Polymer Chemistry 1991, 29:1197-1202.
- Ye L, Cormack PAG, Mosbach K: Molecularly imprinted monodisperse microspheres for competitive radioassay. *Analytical Communications* 1999, 36:35-38.
- 179. Esfandyari-Manesh M, Javanbakht M, Atyabi F, Badiei A, Dinarvand R: Effect of Porogenic Solvent on the Morphology, Recognition and Release Properties of Carbamazepine-Molecularly Imprinted Polymer Nanospheres. Journal of Applied Polymer Science 2011, 121:1118-1126.
- Liu Y, Hoshina K, Haginaka J: Monodispersed, molecularly imprinted polymers for cinchonidine by precipitation polymerization. *Talanta* 2010, 80:1713-1718.
- 181. Long YY, Philip JYN, Schillen K, Liu F, Ye L: Insight into molecular imprinting in precipitation polymerization systems using solution NMR and dynamic light scattering. *Journal Of Molecular Recognition* 2011, **24:**619-630.
- 182. Carter S, Lu SY, Rimmer S: Core-shell molecular imprinted polymer colloids. *Supramolecular Chemistry* 2003, 15:213-220.
- 183. Yu KY, Tsukagoshi K, Maeda M, Takagi M: METAL ION-IMPRINTED MICROSPHERES PREPARED BY REORGANIZATION OF THE COORDINATING GROUPS ON THE SURFACE. Analytical Sciences 1992, 8:701-703.
- 184. Perez N, Whitcombe MJ, Vulfson EN: Molecularly imprinted nanoparticles prepared by core-shell emulsion polymerization. *Journal of Applied Polymer Science* 2000, **77**:1851-1859.
- Perez N, Whitcombe MJ, Vulfson EN: Surface Imprinting of Cholesterol on Submicrometer Coreâ[^]Shell Emulsion Particles. *Macromolecules* 2001, 34:830-836.
- 186. Carter SR, Rimmer S: Molecular recognition of caffeine by shell molecular imprinted core-shell polymer particles in aqueous media. Advanced Materials 2002, 14:667-670.
- 187. Ziyun X, Junfen W, Shuang L, Xuejun C: Separation of ursodeoxycholic acid from its isomeric mixture using core-shell molecular imprinting polymer. *Biochemical Engineering Journal* 2008, 41:280-287.
- 188. Ma J, Yuan L, Ding M, Wang S, Ren F, Zhang J, Du S, Li F, Zhou X: The study of core-shell molecularly imprinted polymers of 17[beta]-estradiol on the surface of silica nanoparticles. *Biosensors and Bioelectronics* 2011, 26:2791-2795.
- 189. Li W-H, Stöver HDH: Monodisperse Cross-Linked Coreâ[^]Shell Polymer Microspheres by Precipitation Polymerization. *Macromolecules* 2000, 33:4354-4360.
- 190. Barahona F, Turiel E, Cormack PAG, Martin-Esteban A: Chromatographic Performance of Molecularly Imprinted Polymers: Core-Shell Microspheres by Precipitation Polymerization and Grafted MIP Films via Iniferter-Modified Silica Beads. Journal of Polymer Science Part a-Polymer Chemistry 2010, 48:1058-1066.
- 191. Huang CY, Syu MJ, Chang YS, Chang CH, Chou TC, Liu BD: A portable potentiostat for the bilirubin-specific sensor prepared from molecular imprinting. *Biosensors & Bioelectronics* 2007, **22**:1694-1699.

- 192. Liu YT, Deng J, Xiao XL, Ding L, Yuan YL, Li H, Li XT, Yan XN, Wang LL: Electrochemical sensor based on a poly(para-aminobenzoic acid) film modified glassy carbon electrode for the determination of melamine in milk. Electrochimica Acta 2011, 56:4595-4602.
- 193. Mizutani N, Yang D-H, Selyanchyn R, Korposh S, Lee S-W, Kunitake T: Remarkable enantioselectivity of molecularly imprinted TiO2 nano-thin films. *Analytica Chimica Acta* 2011, **694:**142-150.
- 194. Atta NF, Abdel-Mageed AM: Smart electrochemical sensor for some neurotransmitters using imprinted sol-gel films. *Talanta* 2009, **80**:511-518.
- 195. Syu MJ, Chiu TC, Lai CY, Chang YS: Amperometric detection of bilirubin from a micro-sensing electrode with a synthetic bilirubin imprinted poly(MAA-co-EGDMA) film. *Biosensors & Bioelectronics* 2006, 22:550-557.
- 196. Lu F, Wang LY, Wu FQ, Ying XF, Gu RA, Yao JL, Zhang HL, Deng ZW: Evaluation of the binding specificity of electrosynthesized poly-orthophenylenediamine molecularly imprinted with metal chelates. *Microchimica Acta* 2011, **174:**47-54.
- 197. Hiratani H, Alvarez-Lorenzo C: Timolol uptake and release by imprinted soft contact lenses made of N,N-diethylacrylamide and methacrylic acid. *Journal of Controlled Release* 2002, 83:223-230.
- 198. Hiratani H, Fujiwara A, Tamiya Y, Mizutani Y, Alvarez-Lorenzo C: Ocular release of timolol from molecularly imprinted soft contact lenses. *Biomaterials* 2005, **26**:1293-1298.
- 199. Hiratani H, Mizutani Y, Alvarez-Lorenzo C: Controlling drug release from imprinted hydrogels by modifying the characteristics of the imprinted cavities. *Macromolecular Bioscience* 2005, **5**:728-733.
- 200. Schmidt RH, Haupt K: Molecularly imprinted polymer films with binding properties enhanced by the reaction-induced phase separation of a sacrificial polymeric porogen. *Chemistry of Materials* 2005, **17**:1007-1016.
- 201. Wang X, Pan JM, Guan W, Dai JD, Zou XH, Yan YS, Li CD, Hu W: Selective Removal of 3-Chlorophenol from Aqueous Solution Using Surface Molecularly Imprinted Microspheres. Journal of Chemical and Engineering Data 2011, 56:2793-2801.
- 202. Pereira L, Rath S: Molecularly imprinted solid-phase extraction for the determination of fenitrothion in tomatoes. *Analytical and Bioanalytical Chemistry* 2009, **393**:1063-1072.
- 203. Weng Z, Muratsugu S, Ishiguro N, Ohkoshi S-i, Tada M: Preparation of surface molecularly imprinted Ru-complex catalysts for asymmetric transfer hydrogenation in water media. *Dalton Transactions* 2010, 40:2338-2347.
- 204. Holland N, Frisby J, Owens E, Hughes H, Duggan P, McLoughlin P: The influence of polymer morphology on the performance of molecularly imprinted polymers. *Polymer* 2010, **51**:1578-1584.
- 205. Zhang Z, Yang X, Zhang H, Zhang M, Luo L, Hu Y, Yao S: Novel molecularly imprinted polymers based on multi-walled carbon nanotubes with binary functional monomer for the solid-phase extraction of erythromycin from chicken muscle. *Journal of Chromatography B* 2011, **879:**1617-1624.
- 206. Sing KSW: **Reporting physisorption data for gas/solid systems with special** reference to the determination of surface area and porosity (**Recommendations 1984**). *Pure and Applied Chemistry* 1985, **57**:603-619.

- 207. Umpleby Ii RJ, Bode M, Shimizu KD: Measurement of the continuous distribution of binding sites in molecularly imprinted polymers. *Analyst* 2000, **125**:1261-1265.
- 208. Ying XG, Cheng GX, Li X: The Imprinting Induce-Fit Model of Specific Rebinding of Macromolecularly Imprinted Polymer Microspheres. Journal of Applied Polymer Science 2011, 122:1847-1856.
- 209. Rampey AM, Umpleby RJ, Rushton GT, Iseman JC, Shah RN, Shimizu KD: Characterization of the Imprint Effect and the Influence of Imprinting Conditions on Affinity, Capacity, and Heterogeneity in Molecularly Imprinted Polymers Using the Freundlich Isotherm-Affinity Distribution Analysis. Analytical Chemistry 2004, 76:1123-1133.
- 210. Rushton GT, Karns CL, Shimizu KD: A critical examination of the use of the Freundlich isotherm in characterizing molecularly imprinted polymers (MIPs). Analytica Chimica Acta 2005, 528:107-113.
- 211. Umpleby RJ, Baxter SC, Bode M, Berch JK, Shah RN, Shimizu KD: Application of the Freundlich adsorption isotherm in the characterization of molecularly imprinted polymers. *Analytica Chimica Acta* 2001, **435**:35-42.
- 212. Umpleby RJ, Baxter SC, Chen Y, Shah RN, Shimizu KD: Characterization of Molecularly Imprinted Polymers with the Langmuir-Freundlich Isotherm. *Analytical Chemistry* 2001, **73**:4584-4591.
- 213. Umpleby RJ, Baxter SC, Rampey AM, Rushton GT, Chen Y, Shimizu KD: Characterization of the heterogeneous binding site affinity distributions in molecularly imprinted polymers. *Journal of Chromatography B* 2004, 804:141-149.
- 214. Sellergren B: Direct drug determination by selective sample enrichment on an imprinted polymer. *Analytical Chemistry* 1994, **66**:1578-1582.
- 215. [http://www.sigmaaldrich.com/analytical-chromatography/samplepreparation/spe/supelmip/html]
- 216. [http://www.polyintell.com/index_uk.htm]
- 217. Zhu T, Li SN, Row KH: Molecularly Imprinted Monolithic Material for the Extraction of Three Organic Acids from Salicornia herbacea L. Journal of Applied Polymer Science 2011, 121:1691-1696.
- 218. Chen CY, Wang CH, Chen AH: Recognition of molecularly imprinted polymers for a quaternary alkaloid of berberine. *Talanta* 2011, 84:1038-1046.
- 219. Zhu T, Yoon C, Row K: Solid-phase Extraction of beta-Sitosterol from Oldenlandia diffusa Using Molecular Imprinting Polymer. *Chinese Journal* of Chemistry 2011, **29:**1246-1250.
- 220. Kecili R, Ozcan AA, Ersoz A, Hur D, Denizli A, Say R: Superparamagnetic nanotraps containing MIP based mimic lipase for biotransformations uses. *Journal of Nanoparticle Research* 2011, **13**:2073-2079.
- 221. Huang X, Yin Y, Liu Y, Bai X, Zhang Z, Xu J, Shen J, Liu J: Incorporation of glutathione peroxidase active site into polymer based on imprinting strategy. *Biosensors and Bioelectronics* 2009, **25**:657-660.
- 222. Alexander C, Davidson L, Hayes W: Imprinted polymers: artificial molecular recognition materials with applications in synthesis and catalysis. *Tetrahedron* 2003, **59**:2025.
- 223. Strikovsky AG, Kasper D, Grun M, Green BS, Hradil J, Wulff G: Catalytic Molecularly Imprinted Polymers Using Conventional Bulk Polymerization or Suspension Polymerization: Selective Hydrolysis of Diphenyl

Carbonate and Diphenyl Carbamate. *Journal of the American Chemical Society* 2000, **122:**6295-6296.

- 224. Carboni D, Flavin K, Servant A, Gouverneur V, Resmini M: The first example of molecularly imprinted nanogels with aldolase type I activity. *Chemistry-A European Journal* 2008, 14:7059-7065.
- 225. Hedin-Dahlstrom J, Rosengren-Holmberg JP, Legrand S, Wikman S, Nicholls IA: A class II aldolase mimic. *Journal Of Organic Chemistry* 2006, **71:**4845-4853.
- 226. Svenson J, Zheng N, Nicholls IA: A Molecularly Imprinted Polymer-Based Synthetic Transaminase. Journal of the American Chemical Society 2004, 126:8554-8560.
- 227. Kirsch N, Hedin-Dahlström J, Henschel H, Whitcombe MJ, Wikman S, Nicholls IA: Molecularly imprinted polymer catalysis of a Diels-Alder reaction. *Journal of Molecular Catalysis B: Enzymatic* 2009, **58:**110.
- 228. Li SJ, Ge Y, Turner APF: A Catalytic and Positively Thermosensitive Molecularly Imprinted Polymer. Advanced Functional Materials 2011, 21:1194-1200.
- 229. Liu JQ, Wulff G: Molecularly imprinted polymers with strong carboxypeptidase A-like activity: Combination of an amidinium function with a zinc-ion binding site in transition-state imprinted cavities. *Angewandte Chemie-International Edition* 2004, **43**:1287-1290.
- 230. Norell MC, Andersson HS, Nicholls IA: **Theophylline molecularly imprinted polymer dissociation kinetics: a novel sustained release drug dosage mechanism.** *Journal Of Molecular Recognition* 1998, **11**:98-102.
- 231. Sreenivasan K: On the application of molecularly imprinted poly(HEMA) as a template responsive release system. *Journal of Applied Polymer Science* 1999, **71:**1819-1821.
- 232. Bayer CL, Peppas NA: Advances in recognitive, conductive and responsive delivery systems. *Journal Of Controlled Release* 2008, **132**:216-221.
- 233. Sellergren B, Allender CJ: Molecularly imprinted polymers: A bridge to advanced drug delivery. Advanced Drug Delivery Reviews 2005, 57:1733.
- 234. Wang C, Javadi A, Ghaffari M, Gong S: A pH-sensitive molecularly imprinted nanospheres/hydrogel composite as a coating for implantable biosensors. *Biomaterials* 2010, **31**:4944-4951.
- 235. Byrne ME, Salian V: Molecular imprinting within hydrogels II: Progress and analysis of the field. International Journal Of Pharmaceutics 2008, 364:188-212.
- 236. Hiratani H, Alvarez-Lorenzo C: The nature of backbone monomers determines the performance of imprinted soft contact lenses as timolol drug delivery systems. *Biomaterials* 2004, 25:1105-1113.
- 237. Alvarez-Lorenzo C, Yanez F, Barreiro-Iglesias R, Concheiro A: Imprinted soft contact lenses as norfloxacin delivery systems. *Journal of Controlled Release* 2006, 113:236-244.
- 238. Venkatesh S, Sizemore SP, Byrne ME: Biomimetic hydrogels for enhanced loading and extended release of ocular therapeutics. *Biomaterials* 2007, 28:717-724.
- 239. Noss KR, Vaughan AD, Byrne ME: Tailored binding and transport parameters of molecularly imprinted films via macromolecular structure: The rational design of recognitive polymers. Journal Of Applied Polymer Science 2008, 107:3435-3441.

- 240. Ali M, Horikawa S, Venkatesh S, Saha J, Hong JW, Byrne ME: Zero-order therapeutic release from imprinted hydrogel contact lenses within in vitro physiological ocular tear flow. *Journal Of Controlled Release* 2007, **124**:154-162.
- 241. Ali M, Byrne ME: Controlled Release of High Molecular Weight Hyaluronic Acid from Molecularly Imprinted Hydrogel Contact Lenses. *Pharmaceutical Research* 2009, **26**:714-726.
- 242. Vaughan AD, Zhang JB, Byrne ME: Enhancing Therapeutic Loading and Delaying Transport via Molecular Imprinting and Living/Controlled Polymerization. *Aiche Journal* 2010, **56:**268-279.
- 243. Oh JK, Drumright R, Siegwart DJ, Matyjaszewski K: The development of microgels/nanogels for drug delivery applications. *Progress In Polymer Science* 2008, 33:448-477.
- 244. Kompella UB, Bandi N, Ayalasomayajula SP: **Poly(lactic acid) nanoparticles for sustained release of budesonide.** *Drug Delivery Technology* 2001, **1**:29-35.
- 245. Robinson DN, Peppas NA: Preparation and Characterization of pH-Responsive Poly(methacrylic acid-g-ethylene glycol) Nanospheres. *Macromolecules* 2002, **35**:3668-3674.
- 246. Cunliffe D, Kirby A, Alexander C: Molecularly imprinted drug delivery systems. Advanced Drug Delivery Reviews 2005, 57:1836.
- 247. Javanbakht M, Mohammadi S, Esfandyari-Manesh M, Abdouss M: Molecularly Imprinted Polymer Microspheres with Nanopore Cavities Prepared by Precipitation Polymerization as New Carriers for the Sustained Release of Dipyridamole. *Journal of Applied Polymer Science* 2011, **119**:1586-1593.

Chapter 2

Towards a budesonide specific molecularly imprinted polymer

2. Towards a budesonide specific molecularly imprinted polymer

2.1 Introduction

2.1.1 Steroids

Steroids are an important class of biologically active compounds with a wide reaching range of biological functions [1, 2]. These compounds are based on a tetracyclic hydrocarbon structure with variation in the degree of saturation and substitution of the rings. The most widely recognised member of this class of compounds is cholesterol, a steroid which is metabolised to bile acids in the liver and functions as a precursor for the synthesis of steroid hormones. An example of the former is cholic acid, which is necessary for digestion to proceed in the intestine where this compound forms micelles with water-insoluble fats and fatty acids [1].

Steroid hormones are further divided based on their function [1]. Subdivisions include progestins such as progesterone, which are synthesised in the ovarian cells, androgens such as testosterone, important for the development of male characteristics and estrogens such as estradiol, which is responsible for the growth of female characteristics. Another such subdivision is the glucocorticoids, for example hydrocortisone, a steroid present in abundance during situations of stress and inflammation and which also has a suppressant effect on the immune system. Due to their obvious importance in biological functions, steroids are often an integral part of pharmacological formulations [2]. Steroids are used in the treatment of a diverse range of conditions, for example, asthma [3], Crohn's disease [4] and age-related macular degeneration [5]. While steroids have many beneficial properties, they can also be used as illegal growth promoters, as in the case of anabolic steroids [6] and estrogens can act as environmental endocrine disruptors, which are environmental pollutants that can have adverse affects on the health of humans and other organisms [7, 8].

2.1.2 Steroids in molecular imprinting

In molecular imprinting strategies the bulk of the polymer is typically hydrophobic, with specific binding sites created using functional monomers capable of interaction, such as hydrogen bonding, with the template. Similarly, in biological molecules, such as the DB3 monoclonal antibody, the vast majority of the hydrophobic steroid structure is incorporated into hydrophobic pockets, where a small number of hydrogen bonding contacts effect further selectivity and interaction [1, 9]. The similarity of these situations and the dual positive and negative roles of steroids in areas such as medicine and environmental pollution, as well as an urgent need for the development of new analytical strategies for steroidal components of pharmaceutical formulations [10] serve to demonstrate why steroids are a suitable target for research into the preparation of biomimetic polymers within the MIP community.

Early imprinting work involving development of a polymer capable of the selective catalytic reduction of steroidal 3- and 17-ketones was carried out by Bystrom *et al.* [11]. The chemical structures of one of the template-monomers used in the covalent imprinting procedure and the corresponding target analyte, androstane-3,17-dione, are given in Figure 2.1.



Figure 2.1: I Androstane-3,17-dione, the steroid targeted for selective reduction using an imprinted polymer prepared with II, a polymerisable template-monomer [11].

In this study polymerisable template monomers, such as **II** in Figure 2.1, were used to create imprinted polymers that allowed for the selective reduction of either the 3- or the 17-ketone positions on selected substrates. This was exemplified by the use of a steroid with diketone functionality, Figure 2.1 **I**, where it was possible to completely direct reduction to the 17-position using an imprinted polymer. Conversely, when in solution without the polymer, 99% of the product was reduced at the 3-position.

Semi-covalent methods were later employed for cholesterol by Whitcombe *et al.* [12] (discussed in detail in Section 1.3.3.1) and the first steroid to be imprinted by non-covalent means was hydrocortisone in the work of Ramström *et al.* [13] (discussed in detail in Section 1.3.2). Since this early work, a number of steroids have been studied and molecular imprinting technologies have been investigated for a number of purposes. Figure 2.2 presents a number of the steroids that have been investigated to date while examples of the steroid imprinted polymers that have been prepared are presented in Table 2.1.



Figure 2.2: Structures of some commonly imprinted steroids.

Steroid	Template	Monomers		Polymerisation	References
	monomer	Functional	Crosslinking	strategy	
	interaction	monomer	monomer		
Cholesterol	Non-covalent	MAA	TRIM	Precipitation	[14]
	Non-covalent	MAA	EGDMA	Seeded suspension	[15]
	Semi-covalent	Synthesised	Toluene 2,4-	Bulk monolith	[16]
		from chitin	diisocyanate		
	Non-covalent	MAA	EDGMA/PEGDMA	Hydrogel film	[17]
β-estradiol	Non-covalent	TFMAA	TRIM	Precipitation	[7]
	Non-covalent	MAA	DVB/TRIM	Precipitation	[18]
	Covalent	Synthesised	DVB/EDGMA	Precipitation	[19]
	Non-covalent	MAA	TRIM	Precipitation	[20]
	Non-covalent	MAA	DVB/EGDMA	Precipitation	[21]
Testosterone	Non-covalent	MAA	EGDMA	Bulk monolith	[22]
	Non-covalent	MAA	EGDMA	Bulk monolith	[23]
	Non-covalent	MAA	EGDMA	Bulk monolith	[24]
Hydrocortisone	Non-covalent	MAA	EGDMA	Bulk monolith	[13], [25]
	Non-covalent	HEMA	EGDMA	Bulk monolith	[26]
	Non-covalent	HEMA	EGDMA	Bulk monolith	[27]

Table 2.1: Examples of some steroid selective molecularly imprinted polymers that have been reported in the literature.

Figure 2.2 and Table 2.1 serve to demonstrate that a number of steroids have been used as templates for the creation of MIPs prepared in a variety of formats. Much of the molecular imprinting of steroids has been carried out with a view to the chromatographic separation of steroids [25, 28] where the absolute binding capacity for the imprinted polymers is not of any particular relevance and as such, is often not calculated. Such applications aside, the absolute binding capacity is still rarely quoted and in instances where it is, the values are predominantly low, for example 2.9 mg/g for hydrocortisone [27] and 5 mg/g for testosterone [22]. In the majority of cases, relative values, i.e. values for the MIP versus the NIP, rather than absolute values are quoted. In these cases the conditions used for rebinding experiments may be a clue to the binding capacity. This is exemplified by the work of Ye et al. [20] and Wei et al. [21] where 30 mg of polymer was added to 1 mL of solvent containing 417 fmol of template. These relatively low capacities suggest that the creation of steroid selective binding sites within polymers is no trivial matter. The relevance of the binding capacity would be dependant on the end application, for example, MIPs for trace environmental analysis or sensing applications would not require a high capacity, however, long-term controlled release may require higher polymer loading capacities.

2.1.3 Strategies for the synthesis of steroid imprinted polymers

As demonstrated in Table 2.1 a number of different polymer formats have been used for the synthesis of steroid specific MIPs and while the majority of examples presented in this table are comprised of bulk monolith or precipitation, other formats have also been employed. Cholesterol has been imprinted by Lee *et al.* using polystyrene microbeads for seeded suspension polymerisation [15]. This template has also been imprinted by Spizzirri and Peppas [17] using hydrogels prepared from MAA as the functional monomer and either EGDMA or PEGDMA as crosslinking agents. These examples notwithstanding, the majority of steroid imprinting has been carried out using either bulk monolith or precipitation methods [7, 8, 20, 21, 29-31].

A number of studies have compared the performance of steroid imprinted MIPs that have been prepared using different strategies. One such study was that carried out by Ye *et al.* where imprinted polymers were prepared for 17β -estradiol by bulk monolithic and precipitated microspheres [32]. MAA was used as the functional monomer in all cases while EGDMA was used as the crosslinking monomer in both methods. Additional microspheres were prepared using TRIM as the crosslinking monomer. Acetonitrile was the solvent used for all polymer synthesis. In the case of the monolithic polymer the MIP bound 30% of the added template while the NIP bound <13%, i.e. the MIP bound more than twice as much as the NIP. In the case of the microspheres using EGDMA as the crosslinking monomer, the MIP bound only 14% of the added template but the NIP in this case bound almost five times less template, at 3%.

While the bulk monolithic polymers bound more template than the microspheres, the binding was far more selective in the case of the microspheres. In an effort to improve the binding capacity of the microspheres, TRIM was used as crosslinking monomer to reduce the content of non-binding monomer in the composition of the polymer. The use of the tri-functional monomer TRIM allowed for a reduction in the percentage of crosslinker used, thereby increasing the percentage of functional monomer in the polymer composition. In this case, the polymers bound approximately 50% of the added template while the control microspheres bound three times less than this. Thus, with careful tailoring of the properties of the microspheres it was possible to achieve both a higher selectivity and a higher binding capacity than with the polymers prepared as bulk monoliths.

In a similar study, Wei and Mizaikoff compared the performance of 17β -estradiol imprinted polymers that were prepared in three formats: bulk monolith, microspheres and sub-microspheres [33]. They investigated the performance of the polymers in three areas: release studies, equilibrium binding and non-equilibrium binding. During equilibrium binding studies it was observed that in all cases the control NIPs bound less of the template than the MIPs but the control microspherical polymers rebound less template than the other two control polymers. This corroborates the findings of Ye *et al.* in that less non-specific binding occurred in the case of the microspheres (*vide supra*). Using binding isotherms based on the Freundlich model the microspheres showed a higher median binding affinity constant $(1.3 \times 10^{-2} \pm 4.8 \times 10^{-4} \text{ mM}^{-1})$ than the values for the bulk particulate $(9.0 \times 10^{-3} \pm 5.3 \times 10^{-4} \text{ mM}^{-1})$. During release studies the polymer (20 mg) was placed in 1 mL of acetonitrile and the amount of template released was determined by HPLC. The kinetics of template release was deemed to be comparable for all three formats but it was observed that overall, more template was

released from the microspheres. A possible reason for this was the increased mobility of the microspheres, thereby improving extraction conditions. Given the suggested superior performance of microspheres prepared by precipitation methods, and its relevance to this study, it is considered worthwhile to expand on the introduction to this method presented in Section 1.5.2.

2.1.4 Precipitation polymerisation

Early work on the preparation of polymeric microspheres without the use of stabilisers and surfactants was carried out by Naka *et al.* [34, 35]. This work involved the use of diethylene glycol dimethacrylate as the monomer and initiation was carried out by exposure of the monomers to ionising radiation. This yielded microspherical particles in good yields with a high degree of monodispersity. Findings in this early work pointed to the importance of the total monomer concentration with respect to solvent and above a certain 'critical' concentration, 'gelation' occurred and a polymeric gel was formed rather than microspherical particles. This was attributed to the ability of the polymer network to grow at a rapid rate under the conditions used. The evidence also suggested that 'good' solvents for the monomer, i.e. solvents with solubility parameters close to that of the monomer, were good solvents for the preparation of microspheres. The size of the microspheres ranged between 0.4 and 4.8 μ m, depending on the solvents used.

A series of studies carried out by Stover *et al.* investigated the precipitation polymerisation of DVB to form microspheres between 2 and 7 μ m [36-38]. In contrast to the work of Naka *et al.*, an initiator was used to carry out the polymerisation. In the first of these studies [36] Li and Stover stated that the precipitation of the microspherical particles was due to the decreased level of solubility of the growing polymer because of the rising degree of crosslinking. This was later termed 'entropic' precipitation [38].

In the initial study, the effect of monomer concentration, solvent type, initiator type and concentration were all investigated. It was found that here, as in the work of Naka *et al.*, that above a certain monomer concentration (in this case 5 vol%) coagulation of the particles occurred, to form either space-filling gels or coagulum and it was again

suggested that this was due to increased solvating power of the continuous phase leading to increased likelihood of coagulation.

Increasing the concentration of initiator increased the conversion of monomer but was incomplete even at high initiator concentrations while the type of initiator also had a profound effect on the conversion and size distribution of the particles. Later work [37] showed that the use of toluene as a co-solvent produced porous microspheres with surface areas of 800 m²/g, in sharp contrast to values of approximately 9 m²/g for polymers prepared in neat acetonitrile. This was attributed to the fact that in a poor solvent for the growing polymer, such as acetonitrile, phase separation occurs early and leads to non-porous polymeric microspheres whereas the addition of toluene delays the onset of phase separation causing the entrapment of some solvent in the growing polymer ultimately leading to pore formation.

Ye *et al.* [20] first introduced the use of precipitation in the formation of molecularly imprinted polymers. Both theophylline and 17- β -estradiol were imprinted in polymers prepared using MAA and TRIM as the functional and crosslinking monomers respectively. Microspheres in the range of 0.2-0.3 µm were prepared in this fashion. Since this initial imprinting work with precipitated microspheres, a wide range of templates have been imprinted using this polymer format including the steroid dexamethasone [31], phenols [39], the alkaloid cinchonidine [40] and the antineoplastic agent, fluorouracil [41].

In addition to the work outlined above, comparing MIPs prepared by precipitation methods to those prepared by bulk monolith methods, Mayes and Pérez-Moral carried out a comprehensive investigation of the various polymeric formats used in molecular imprinting [42]. In this study five polymeric formats were investigated: bulk; suspension in fluorocarbon solvent; two-step swelling, emulsion core-shell and precipitation. MAA was the functional monomer used while EGDMA was used as the crosslinker and all of the polymers were imprinted with the β -blocker, propranolol.

In the radioligand binding assays that were used, the imprinted precipitation polymer rebound 53% of the template in the toluene solution in comparison to the control polymer which bound only 3%. This suggests that the binding in the imprinted polymer was almost 100% specific. The bulk imprinted polymer bound only 39% of the template in toluene. In aqueous solution, the imprinted precipitation polymer bound 43% of the template in aqueous buffer (the second highest percentage rebound) but none of the binding was specific, i.e. the same amount of template was bound to both the MIP and the NIP.

In 2011 the work of Mohajeri *et al.*, performing a comparison of bulk monolith particles and precipitated beads for clozapine, showed improved specificity for the precipitated polymers with an imprinting factor of 3.26 compared to that of 1.95 for the bulk particles [43]. Scatchard analysis showed that the binding sites of the precipitated polymers were more homogenous than those of the bulk particles, with only one dissociation constant observed for the beads while two were observed for the bulk particles. The binding affinity was also stronger towards the beads, with the dissociation constant of only 0.45 μ M for the beads but a value 32 times higher at 14.5 μ M for the bulk particles.

2.1.4.1 Advantages and disadvantages of precipitation polymers

There are a number of advantages and disadvantages associated with the use of precipitation strategies and these are summarised in Table 2.2.

Advantages	Disadvantages
Controllable monodisperse particles in	Possible pressure build up on
the micron and sub-micron range are	chromatographic columns due to small
easily synthesised.	particle size.
No grinding or sieving necessary	Optimisation necessary for each polymer
	composition.
No additional surfactant needed.	
Rapid synthesis allowing for readily	
optimised conditions.	
High yield of usable polymer.	

Table 2.2: Advantages and disadvantages of the use of precipitation polymerisation

It is also worth noting that spherical particles are well suited for use in polymer based drug delivery systems [44]. Typically having a large surface area to volume ratio and an interior network capable of incorporating drugs and biomolecules [45], micro- and nanospheres can have a high degree of uptake into cells as well as causing minimal irritation if administered subcutaneously [4, 46].

2.2 Research aims and objectives

The aims and objectives of this chapter were as follows:

- To prepare molecularly imprinted polymers for budesonide, a corticosteroid used for respiratory, bowel and eye conditions and which has been studied for lung cancer treatment.
- To determine the most suitable monomer composition based on previously reported MIPs for structural analogues of budesonide.
- To adapt the selected monomer composition for the preparation of polymers using precipitation methodologies outlined in the literature.
- To determine the factors affecting the polymerisation and thus optimise the conditions for the preparation of microspherical polymers.
- To develop a microspherical molecularly imprinted polymer capable of the selective uptake and release of budesonide.

2.3 Experimental

2.3.1 Materials

Reagent	Assay	Supplier
2-Hydroxyethyl methacrylate	98%	Sigma-Aldrich
Ethylene glycol dimethacrylate	98%	Sigma-Aldrich
Methacrylic acid	99%	Sigma-Aldrich
Divinyl benzene	80%	Sigma-Aldrich
2,2'-Azobis(2-	98%	Acros
methylpropionitrile)		
1,1-Azo bis	98%	Sigma
(cyclohexanecarbonitrile)		
4',4-azobis(4-cyanovaleric acid)	98%	Sigma
Acetonitrile	>99.9%	Romil
Acetonitrile- <i>d</i> ₃	99.8 atom% D	Sigma-Aldrich
Toluene	>99.9%	Romil
Dichloromethane	>99.9%	Romil
Chloroform	>99.8%	Sigma-Aldrich
Chloroform- <i>d</i> ₃	99.8 atom% D	
Methanol	>99.9%	Sigma-Aldrich
Acetic acid	>99.8%	Fluka
Acetic acid- d_4	99.5 atom% D	Sigma-Aldrich
Tetrahydrofuran	>99.8%	Romil
Dimethyl sulfoxide	≥99.5%	Riedel-de-Hae
Budesonide	98%	Eirgen Pharma
Prednisolone	99%	Sigma

Table 2.3: List of materials employed during the course of this study.

All reagents from Sigma-Aldrich were obtained from Sigma-Aldrich (Dublin, Ireland) while all of the solvents from Romil were obtained from Lennox Laboratory Supplies (Dublin, Ireland). All solvents used were of HPLC grade. AIBN was obtained from Acros Chimica (Geel, Belgium) and was recrystallised from acetone prior to use.

Inhibitor removal was carried out for MAA by vacuum distillation at 80 °C; EGDMA was purified by sequentially washing with 10% NaOH, saturated NaCl, dried over magnesium sulphate and vacuum distilled at 120 °C; DVB was purified using a basic alumina column. HEMA was used as received and all monomers were stored at -20 °C prior to use.

2.3.2 Instrumentation

Instrument	Model	Supplier
Photochemical	RMR-600	Branford, CT, USA
mini-reactor (8 W		
lamps)		
UV-Vis	UV-2401 PC	Shimadzu, Japan
spectrophotometer		
Optical microscope	Olympus CH20	Optical Co. Ltd., Japan
Microscope camera	MiniVid	LW Scientific, USA
Scanning electron	Hitachi S-2460N	Oxford instruments,
microscope		England
Shaking incubator	C24	New Brunswick
		Scientific, USA
HPLC	1200	Agilent Technologies,
		Ireland
Particle size	Mastersizer 2000S	Malvern, Ireland
analyser		
NMR spectrometer	EX 400	Jeol, England

Table 2.4: Instrumentation used for this study

2.3.3 Preparation of polymers

The polymers were prepared in accordance with Table 2.5.

Polymer	Template	Functional	Crosslinking	Molar ratio ^a	Solvent	Polymerisation
		monomer	monomer			conditions
M1	BDN	MAA	EGDMA	1:10:50	MeCN	UV; 350 nm 4 °C
						Thermal 60/80 °C
M2	BDN	MAA	EGDMA	1:10:30	MeCN	Thermal 60/80 °C
M3	BDN	HEMA	EGDMA	1:10:50	MeCN	Thermal 60 °C
M4	BDN	MAA	EGDMA	S ^b :10:50	MeCN	Thermal 60 °C
M5	PRD	MAA	EGDMA	1:10:50	MeCN	Thermal 60 °C
M6	BDN	MAA	EGDMA	1:10:50	MeCN/TOL	Thermal 60 °C
					3:1 and 4:1 (v/v)	
M7	BDN	MAA	DVB	S ^b :10:50	MeCN/TOL	Thermal 60 °C
					3:1 and 4:1 (v/v)	
M8			DVB	N/A	MeCN/TOL	Thermal 60 °C
					3:1 (v/v)	
M9	PRD	MAA	DVB	1:10:30	MeCN/TOL	Thermal 60 °C
					3:1 (v/v)	
M10	PRD	MAA	DVB	1:10:50	MeCN/TOL	Thermal 60 °C
					3:1 (v/v)	

 Table 2.5: Polymer compositions and polymerisation conditions used for this study.

^a molar ratio of template:functional monomer:crosslinking monomer; ^b saturated solution of template in porogen used to drive complex formation

Polymers were prepared in accordance with Table 2.5. For ease of observation, the nonimprinted polymers are not included in the table; in all cases they were prepared in the same manner but without the presence of the template. Typically, the template and functional monomer were added to the solvent, sonicated then placed in ice and allowed to equilibrate for 20 minutes. The crosslinking monomer and initiator were then added. Further sonication was carried out to ensure a homogenous solution and the prepolymerisation mixture was purged with N₂ gas for 5 minutes while placed in an ice bath. Total monomer concentration was adjusted from 0.5-6% (w/v) by varying the volume of the solvent or the total mass of the monomers.

Polymerisation was carried out either by thermal initiation or UV initiation. Thermal initiation was carried out in a shaking water bath with gentle agitation (20 RPM) where the temperature was raised from room, to the required temperature (Table 2.5) over a 2 hour period. UV initiation was carried out using Rayonet photochemical mini-reactor at 4 °C and the polymerisation vial was rotated about its long axis at 45 RPM.

Upon completion of polymerisation the polymers were filtered using 0.45 µm PTFE Cronus membrane filters from Antech (Waterford, Ireland). They were then rinsed with 50 mL of each porogen used. The polymers were then suspended in 4:1 MeOH:AcOH at a concentration of 1 g of polymer/50 mL of solvent. They were stirred with gentle heating for 2 hours, filtered and rinsed with three 30 mL aliquots of hot MeOH. This was repeated twice and the polymers were then suspended in hot MeOH at the same concentration and stirred for 2 hours, filtered and rinsed and rinsed with 100 mL of hot MeOH. The removal of template and polymeric bleed was monitored by UV-Vis spectroscopy using the UV-2401 PC UV-Vis spectrophotometer from Shimadzu (Japan). The polymer particles were finally rinsed with HPLC grade acetone and allowed to dry for 48 hours at 35 °C.

(Note on polymer notation: Throughout the text, M denotes precipitation polymer, MB bulk polymer, while N denotes precipitated non-imprinted polymer and NB, non-imprinted bulk polymer. The number, e.g. 1, 3, etc. serves to denote the formulation used. For example, N3 would be the non-imprinted polymer prepared using BDN, MAA and HEMA in a 1:10:50 ratio. Where only the formulation is discussed and not a comparison of MIP vs. NIP, M or MB will be used).

2.3.4 Scanning electron and optical microscopy

Optical microscopy was carried by depositing the beads on a microscopic slide with gridlines 50 μ m in length. Optical micrographs were obtained using the MiniVid microscope camera and images recorded using the ScopePhoto software program. These were used for preliminary basic particle size measurements. The size of the particles was determined as follows: using an optical micrograph obtained as described, the image of the 50 μ m line on the slide was physically measured and the size of polymeric particles were calculated based on this physical measurement. Where a single value is quoted values were calculated by getting an average value for ten polymeric particles per image and an average of three images per polymer formulation

Prior to scanning electron microscopy (SEM) the polymers were sputter-coated with gold.

2.3.5 Equiibrium binding studies

For polymers M1-M7, dry polymer (50 mg) was placed in 1, 5 or 10 mL of a solution of the template used in polymer synthesis, ranging in template concentration from 0.0175 mM to 1 mM. The suspended polymer solution was then placed in a C24 incubator shaker from New Brunswick Scientific (Edison, NJ, USA) at 20 °C at 300 RPM. After 24 hours the suspended polymer solution was centrifuged at 45000 RPM for 1 hour and the supernatant was filtered through 0.45 and 0.2 µm syringe filters. The rebinding analyses were carried out by measuring the absorbance of the rebinding solution at the λ_{max} of the template and comparing it to the original template solution to determine the percentage of template bound. The supernatant was also compared to a solution of the polymer and the rebinding solvent only, in order to determine the level of polymeric bleed that was present. Rebinding analyses were carried out using UV-Vis spectroscopy.

Polymers M8-M10 were tested using 30 mg dry polymer in a 0.9 mL solution of the target analyte in 1.5 mL HPLC vials, in concentrations ranging from 2-140 μ g/mL. Equilibrium binding conditions were as above but the supernatant was filtered into fresh HPLC vials and the concentration of unbound steroid was quantified using reverse phase chromatography.

HPLC was carried out using a method previously reported for budesonide analysis [47]. Here, an Agilent 1200 series system from Agilent Technologies was employed, using an Agilent 150 × 4.6 mm 5 μ m Eclipse XDB-C₁₈ column and a mobile phase composed of 69:31 methanol-aqueous buffer (v/v). The aqueous buffer was prepared using 0.1% acetic acid at pH 3 with a flow rate of 1 mL.min⁻¹. The run time was 8 minutes. A standard curve was constructed for the steroids analysed by plotting the peak areas against the concentration. The method was linear from 0.25-20 µg.mL⁻¹. The amount of steroid bound to the polymer, was again calculated by subtracting the concentration of the free steroid from the initial known concentration. All injections were carried out in triplicate. Imprinting factors, IF, in all cases were calculated using the following formula:

$$IF = \frac{\%Bound\,MIP}{\%Bound\,NIP}$$
 Equation 2.1

Where, *IF* is the imprinting factor and *%Bound MIP* and *%Bound NIP* are the percentages of the steroid bound to the imprinted and non-imprinted polymers, respectively.

2.3.6 Particle size analysis

Particle size analysis was carried out using a Malvern Mastersizer 2000S laser diffraction liquid particle size analyser. MeOH was used as the dispersant and a stirring speed of 1750 RPM was applied. 20 mg of dry polymer was suspended in 1 mL of MeOH and sonicated for 5 minutes. The polymer solution was then added to the dispersant until the laser obscuration was between 7 and 9%. Where particle aggregation was observed, sonication was applied until no evidence of aggregation was present. The optical properties were empirically determined post-analysis to obtain the lowest residual value, giving the best fit between the predicted and observed values. The d(0.1), d(0.5), d(0.8) and d(0.9) values were determined and particle size values are given as a range between d(0.1) and d(0.9), thus giving a size range that contains 80% of the particles analysed. Where aggregation persisted the size range given was determined based on the values between the d(0.1) and d(0.8) values, which contained 70% of the sample analysed. In a small number of cases, where particle aggregation

could not be prevented the size range quoted is between d(0.1) and d(0.5). All values given are an average of three analyses.

2.3.7 NMR analysis

Preliminary ¹H NMR spectra were recorded on a Bruker spectrometer operating at 500 MHz and a temperature of 25 °C, while a Jeol EX 400 spectrometer operating at 400 MHz at 18.5-18.9 °C was used for the bulk of the work. Chemical shifts were referenced to the TMS peak. Where relevant deuterated acetic acid, AcOD, was used as a functional analogue of MAA.

2.3.7.1 Job plot analysis

Equimolar solutions of the template and the functional monomer (or functional analogue) were prepared at the following concentrations: AcOD and BDN in CDCl₃: 0.04 M; AcOD and BDN in CD₃CN: 0.02 M; HEMA and BDN in CDCl₃: 0.03 M; HEMA and BDN in CD₃CN: 0.02 M. The ratios of the template:monomer were systematically varied between 1:9 and 10:0. The total concentration, of the monomers and template, was kept constant while the volume was held at 0.75 mL. The shim map was determined by carrying out a 'manual gradient shim' on the 8:2 template:monomer solution followed by fine tuning in manual mode. This was carried out for each separate Job plot analysis, which utilised 64 scans and 32768 data points per spectrum. The spectra were analysed and the data used to construct a plot of: $\Delta\delta \times \chi_{BDN}$ vs. χ_{BDN} . Where $\Delta\delta = \delta_{\text{free template}} - \delta_{\text{obs}}$ and χ_{BDN} is the mole fraction of budesonide in the solution.

2.3.7.2 NMR titration

To a 500 μ L sample of BDN (0.03 M) in CDCl₃, a mixture of BDN (0.03 M) and either AcOD (8.73 M) or HEMA (4.12 M) was added incrementally until there were 110 equivalents of AcOD or 90 equivalents of HEMA present. The shim map was determined by carrying out a manual gradient shim on the free template solution and tuned manually when necessary thereafter. 16 scans and 16384 data points were utilised to obtain the NMR spectra. The changes of the chemical shift values of the BDN protons were monitored and association constants were obtained using Prism, GraphPad software 4.0.

2.4 Results and discussion

2.4.1 Initial polymer formulation

Given the advantages of the use of polymeric beads prepared by precipitation polymerisation outlined in Section 2.1.4 and the successful use of this method for imprinting steroids as outlined in Section 2.1.3, it was decided to use this polymer format for the synthesis of budesonide imprinted polymers. Budesonide was chosen as a target due to its use in asthma [3], Crohn's disease [4] and age-related macular degeneration [5]. This potent corticosteroid had, hitherto, not been used in molecular imprinting protocols, thus it was necessary to find a close structural analogue that had been successfully imprinted and adapt the methodology. This is often used as an initial guideline for the development of imprinting protocols [21]. The closest structural analogue observed in the literature was hydrocortisone which, as outlined in Section 2.1.2, has been successfully imprinted by the groups of Ramstrom *et al.*, Baggiani *et al.* and Sreenivasan [13, 25, 26]. The structures of hydrocortisone and the two steroids used in this study, budesonide and prednisolone, are given in Figure 2.3.



Figure 2.3: Structures of hydrocortisone, budesonide and prednisolone, demonstrating the structural similarity of the three compounds. C* is the epimeric carbon in budesonide.

The structures of these steroids are quite similar, with OH groups on the C11 and C21 positions in all molecules as well as carbonyl groups at C3 and C20. Further, all three molecules contain conjugation in the A-ring but while the A-ring of hydrocortisone is a cyclohexenone, having only one internal double bond, both budesonide and prednisolone have two internal double bonds and as such the A-ring is a cyclohexadienone. Hydrocortisone and prednisolone both possess a further OH at the C17 position while budesonide has an acetal grouping at the C16 and C17 positions. It

must also be noted that budesonide is prepared and used as a 1:1 mixture of epimers with the epimeric carbon marked '*' in Figure 2.3 above.

The monomers chosen for initial investigation were MAA and EGDMA, as these monomers are commonly used in imprinting and had been successfully employed by Ramstrom *et al.* and Baggiani *et al.* [13, 25] at template:functional monomer:crosslinking monomer ratios of 1:10:50 and 1:10:90, respectively. A similarly high ratio of functional monomer to template has also been utilised for the imprinting of estradiol [7], cholesterol [48] and testosterone [22]. This is necessary to drive the equilibrium towards interaction, in particular when there can be a relatively low degree of functionality in comparison to the size and rigidity of the molecule.

As discussed in Section 2.1.4, it is the crosslinking process that induces entropic precipitation of the polymeric beads [38]. To exploit this characteristic Cormack *et al.* have used a ratio of 1:5 for functional to crosslinking monomer for precipitation polymerisations using MAA and EGDMA as the monomers for 17β -estradiol [20]. With this in mind, this ratio of crosslinking to functional monomer was chosen as a starting point for this study.

As well as the importance of the total monomer concentration relative to the reaction solvent, typically below 5% for imprinted polymers [32, 37], the concentration of the initiator used also plays a vital role and is typically used at concentrations of 1-3% of the total number of moles of polymerisable double bonds [36, 49]. Given these parameters, a total monomer concentration of 4% (w/v) was initially used, as this allowed for small scale syntheses using 30 mL sample vials for polymerisation reactions having theoretical yields of approximately 1 g of polymer. An initiator concentration of 3 mol% was decided upon in an effort to promote the growth of microspheres and ACCN was used as the initiator during initial trials, as this had previously been used to carry out successful polymerisation reactions within the research group.

The solvents that were initially tested were selected due to their use in the imprinting of steroids [50] and hydrocortisone in particular [13]; their use in precipitation polymerisation [49] and their polarities. These solvents, in order of increasing polarity,
were: toluene, dichloromethane, THF, chloroform, acetonitrile and DMSO. Thermal initiation was carried out as described in Section 2.2.2 and of the seven solvents tested with this monomeric concentration, beads were formed in acetonitrile only.

In all other solvents the polymers formed had the appearance of bulk monolith polymers but appeared to be less densely crosslinked as they did not form rigid materials, being readily broken apart with a spatula. The observed results appeared to be analogous to what was described as 'gelation' by Naka *et al.* [34] where the cause was ascribed to a high percentage of monomers. Sherrington refers to this as macrogelation where the "mass of growing polymer molecules dissolved in solution becomes crosslinked into one infinite network", the result of which is a swollen monolithic soft gel filling the containing vessel [51].

In the present study, this will be termed a space-filling gel. The fact that space-filling gels were formed in the solvents other than acetonitrile, can be attributed to the fact that the monomer loading may have been too high or the growing polymer was too soluble in the solvent system used. This may have led to rapid formation of the growing polymer network, thereby allowing the polymer chain to grow rapidly in a short space of time [35, 36]. The swelling effect of chlorinated solvents on polymers and the resultant difficulties with preparing microspherical beads has also been documented [52]. Some preliminary investigations were carried out on the original space-filling gels formed in these solvents. In these investigations it was observed that upon drying, the polymers (contained in a 30 mL vessel) lost up to 18 g of solvent. Upon addition of 30 mL of solvent that had been lost on drying. This suggested that the polymers were indeed loosely crosslinked.

To investigate whether these space-filling gels were caused by the solubility of the growing polymer or the monomer concentration, the total monomer concentration in relation to the solvent was varied from 0.5-4% for all of the solvents. As the concentration was reduced the polymer morphology changed from an opaque space filling gel to a translucent gel-like polymer that was suspended in the solvent and could only be investigated upon evaporation of the solvent. This suggested that the growing polymer was too soluble in the solvent for the preparation of polymeric beads. For this

reason, solvents other than acetonitrile were no longer investigated for the M1 composition.

Acetonitrile was considered to be an acceptable solvent for use in further investigations because of its initial success in bead formation and due to its previous use for imprinting steroids in general [53] and steroids in precipitation polymerisation [7, 20, 33]. Seminal work on the effect of the porogen on the morphology of polymers used for molecular imprinting by Sellergren and Shea quote MeCN, while being a polar solvent, as having poor hydrogen bonding capacity [54]. This would suggest that the interference with hydrogen bonding between the template and the functional monomer could be minimised. The M1 formulation was synthesised as outlined in Table 2.5, using thermal and photochemical initiation at 60 and 4 °C, respectively. SEM images of the imprinted and non-imprinted beads obtained from thermally and photochemically initiated polymerisations are shown in Figure 2.4.



Figure 2.4: Microspherical and sub-microspherical beads prepared with the M1 formulation at 4% (w/v) monomer concentration and 3 mol% ACCN as initiator. (a) M1 by thermal polymerisation; (b) N1 by thermal polymerisation; (c) M1 by UV polymerisation; (d) N1 by UV polymerisation.

Figure 2.4 shows that for imprinted and non-imprinted polymers, spherical beads in the micron to sub-micron size range were formed in acetonitrile using thermal and photochemical polymerisation. Figure 2.4 also demonstrates that in both cases the imprinted polymer differs in size to the non-imprinted polymer. Particle size analysis showed that the thermally initiated MIP had a size ranging from 0.6-2.1 μ m while the NIP had a range of 1.6-5.2 μ m. For the UV initiated polymers the ranges were 0.6-1.9 μ m for the MIP and 0.8-2.8 μ m for the NIP.

Similar differences in particles size have been reported for a number of templates, e.g. sulfasalazine, where the MIP was larger than the NIP [44], theophylline, where the MIP was smaller than the NIP [55] and cinchonidine, where the non-imprinted polymers were monodisperse but those prepared in the presence of the template had a high degree of polydispersity [40]. While Wang *et al.* demonstrated that the morphology of the polymer was not affected to a large extent by the presence of the template [55], Yoshimatsu *et al.* discussed the effect of particle growth when the template was present in the pre-polymerisation solution [56]. They attributed this in part to the different forms of MAA in the pre-polymerisation solution for the MIP and the NIP. In the absence of the template is present interaction between the template and the monomer also plays a role in particle growth. Lai *et al.* also noticed a difference in the bead size of the MIP and NIP when imprinting di(2-ethylhexyl)phthalate [57]. In this instance, the difference in bead size was attributed to a difference in polarity of the polymerisation solution brought about by the presence of the template.

Another notable point observed in Figure 2.4 is that the beads formed during UV polymerisation were more monodisperse than those prepared by thermal methods but were much smaller also. This was confirmed by particle size analysis for the non-imprinted polymers in both cases (unless otherwise stated, discussion of particle size will state the size of the NIP as this negates the involvement of the template and demonstrates the effect of the polymerisation conditions on the polymeric system in question). The polymer prepared by thermal initiation had a particle size range of 1.6- $5.2 \mu m$ while the UV initiated polymers had a particle size range of 0.8-2.8 μm . It is necessary to note that differences in particle size from the dry (optical or scanning

electron microscopy) or solvated state (particle size analysis) are due to different swelling behaviours of the polymers. The particle size distributions of the polymers are shown in Figure 2.5.



Figure 2.5: Particle size distribution of (a) the thermally initiated N1 polymer and (b) the photochemically initiated N1 polymer.

It has been suggested that particle growth is dependent on exhaustion of the initiator, which in this case could indicate that there is a difference between the rate or level of initiation between UV and thermal methods [36]. It is possible that dissociation of the initiator occurs almost instantaneously for photochemical initiation causing the particles to start forming at the same time, whereas in thermal initiation, there may be some heterogeneity in terms of heat transfer. This would suggest a difference in the efficiencies of the initiator based on the method of initiation. This possible difference could determine the number and respective sizes of the beads, depending on the number of particles that start growing and the time at which they begin to grow. While temperature will affect the solubility of the growing polymer and may delay precipitation of the particles, Downey *et al.* suggest that the growth mechanism involves reaction of the vinyl groups on the surface of the particle with oligomers still dissolved in solution [38].

It was decided to employ simple UV-Vis spectroscopy to carry out equilibrium binding studies. To ascertain if this was a suitable method of analysis, the polymer was suspended in the rebinding solvent at the concentration at which the studies were to be carried out and to then investigate if there was any material present that would interfere with the absorbance spectrum of the template. During the course of this investigation it was observed that there was a considerable amount of material leaching out of the polymer after template extraction and washing procedures had been carried out. This was observed by leaving 50 mg of the treated polymer in the rebinding solvent for 24 hours. The result of this is presented in Figure 2.6.



Figure 2.6: UV-Vis spectra of the (a) M1 and (b) N1 thermally initiated polymers after initial washing was carried out. The peaks are at 210 nm and 270 nm.

Figure 2.6 is representative of the absorbance spectra obtained for the filtrate obtained from a suspension of polymer left overnight in solvent. The peaks were at 210 nm and 270 nm, which are the λ_{max} values for EGDMA and ACCN respectively. The polymers were analysed further using GC-MS based thermal desorption studies [58]. This allowed for an investigation of the material that was leaching from the polymer. A representative spectrum is shown in Figure 2.7.



Figure 2.7: Representative GC-MS thermal desorption profile for the M1 polymers with the fragmentation pattern showing evidence of the presence of EGDMA, having peaks with m/z values of 69, 113 and 170.

GC-MS thermal desorption profiles similar to that depicted in Figure 2.7 suggested that the main component of the polymeric bleed was the crosslinking monomer, EGDMA. This was not unexpected, as this was the main component of the polymers. It was postulated that this unreacted crosslinker was due to poor dissociation of the initiator and the presence of absorbance peaks at both λ_{max} values was due to oligomers and growing polymer chains with initiator molecules at the chain end. The possible poor dissociation of the initiator was proposed to be related to the temperature at which polymerisation was carried out. Since the decomposition temperature of ACCN was 88 °C, the synthesis was repeated at 80 °C (the boiling point of MeCN is 79 °C).

The microspherical polymeric particles prepared at 80 °C resulted in beads that were readily prepared, tested and cleaned of any partially polymerised material. The

increased temperature produced smaller beads, with particle size ranges of 0.46-1.15 μ m and 0.46-1.76 μ m for the MIP and NIP, respectively. This may have been due to the fact that greater rate of initiator dissociation had taken place, producing more radical centres thereby creating a greater number of beads but with a smaller average size. Tunc *et al.* have recently seen a similar effect when raising the polymerisation temperature from 60 to 80 °C for polymers using EGDMA or TRIM as the crosslinking monomers [59]. They also attributed the decrease in bead size to an increase in the number of radicals and, thus a smaller average bead size. This was similar to using UV as the method of initiation, which was further evidence that the original difference between thermal and UV initiation was due to the initiation efficiency. SEM images in Figure 2.8 demonstrate the difference in polymer morphology in the dry state.



Figure 2.8: SEM images of (a) N1 using ACCN as initiator and prepared at 60 °C and (b) N1 using ACCN as initiator and prepared at 80 °C.

2.4.1.1 Equilibrium binding studies on the M1 formulation

In order to test the polymers for selectivity and binding capacity the equilibrium binding studies in the porogen in which the polymers were prepared were carried out. Conventional wisdom in molecular imprinting states that rebinding is considered to be most successful in the porogen in which the polymer is made. Shea and Spivak postulate that the swelling of the polymer in the porogen gives the desired functional group proximity and location and therefore enhances binding site fidelity [60].

The studies were carried out as described in Section 2.2.4. Briefly, 50 mg of dry polymer were placed in 5 mL of template solution, agitated for 24 hours, centrifuged and the supernatant was filtered and analysed to test the concentration of template. Template solutions of 0.07 mM and 0.035 mM were prepared in MeCN. Using a

template solution of 0.035 mM equates to carrying out the binding study at a concentration of 1.5 mg of budesonide per gram of polymer. This would allow for a very low binding capacity for the imprinted polymers. A sensitivity study was carried out for the UV-Vis spectrophotometer and it was shown that a drop in concentration of only 5% (equating to a binding capacity of only 0.075 mg/g) could be observed. No affinity, either selective or non-selective was observed in the rebinding studies.

To ascertain the cause of this apparent lack of affinity a number of different rebinding strategies were applied. The first of these was to increase the concentration of the template in solution in a bid to drive the equilibrium towards binding. Concentrations of 0.5 mM and 1.0 mM were then tested. The binding studies were carried out exactly as before but in order to read the absorbance of the rebinding solvent, it had to be diluted prior to analysis. Again, no affinity was observed.

It was then postulated that the affinity of the template for the solvent was greater for the polymer. In order to test this, rebinding was carried out in a mixture of MeCN and H_2O in a 4:1 ratio, as BDN is practically insoluble in H_2O . This again showed no affinity. MeOH was also tested to determine whether or not a different polar solvent would drive the mainly hydrophobic template towards the relatively hydrophobic polymer. This again proved unsuccessful. At this stage it was decided to try a range of different polymer formulations and template analogues in an attempt to effect affinity for budesonide in a polymer.

2.4.2 Systematic synthesis and testing of polymers M1-M7

Once the initial polymer formulation had proved unsuccessful over a range of template concentrations and rebinding solvents, it was decided to systematically test a range of polymers by varying one factor at a time, to ascertain what properties were responsible for the lack of affinity and to use this information to prepare a budesonide selective polymer. A detailed list of the compositions of these polymers is presented in Table 2.4 while a summary of the results is presented in Table 2.6.

inity Particle
udies size (µm)
finity 0.3-1.2
ved
finity 1.7-4.7
ved
finity 0.5-3.6
ved
finity 1.5-3.7
ved
finity 0.2-0.9
ved
finity 0.5-6.6
ved
r f r

Table 2.6: Systematic investigation of polymer properties and their effect on template affinity (particle size ranges include smallest to largest for both MIP and NIP combined).

The polymers outlined in Table 2.6 were prepared in an effort to systematically investigate the properties of the polymer and method of polymerisation that may be responsible for the apparent lack of affinity for the template.

As discussed in detail in Section 2.4.1 the high degree of crosslinking monomer is typical for imprinting steroids. It was considered that this high degree of crosslinking monomer in the polymer composition may have caused inaccessibility to the binding sites. For this reason, the M2 polymers were prepared where the monomers remained the same but the ratio was changed from 1:10:50 to 1:10:30. No affinity was observed; ruling out problems with the degree of crosslinking monomer and the 1:5 ratio of functional to crosslinking monomer was continued for future investigations at this stage. It was deemed possible that the monomer may not have sufficient affinity for the template and might have caused the lack of affinity for budesonide. Since HEMA had been used successfully as a functional monomer for the structural analogue of the target, hydrocortisone [26], it was decided to test this as the functional monomer, for the M3 composition but using the same template to monomer ratio of 1:10:50. Again, MeCN was used as the porogen. While microspherical beads of a suitable size, 1.7-4.7 μ m, were obtained, no affinity was observed.

In early work on imprinting steroids by precipitation Mosbach *et al.* reported the use of a saturated template solution to enhance complex formation in such dilute conditions [20]. M4 polymers were prepared using the same ratio of functional to crosslinking monomer but with a saturated template solution. While this appeared to have the effect of forming more monodisperse polymeric beads, in the range of 0.5-3.6 μ m, it did not produce affinity in the resultant polymer, with no binding of the template observed to either MIP or NIP.

In order to determine whether it was the template itself causing the lack of affinity, i.e. due to lack of functionality or steric issues, the M5 formulation was tested. This formulation incorporated prednisolone (Figure 2.3), a close structural analogue of budesonide, which instead of the acetal grouping on the C16 and C17 positions, has an extra OH at C17. The composition of the polymer was the same in all other respects but again, no affinity was observed for either the imprinted or the non-imprinted polymer.

It has been reported that the use of MeCN alone can lead to non-porous polymeric particles when preparing precipitated beads using DVB as the crosslinking monomer [37]. In that study toluene was used as a co-solvent to impart porosity, as it was for EGDMA based polymers by Puoci *et al.* [44]. M6 polymers were prepared in 3:1 and 4:1 mixtures of MeCN/Toluene. It was also hoped that the presence of a non-polar co-solvent would enhance hydrogen bonding between the monomers and the template. Neither of these factors appeared to improve affinity and again, none was observed during rebinding studies.

M7 polymers were prepared using divinyl benzene as the crosslinking monomer to investigate the role of the crosslinking monomer. It was suggested that perhaps repulsion of the hydrophobic template from a relatively polar polymer molecule was taking place. This was also tested due to the fact that Wallimann *et al.* state that steroids are best bound to biological molecules comprised mainly of hydrophobic aromatic amino acid side chains incorporating a number of hydrogen bonding sites for interaction [1]. The ratio of functional to crosslinking monomer used for M7 polymers was 1:5 and budesonide saturated MeCN/TOL mixtures were used in 3:1 and 4:1 ratios. No affinity was observed for these polymers.

After the systematic study on a range of polymer compositions based on factors reported in the literature to affect polymerisation and affinity, a polymer with affinity for budesonide had not been successfully synthesised. Although a budesonide specific polymer was not prepared, during the course of synthesising polymers M1-M7 a good deal of information regarding the factors that affected the polymerisation process was obtained. This information would prove beneficial for the rapid optimisation of polymers once baseline affinity was achieved. The following sections present a summary of this information.

2.4.3 Factors affecting polymerisation

Starting from the initial M1 formulation a number of factors were varied to determine their impact on the resultant polymers. These factors included the type and concentration of components, such as the initiator and the monomer, as well as the solvent composition. Since no affinity was observed for any of the polymers formed, this could not be used as a guide to the success or otherwise of a polymer formulation. For this reason it was necessary to outline a set of parameters by which the success of the polymer formulations and conditions could be judged. To this end, for a polymer to be considered successful it was decided that the synthesised polymers must have the following properties:

- Bead size in the range of 0.5-10 μm (beads <0.5 μm could not be readily cleaned and analysed due to pore size of filters, etc.)
- The template must be readily extracted.
- The level of polymeric bleed after washing should be negligible (polymer bleed would be an undesirable characteristic for applications such as drug delivery)

Based on these guidelines the following factors were investigated for their affect on polymerisation conditions in this systematic study.

2.4.3.1 Concentration and type of initiator

As stated in Section 2.1 above, the role of the initiator is an important one in the preparation of polymers by precipitation methods [36, 38] and during the course of these investigations, this was seen to be the case, as the type and concentration of the initiator had a profound effect on the size and monodispersity of the polymeric beads. Figure 2.9 shows the structures of the three initiators used in this study.



Figure 2.9: Structures of the initiators used in this study: I azobisisobutyronitrile (AIBN); II 1,1'- azobis(cyclohexanecarbonitrile) (ACCN); III 4,4'-azobis(4-cyanovaleric acid) (CVA).

The 10 hour half-life decomposition temperatures of the initiators used in this study are 88 °C, 65 °C and 70 °C for ACCN, AIBN and CVA, respectively. While these temperatures are given for specific solvents, they serve as a guide as to the relative decomposition temperatures. A summary of the main observations on varying the initiator type are presented in Table 2.7.

Polymer	Solvent	Initiator ^a	Bead size ^b	Observation
formulation			(µm)	
M1	MeCN	ACCN	0.5-1.8	CVA produced the most
		(80 °C)		monodisperse beads of the
		CVA	1.6-5.4	largest size at the lowest
		AIBN	0.5-1.3	temp
M3	MeCN	ACCN	1.7-4.3	CVA produced the largest
		(80 °C)		beads at the lowest temp
		CVA	1.7-4.7	and were most readily
				cleaned
M7	MeCN	AIBN	2.2-4.1	AIBN produced the biggest
		CVA	1.7-3.1	particles for both solvent
	MeCN:TOL	AIBN	3.2-6.6	compositions
	(3:1)	CVA	1.3-3.5	

Table 2.7: Summary of the observations made upon investigation of type of initiator

^a Initiator concentration was maintained at 3 mol% and temperature was 60 °C in all cases except where otherwise stated.

^b Size range quoted for non-imprinted polymers only.

The variation in the bead size of the various formulations shown in Table 2.7 demonstrates that the initiator plays an important role in the resultant bead size when other parameters are kept constant. The first point to note is that due to the higher decomposition temperature of ACCN compared to CVA and AIBN, it was necessary to carry the polymerisation out at a higher temperature, to ensure complete polymerisation and to produce particles with a low level of polymeric bleed, as discussed in Section 2.4.1. Since this proved to be the case for both M1 and M3, the use of ACCN was

discontinued as high polymerisation temperatures could interfere with the interaction of the functional monomer and the template during polymerisation.

The use of CVA produced larger beads in the EGDMA based M1 formulation, 1.6-5.4 μ m, than the use of AIBN, 0.5-1.3 μ m. In the DVB based M7 formulation the reverse was observed with beads of 2.2-4.1 μ m and 1.7-3.1 μ m for AIBN and CVA, respectively, in MeCN and 3.2-6.6 μ m and 1.3-3.5 μ m for AIBN and CVA, respectively, in the MeCN/TOL mixture.

An explanation of the results observed is not straightforward, due to the complex factors involved in polymer particle growth, but a discussion of the fundamental aspects of polymerisation processes and precipitation methods in particular may be beneficial. During polymerisation, particle size is related to the rate of initiator decomposition and if the rate of decomposition is high, a large number of growing reactive centres or nuclei will be generated, which may lead to a higher rate of termination and low molecular weight polymers [59, 61]. Lower molecular weight polymer molecules will be more soluble in the polymerisation solvent and this will affect critical polymer chain length, thereby influencing the size of the polymer particles that precipitate [62].

In looking at the effects of the initiators in the two examples above, a direct comparison between the rates of decomposition cannot be made as the opposite effects took place with each formulation. This would indicate that the differences in bead size are related to the monomers used rather than the initiator. It may suggest, however, that the decomposition temperature or rate of initiation is altered by a change in the properties of the solvent, due to the presence of a particular monomer or set of monomers. While different rates of decomposition would lead to different degrees of aggregation early in the polymerisation, and therefore particle size, these effects may be offset by different rates of crosslinking, which is responsible for the stabilisation of particles against aggregation [36, 57]. This complex set of factors would suggest that for each polymer formulation, a number of small scale syntheses should be carried out to determine which initiator is most suitable for the particular formulation.

One notable difference, between precipitated beads and monolithic preparation strategies for imprinted polymers, is the amount of initiator used. The amount of initiator is considerably higher in the former, with values of 1-3 mol% of the polymerisable double bonds commonly used [38, 49, 56]. The effect of the initiator concentration was investigated for the M1 formulation only and it was observed that upon increasing the concentration of the initiator from 1 to 3 mol% the bead size range increased from 0.8-1.5 μ m to 1.6-5.4 μ m. This difference is demonstrated in the images presented in Figure 2.10.



Figure 2.10: Investigation into the effect of initiator concentration (a) M1 polymer prepared with 1 mol% CVA; (b) M1 polymer prepared with 3 mol% CVA.

It is readily observed in Figure 2.10 (a) and (b) that the change in the concentration of the initiator produces quite a notable difference in the bead size of two polymers prepared under identical conditions. This effect has also been reported by Li *et al.* and Tunc *et al.* and has been attributed to a number of reasons [59, 62]. Firstly, an increase in initiator makes the reaction more exothermic during the initial phase of the polymerisation, possibly leading to the formation of larger particles. A second possibility is that with an increase in the concentration of the initiator, the rate of radical creation and chain termination could be increased thus decreasing the average molecular weight of the polymer. A decrease in the molecular weight and perhaps, the degree of crosslinking, would make the growing polymer chain more soluble in the continuous phase. If the growing polymer chain is more soluble in the continuous phase, entropic precipitation of the bead may be delayed, thereby allowing the polymer particles are required, a higher concentration of initiator should be employed.

2.4.3.2 Monomer concentration

During the course of this work the growing polymers resulting from the formulations used, proved to be too soluble in a number of the solvents tested and failed to form precipitated microbeads. These solvents were dichloromethane, chloroform, DMSO and THF and despite testing the total monomer concentration over a range of 0.25-4% no beads were formed. The following discussion outlines the cases where beads were formed and the effects observed upon varying the monomer concentration in these solvents.

The concentration of the monomers in relation to the total volume of the solvent plays an important role in precipitation methodologies and it is generally considered that the concentration should be from 2-5% (w/v) [49]. The varying effects of monomer concentrations in different polymeric systems prepared by precipitation methods have been reported [35, 36, 59, 62, 63]. These varying effects are attributed to different solvation properties of the solvents for the monomers and growing polymers, as well as the ability of the growing polymer chains to aggregate and form large networks before entropic precipitation takes place.

The effect of varying monomer concentration was studied and an example of where a change in the monomer concentration influenced the resultant bead size is illustrated in Figure 2.11 while the observations are summarised in Table 2.8.



Figure 2.11: Change in bead size observed upon increasing the total monomer concentration of M7 from (a) 2% (w/v) to (b) 4% (w/v) in MeCN/TOL 3:1 (v/v).

Polymer	Solvent	Monomer	Polymer	Observation
		Conc.	morphology ^c	
		(% w/v)		
M1	MeCN	6	0.8-2.3	4% (w/v) was the
		5	0.6-1.5	optimum monomer
		4	1.6-5.4	concentration
		3	Soluble	
			polymer	
	MeCN/	4	SFG	This solvent
	$\mathrm{TOL}^{\mathrm{b}}$	3	Mixture	composition produced
		2	0.2-0.8	no suitable particles
		1	0.2-0.9	
M3	MeCN	4	SFG	1% (w/v) was the
		2	Mixture	optimum monomer
		1	1.7-4.7	concentration
		0.5	1.3-2.3	
M7	MeCN	4	1.8-3.42	4% (w/v) was the
		2	0.2-2.1	optimum monomer
	MeCN/	4	3.2-6.6	concentration for both
	TOL	2	1.5-6.1	solvent compositions
	(3:1)			

Table 2.8: Results obtained during investigation into the effects of monomer concentration^a

^a Reaction conditions: CVA was initiator for M1 and M3, AIBN was the initiator for M7; Initiator concentration was 3 mol% of polymerisable double bonds and initiation temperature was 60 °C;

^b Tested for ratio of 4:1, 3:1 and 2:1 and similar results observed .

 $^{\rm c}$ Bead size given in $\mu m;$ SFG: space-filling gel.; Mixture: combination of coagulum, fused particles and spherical beads

As is demonstrated in Table 2.8 and illustrated in Figure 2.11, the monomer concentration had a considerable effect on the resultant polymer morphology, with many smaller particles observed for M7 using 2% (w/v) compared to 4% (w/v). The 2% (w/v) concentration also contained more fused particles and particles with a less defined shape than the 4% (w/v). This may have been due, in part, to the solubility of the growing polymer in the continuous phase. Li and Stover report that coagulation of DVB

microspheres is prevented by their crosslinked surface and that the particles precipitate out of solution, due to insolubility, before coagulation can occur [36].

In formulations such as M1 in MeCN/TOL mixtures and M3 in MeCN, it was observed that at high monomer concentrations a coagulum (or a space-filling gel) was formed, while at lower concentrations discrete particles were prepared. This may be a result of an increased solvating power (due to changes in the monomer concentration) of the continuous phase leading to swelling of the particle surface and an increased likelihood of coagulation [36].

In some cases a shift from discrete particles to coagulum with an increase of just 0.1% was observed when using TRIM as a crosslinker [59]. The increased monomer concentration would also mean an increase in the concentration of the initiator in relation to solvent volume, thereby causing a more exothermic environment which may also affect the solvating ability of the continuous phase, and so, produce the same effect. Naka *et al.* also relate the formation of coagulated material upon increased monomer concentration, to solubility in the solvent and the fact that the polymer network grows at a faster rate than that which would allow discrete spherical particles to precipitate [34, 35].

In the case of M3 it can be seen that at 2% (w/v) a mix of coagulated material, agglomerates and spherical particles were obtained but on further reduction to 1% (w/v) particles of 1.7-4.7 µm were observed and reduction again to 0.5% (w/v) gave particles of 1.3-2.3 µm. A similar effect was observed for the M7 formulation where a decrease in monomer concentration resulted in a decrease in particle size, i.e. from 3.2-6.6 µm at 4% (w/v) to 1.5-6.1 µm at 2% (w/v), an effect also observed by Yang *et al.* [63]. This could suggest that as the concentration is reduced the growing particles are too dilute in solution to allow them to capture nascent oligomers, thereby preventing particle growth. The opposite effect was observed for M1 polymers however, in that as the concentration was increased from 4 to 6% (w/v), the bead size dropped from 1.6-5.4 µm to 0.8-2.3 µm. As with initiator type, the most suitable monomer concentration must be determined using small scale syntheses to test the effects of varying monomer concentration.

An understanding into the control of the factors affecting polymerisation is necessary for optimising bead formation for the required application; at this stage, however, there was still no affinity, specific or otherwise, for the template. To gain a better understanding of why this was the case, nuclear magnetic resonance (NMR) spectroscopy, was employed to investigate the interaction between the components in the pre-polymerisation solution.

2.4.4 NMR Spectroscopy

NMR is one of the most widely used techniques for the investigation of molecular interaction in organic chemistry and may be used to obtain information regarding complex stoichiometry in solution and the extent of that binding based on association constants [64-66]. In order to determine the different factors involved in molecular recognition and interaction a number of different experiments may be conducted by NMR. The Job plot, or method of continuous variation, is commonly used for determination of the complex stoichiometry [64, 65, 67]. The construction of a Job plot involves the preparation of a series of solutions where the mole fraction of the substrate and ligand vary but the total substrate and ligand concentration, together with the total volume, remains constant. A measurable property, e.g. the chemical shift of a proton, which changes in relation to the complex formed, is monitored and plotted against the mole fraction of either component. The maxima of the resultant plots are capable of yielding the ratio of substrate to ligand.

An NMR titration is commonly carried out to determine the strength of interaction based on the binding constants [64-66]. Here, the concentration of the substrate remains constant while the change in the chemical shift of a chosen proton or carbon nucleus is monitored upon each addition of a ligand/template solution. The application of linear or non-linear mathematical models to the resulting data allows for the determination of the binding constants and hence, an indication of the strength of interaction.

Since the performance of non-covalent molecularly imprinted polymers is generally considered to be influenced by the number, nature and extent of the molecular interactions between the functional monomer and the template in the pre-polymerisation reaction solution [68-70], the use of NMR studies to determine the nature and strength

of these molecular level interactions is therefore a valuable tool in both understanding the processes involved in molecular imprinting as well as the rational design of MIPs in general.

With this in mind, NMR spectroscopy was used to study the interaction of the template with the functional monomers (or analogues thereof) used in the polymerisation strategies previously outlined, in a bid to understand the reasons for the lack of affinity observed to this point. The construction of Job plots was carried out to determine the stoichiometry of the complexes that could possibly be formed in the pre-polymerisation solution, while NMR titrations were carried out to determine the strength of these interactions. While HEMA was used in these experiments deuterated acetic acid (AcOD) was used as a functional analogue of methacrylic acid (MAA) as this has been previously used to simplify the resulting NMR spectrum [71]. This is particularly useful given the complex spectrum for budesonide and the potential for overlapping of the vinyl protons of MAA with similar protons in budesonide. To aid discussion of the results, the NMR spectrum of budesonide, its chemical structure with the relevant positions numbered and potential sites of hydrogen bonding interaction outlined, are presented in Figure 2.12. (Assignment of the relevant protons was carried out by ¹H and COSY NMR experiments and was aided by analysis of literature reports for analogous compounds [72-75]).



Figure 2.12: Chemical structure of budesonide with the relevant positions numbered.

2.4.4.1 Job plot analyses

For Job plot analysis it is recommended to use the highest concentration possible for the method of analysis employed [67]. During the course of this study, however, it was observed that both the solvent and the functional monomer, or functional analogue used, played a role in both the solubility of the mixture and maintaining an appropriate shim lock on the solvent. For this reason the concentration values had to be adjusted, not only on changing the solvent but also on changing the functional monomer. Despite careful adjustment of these values, there appeared to be some problems with maintaining a shim lock on the solvent. This seemed to have been caused in part by abrupt changes in the composition of the mixture upon varying the ratio of template to functional monomer and in part by the occasional formation of precipitate in the solvent.

The precipitate was also a problem observed by Ansell *et al.* upon titration of itaconic acid and ephedrine where, as here, it did not hinder the collection of NMR data [76]. To counter these problems a manual gradient shim was carried out on a mixture containing both the functional monomer and the template. The obtained shim map was then carefully fine-tuned by manual shimming to obtain the maximum lock on the mixture and in turn produce the most accurate spectrum. This was quite effective and allowed for the collection of reliable NMR data.

The Job plot analyses carried out looked at the interactions of the functional monomers in both acetonitrile and chloroform. In acetonitrile, the Job plots suggested no straightforward interaction between either HEMA or AcOD with no single maximum obtained for either monomer. This suggested that the polar solvent was preventing any meaningful or predictable interactions between budesonide and the monomers. Despite this setback, it was considered worthwhile to investigate any possible interactions that could take place in the less polar solvent, chloroform. For this reason, the remainder of the discussion is focussed on the interactions that were observed in deuterated chloroform. In the less polar solvent the situation was somewhat different, with points of interactions observed for both AcOD and HEMA and the template. Job plots of the type shown in Figure 2.13, for the interaction of budesonide and HEMA in CDCl₃, were observed in both cases and the results obtained for budesonide with AcOD and HEMA in CDCl₃ are summarised in Table 2.9.



Figure 2.13: Representative Job plot for the interaction between 0.03M BDN and 0.03 M HEMA in CDCl₃ with maxima observed at χ_{BDN} = 0.5 for both plots.

AcOD			HEMA		
χbdn	Inference	χbdn	Inference		
0.3	2:1 complex due to	0.5	1:1 complex with		
	interaction at C3 carbonyl		interaction at OH11		
	and OH11				
0.5	1:1 complex with interaction n.d				
	at C3 carbonyl				
0.5	1:1 complex with interaction n.d				
	at C3 carbonyl				
o -		.11.			
0.5	1:1 complex with interaction	**	Possible interaction at		
	at OH11		OH11		
*	Possible interaction at either	0.5	1:1 complex with		
	C20 carbonyl or OH21 interaction at OH21				
	 χBDN 0.3 0.5 0.5 0.5 * 	AcODχBDNInference0.32:1complexduetointeractionatC3carbonylandOH11	XcODXBDNInferenceXBDN0.32:1complexdueto0.5interaction at C3 carbonylinteraction at C3 carbonyland OH110.51:1 complex with interactionn.dat C3 carbonylat C3 carbonyln.d0.51:1 complex with interactionn.dat C3 carbonylx**at OH11x***Possible interaction at either0.5C20 carbonyl or OH21C20C20		

Table 2.9: Summary of Job plot analyses for budesonide with AcOD and HEMA in CDCl₃.

* no plot constructed but signal shifted and changed from a triplet to a broad singlet

** complex multiplet that shifted but was unreliable for tracking and construction of plot n.d.: no data The results presented in Table 2.9 serve to demonstrate that interaction did take place between the monomers/analogues tested and the template. The results also suggest that the interactions between the template and the monomers were different from each other. For AcOD maxima were observed for H2 and H4 at 0.5 for the mole fraction of budesonide. This suggests a 1:1 complex between the template and the AcOD. This would be expected as the carbonyl on the A-ring of the steroid lies between these two protons and it would be expected that the acetic acid would interact at this point. The maximum for the H1 proton was at 0.3, suggesting a 2:1 interaction of monomer:template.

There was some evidence of interaction at the OH11 position with a 1:1 complex predicted by the construction of a Job plot for H11 and a maximum observed at 0.5. The possibility of interaction at the C3 carbonyl and the OH11 position could be a reason for the 2:1 prediction for the H1 proton. An example of this in the literature can be seen in the binding of 3-aminopyrazole with different amide protons of *N*-acetyl-*L*-valyl-*L*-valine methyl ester, as studied by Kirsten and Schrader and discussed by Wulff and Knorr [77]. Here the overlaid Job plots of the two amide protons showed a 1:1 complex forming for one of the protons and a 2:1 complex forming with a much lower overall change in chemical shift. This would be analogous to the observed interactions for BDN and AcOD in CDCl₃.

As well as the evidence suggesting two sites of interaction, analysis of the NMR spectra of the Job plots suggested a possible third site of interaction. For the NMR spectrum of budesonide alone, a triplet was observed at 3.03 ppm. Analysis of separate NMR spectra determined that this was the signal of the OH21 proton, which it is predicted, would be involved in intramolecular hydrogen bonding to the oxygen of the C20 carbonyl. This intramolecular bond enables the splitting of the signal by the two H21 protons. This was confirmed by 1-D COSY experiments. The signal was seen as a triplet rather than a quartet due to the overlapping of the two centre peaks of the quartet. Upon increased presence of the AcOD, this peak shifted downfield and broadened to the typical peak seen for an OH proton. This suggests the breaking of the intramolecular hydrogen bond caused by preferential interaction at either the C20 carbonyl or the OH21 proton.

When HEMA was used as the functional monomer, the interaction observed at the OH21 site was more pronounced, with definite shifts observed in the signal for the OH21 proton, allowing for the construction of a Job plot giving a maximum at a budesonide mole fraction of 0.5. While no observable shifts were present for the H2 or H4, suggesting no interaction at the C3 carbonyl, there was a shift, again more pronounced, for the H1 proton, this time suggesting a 1:1 interaction at the OH11 site. Due to the fact that these sites were at the antipodes of the molecule, the combined 1:1 interactions could in fact suggest a 2:1 monomer:template interaction, as opposed to the possible 3:1 monomer:template interaction observed for AcOD. To further investigate these interactions and to try and determine their magnitude, NMR titrations were then carried out in chloroform.

2.4.4.2 NMR titration analysis for budesonide and d₄-acetic acid

Due to the fact that AcOD produced no signal in the resultant NMR spectra it was possible to follow a number of template peaks of interest and it is worth noting that while many of the overall chemical shifts observed are relatively small (*vide infra*) the plots presented in Figure 2.14 suggest that the results are meaningful.



Figure 2.14: Overlaid plots of the change in chemical shift of the protons: H1, H2, H4 and H11 observed in the titration of BDN with AcOD in CDCl₃, where FM/T is the ratio of functional monomer to template.

The plots of the H2 and H4 vinyl protons of the A-ring in BDN changed in quite similar fashions during the titration with AcOD. H1 also acted in quite a similar manner at the

start of the titration but changed slightly after an FM/T ratio of approximately 25. Even at functional monomer to template ratios of 110, saturation had still not been reached for any of the vinyl protons and the overall change in the chemical shift of the protons are: 0.113, 0.101 and 0.102 ppm for H1, H2 and H4 respectively. These were all downfield shifts suggesting that the nuclei deshielded to a greater extent possibly due to the presence of the electronegative carboxyl group of AcOD in close proximity while binding with the carbonyl at C3.

These changes are relatively low when compared with changes in shifts of 1.2 ppm observed for the NH₂ protons of 2-aminopyridine upon interaction with AcOD observed by Osmani *et al.* [71], 2 ppm for NH₂ protons of trimethoprim interacting with MAA [78] and 3.23 and 4.55 ppm for the OH3 proton of estradiol with AcOD and deuterated pyridine respectively, in toluene [79]. Svenson *et al.* observed shifts of up to 1.2 ppm for protons in the interaction of nicotine and AcOD at functional monomer to template ratios as low as 35 [80]. It is necessary to note, however, that strong H-bonding would be expected between carboxyl and amino groups.

For H11, which is an indicator of the interaction at the OH11 proton, the overall change in shift was smaller again, with a value of 0.033 ppm. This change in chemical shift is in an upfield direction which indicates that the H11 nucleus is in a less deshielded environment. This may be due to the fact that the O atom of the OH21 is involved in hydrogen bonding with AcOD, thus diminishing some of its deshielding effects. The interaction does, however, reach saturation at an FM/T ratio of about 40. It is worth noting that the differences in magnitude between the overall change in chemical shifts values correspond with the results seen for the Job plots, where the shift change was again, considerably smaller for H11 than for the vinyl protons. This indicates that interaction at the C3 carbonyl is favoured over interaction at the OH11 proton.

Another point worth noting is that the slight difference in the plot for H1 plus the slightly larger change in shift (~10%) in comparison to the other two vinyl protons may also be an indication of the effect on H1 by the interaction at OH11. Further evidence of the presence of higher order complexes, i.e. 1:2 as well as 1:1, is provided by comparison of the relative changes in complexation induced shifts for the different protons. As stated by Connors [64] and utilised by Quaglia *et al.* [78], the ratio of the

change in shifts due to complexation, should be the same for two different protons when only 1:1 complexes occur. This is not the case when 1:2 complexes occur and the ratio is expected to vary with the change in the concentration of the ligand. This was shown to be the case for the interaction between BDN and AcOD when the ratio: $\Delta\delta/\Delta\delta_{max}$, for each proton was plotted against the total concentration of AcOD in a manner analogous to that employed by Quaglia *et al.* [78], as illustrated in Figure 2.15.



Figure 2.15: Comparison of plots of the relative changes in shifts of the protons of BDN against the total concentration of AcOD in CDCl₃.

The occurrence of non-parallel plots for the relative changes in shifts of the protons in Figure 2.15, suggests the presence of higher order complexes in the interaction between AcOD and BDN [78]. This combination of evidence suggests that there are at least two points of interaction between AcOD and BDN and a possible third based on the effect on the OH21 resonance signal as discussed in Section 2.4.2.1.

In order to determine the association/dissociation constants, the data was manipulated using non-linear regression in Prism, GraphPad software 4 [80, 81]. The likelihood of weak complexation, due to small shift changes, presents its own problem in that it is necessary to use high monomer concentrations to drive the equilibrium [64]. This can lead to problems with medium effects and possible self-association of the ligand. Where a system may contain 1:1 and 1:2 complexes, the binding constants should be considered as approximate values rather than exact numbers due to the fact that an increase in the number of parameters (because of the increased number of complexes)

can affect the reliability of the data [64, 82]. This being said, the values of K_{diss} for the protons affected by interaction during the titration are given in Table 2.10.

Proton	K _{diss} / M
H1	1.094 ± 0.11
H2	0.732 ± 0.08
H4	0.775 ± 0.09
H11	0.534 ± 0.04

Table 2.10: K_{diss} values for the protons in which a complex-induced change in chemical shift occurred during titration with AcOD in CDCl₃.

Table 2.10 shows that the value for H1 indicates weaker association than that observed for H2 and H4, while the latter two protons have very similar K_{diss} values. This would be expected as H1 is further away from the expected site of interaction. The low association predicted by these values are in agreement with the small values for the change in chemical shift. The value for H11 points to slightly higher association which would be in agreement with the achievement of saturation for this proton over the concentration range used.

2.4.4.3 NMR titration analysis for budesonide and HEMA

There were a number of differences observed in the titration of BDN and HEMA in CDCl₃ in comparison to the BDN and AcOD titration in CDCl₃; one such difference involved the signals for H2 and H4. In this case, there was no observable pattern in the change in shift, suggesting that no interaction took place at the C-3 carbonyl. This was in agreement with the findings from the Job plots. For this reason only the H1, H11 and OH21 peaks are presented.

Another difference lay in the fact that it was not possible to carry out the titration of BDN and HEMA to the same extent as BDN and AcOD. This was because once the ratio of 1:90 had been reached it was not possible to get an appropriate shim-lock and the spectra could not be considered to be reliable. It was possible to track the change in chemical shift of the H1 and H11 signals up to the ratio of 1:90 but this was not the case with the OH21 proton, which could only be tracked to the 1:50 ratio due to overlap of

peaks from the HEMA. The HEMA peak was at a much greater intensity than the OH21 peak due to the far greater concentration in the mixture. An overlay of the plots is given in Figure 2.16.



Figure 2.16: Overlaid plots of the changes in chemical shift in the H1, H11 and OH21 protons of BDN, during titration of BDN and HEMA in CDCl₃.

There was a considerable difference between the overall change in shift of the OH21 proton and the H1 and H11 protons. The overall value for the OH21 proton was a shift of 0.584 in a downfield direction. This suggested that the proton was becoming more deshielded, i.e. was in a more electronegative environment. This indicated that that the proton was taking part in a hydrogen bond with an O atom on the HEMA molecule. While it is not possible to track this peak to the full extent of the titration, the obtained data suggests that this interaction is the strongest of the interactions seen thus far.

The overall changes in chemical shifts of the H1 and H11 protons had values of 0.119 ppm downfield and 0.067 ppm upfield respectively. The direction of chemical shift change is analogous to the changes observed for the same nuclei in the titration with AcOD. This would suggest that again, the OH11 atom was involved in hydrogenbonding to a H atom of an OH group on the HEMA. This was thus brought in close proximity to H1, thereby bringing it into a more electronegative environment. At the same time, this would reduce some of the deshielding power of the OH11 oxygen causing the upfield shift in H11. Overall chemical shift values were larger in the HEMA titration, albeit only by 5% in the case of the H1 shift change but this value increased to

50% in the case of the H11 shift change. What may be more noteworthy is the fact that this occurred at a 1:90 ratio, for HEMA and a 1:106 for AcOD. The overall size of these shift changes were still quite small in relation to the majority of the values quoted in the literature for NMR titrations, as the examples given in Section 2.4.4.2 demonstrate.

While it may appear from Figure 2.16 that the plots of the chemical shift change for H1 and H11 were similar, when these plots are put on a similar scale it can be seen that this was not the case, as shown in Figure 2.17.



Figure 2.17: Plot of H1 and H11 protons of BDN obtained during titration of BDN and HEMA in CDCl₃.

Figure 2.17 suggested that, as with AcOD, the interaction for H11 may have neared saturation, while this was not the case for H1 although the overall change in shift was greater for the latter. As discussed in Section 2.4.4.2 a comparison of the ratio of the change in chemical shift values yields information as to the number of complexes present [64, 78]. This has also been applied to the interaction of BDN with HEMA, the plots for which are presented in Figure 2.18.



Figure 2.18: Relative changes for the complex-induced shifts for each of the protons plotted against the total concentration of the functional monomer, HEMA, in CDCl₃.

As before, a difference in the relative complexation induced shifts for the different protons suggests the presence of higher order complexes, e.g. 1:2. This was also seen to be the case in Figure 2.18 suggesting the presence of more than a 1:1 complex. Since there was evidence of interaction at two sites, i.e. OH11 and OH21, in the Job plots shown in Figure 2.13, where shift changes occurred at the H1 and OH21 protons, this would be in agreement with the data presented in Figure 2.18. The K_{diss} values calculated for the three protons are presented in Table 2.11.

Table 2.11: Kdiss values determined for H1, H11 and OH21 using Prism, GraphPad software.

Proton	K _{diss} / M
H1	1.465 ± 0.06
H11	0.726 ± 0.08
OH21	2.130 ± 0.11

The results for the apparent K_{diss} seen in Table 2.11 show that the values obtained for H1 and H11 are quite similar to those obtained for the titration of BDN and AcOD, given in Table 2.10. It must be noted that while the value for OH21 would appear to give very weak association, it must be taken into consideration that the full set of values

were not obtained because of overlapping peaks; the size of the complex induced shift, however, suggested the possibility that the interaction may have been stronger.

2.4.4.4 Summary of NMR titration investigations

A comparison of the effects seen during titration of HEMA, AcOD and BDN in CDCl₃ is presented in Table 2.12.

Proton	AcOD				HEMA	A	
	$\Delta\delta_{max}$	K _{diss} /M	Inference	$\Delta \delta_{max}$	K _{diss} /M	Inference	
			Interaction		1 465	Interaction	
H1	0.113	1.094 ± 0.11	at C3 CO	0.119	+ 0.06	at OH11	
			and OH11		± 0.00		
110	0.102	0.732 ± 0.08 Interaction at C3 CO	Interaction	nd	nd	nd	
Π2	0.102		at C3 CO	n.a.	n.a.	11. u .	
114	0 101	0.775 ± 0.09	Interaction	nd	nd	nd	
П4	0.101		at C3 CO	n.u.	n.a.	11. u .	
H11	0.033	0.534 ± 0.04	Interaction	0.067	0.726	Interaction	
			at OH11	0.007	± 0.08	at OH11	
01101			1	,	0.504	2.130	Interaction
OH21	n.c.	n.d.	n.d.	0.584	± 0.11	at OH21	

Table 2.12: Comparison of data obtained during titration of BDN with HEMA and AcOD in CDCl₃.

n.d.: no data; n.c.: no curve, i.e. interaction detected but not possible to track change.

The proposed interactions summarised in Table 2.12 are depicted in Figure 2.19.



Figure 2.19: Proposed interactions for (a) AcOD and BDN and (b) HEMA and BDN

Table 2.12 outlines the results obtained during the NMR investigations of AcOD and HEMA with BDN in CDCl₃. In all cases the values for the dissociation constants are high when compared to values obtained by other groups. As no titration data for these monomers and corticosteroids have been reported, it is necessary to draw comparisons with studies carried out for different templates used in molecular imprinting. For example Urraca *et al.* obtained values for K_{diss} as low as 2.12×10^{-5} M for interaction between 1-allylpiperazine and the template cyclododecyl 2,4-dihydroxybenzoate [82] while Ansell and Kuah obtained values of 1.4×10^{-4} M for interaction between ephedrine and MAA [83]. Interestingly, O'Mahony *et al.* obtained values as low as 0.045 M for π - π interactions (a relatively weak interaction compared to hydrogen bonding, etc.) between 4-vinyl pyridine and 2,4-dichlorophenoxy acetic acid [81]. This suggests that with dissociation constants as high as were observed in Table 2.12, any interaction between template and functional monomer could be considered very weak. This indication of very weak interaction between the template and monomers, even in

non-polar solvent such as chloroform, may explain the lack of any apparent affinity in the polymers, M1-M7.

The work of Long *et al.*, discussed in Section 1.5.2.2, where the degree of interaction between a template, testosterone, and the functional monomer, MAA, was seen to be relatively weak, only 20% of the template was shown to be bound to the polymer after 24 hours of reaction in a precipitation polymerisation process [84]. In this and any analogous situation, such as the one presented here, the number of binding sites on the polymer itself is probably relatively low. With this in mind, a number of new precipitated polymers were prepared and subsequent binding studies were carried out at a much lower concentration. For quantitative analysis at such low concentrations and with the possibility of interfering compounds (such as polymeric bleed) it was necessary to employ HPLC for analysis of the rebinding solutions.

2.4.5 Initial studies towards the use of a template analogue approach

UV-Vis analysis had shown that a drop in concentration of 75 μ g in the binding solution could be detected but, as stated above, it was now presumed that this could possibly be far higher than the actual capacity of the polymers. The more sensitive method of HPLC, was therefore employed for the quantitative analysis of the steroid in the binding solution. The construction of standard curves using HPLC showed that analysis of budesonide, based on a method by Faouzi *et al.* [47], was linear down to at least 0.25 μ g/mL. Robust and reliable detection at this level allowed for the quantitation of any material bound to the polymers, even at very low levels.

Given the fact that NMR studies had shown that interaction between MAA and HEMA would be very weak in polar solvents, it was decided to first test non-imprinted polymers using the crosslinking monomer alone, in aqueous solutions of an organic solvent. This was an effort to promote the hydrophobic interactions, which would most likely predominate with a molecule such as a steroid, with its largely hydrocarbon skeleton. A polymer, M8 (see Table 2.4), composed of DVB only was prepared in acetonitrile-toluene at a ratio of 3/1 (v/v). In aqueous solutions of methanol, this DVB NIP bound 100% of budesonide at concentrations up to 20 µg/mL. When polymers prepared with EGDMA only were tested at these concentrations, only 30% of the

material was bound to the polymer. For this reason a number of polymers were prepared using DVB as a crosslinking monomer and MAA as the functional monomer.

When HPLC analysis was carried out on a number of the previously prepared polymers (M1 and M7), it was observed that this more sensitive technique revealed a small but significant quantity of template bleeding. This fact, coupled with the by now prohibitive cost of budesonide (€1000 per gram), suggested long-term use of a template analogue strategy. The template analogue approach has been utilised in molecular imprinting applications for a range of templates [25, 85, 86] and can be necessary for imprinting when template bleeding poses a significant problem or when using the actual target is unsuitable due to factors such as cost or toxicity. Prednisolone, as shown in Figure 2.3, has a structure very similar to that of budesonide with only the acetal grouping and alkyl chain attached to the D ring of the steroid in the difference. While it was considered that this steric factor could play a role, the difference in the retention times of these two analytes (~5 minutes), as opposed to more sterically similar steroids made prednisolone an attractive template analogue. Thus, polymers M9 and M10, i.e. polymers with ratio of prednisolone:MAA:DVB of 1:10:30 and 1:10:50, respectively, were prepared, using acetonitrile-toluene 3:1 (v/v) as the porogen.

As this was a target analogue approach, budesonide was again used as the target for equilibrium binding studies. These binding studies were carried out as before, with the exception that acetonitrile or acetonitrile-water 90:10 (v/v) was used and 30 mg of polymer were added to 0.9 mL of binding solution over a range of template solutions (2-100 μ g/mL). While no repeatable imprinting effect was observed for the M10 polymer, some degree of imprinting seemed to be present for the M9 polymer. In acetonitrile, at a budesonide concentration of 100 μ g/mL an average imprinting factor of 1.15 was obtained. Similarly, in the 10% water solution at a concentration of 10 μ g/mL, an imprinting factor of 1.46 was observed. While these results seemed somewhat promising, the actual difference between the percentage rebound for the MIP and NIP was very low, for example, in 10% water, the MIP bound 6.3% while the NIP bound 5.2%. Low as this was, however, it was repeatable, and suggested that it was possible to use the combination of a template analogue and a method of enhancing template-monomer interaction to effect some degree of molecular imprinting and subsequent rebinding of a target steroid.

2.5 Conclusions

The above investigations were carried out in a bid to prepare a polymer selective for budesonide. The initial polymer formulation was selected on the basis of published methods for the imprinting of structural analogues of budesonide. When no affinity was observed, the components and parameters were varied in a systematic fashion in a bid to discern the reason for this apparent lack of affinity. Despite this systematic study of polymer composition and reaction conditions, no affinity was observed for initial polymer formulations.

During the systematic study, however, considerable information was obtained as to the factors which play a role in the preparation of successful polymers, in relation to bead size, ease of preparation and lack of polymeric bleed, etc. Due to the complex set of factors involved in this method of polymerisation, no clear trends were observed. It could be said, however, that for each combination of functional and crosslinking monomer, the solvent composition, monomer concentration, initiator type and concentration must be carefully chosen to achieve the most desired result. The ease of this method and the lack of labour intensive processes such as grinding and sieving should allow for rapid and efficient optimisation of the polymer compositions and conditions for future work and the knowledge gained from these preliminary investigations should prove applicable and beneficial to future polymer synthesis once affinity has been achieved.

NMR studies were applied to the pre-polymerisation solutions to determine possible causes for this apparent lack of affinity. These studies indicated the presence of weak interactions between the template and the functional monomers used, which could lead to a low number of binding sites within the polymeric matrix. This rational understanding of the limitations of corticosteroid imprinted polymers by precipitated means led to studies using much lower concentrations than had previously been employed. These studies revealed that the polymers had a degree of inherent affinity for the target which, while low, suggested the potential to improve this affinity by molecular imprinting means.

Problems with template bleed and cost suggested the use of a target analogue approach, and subsequent experiments using prednisolone as the template indicated some initial imprinting effects. Low though these effects were, they suggested that specific rebinding of budesonide using a target analogue approach was possible. These results, coupled with the information gleaned from NMR studies suggested that an approach that would promote the relatively weak interaction between the template and functional monomers in the pre-polymerisation studies was required to enhance the imprinting effect observed.
References

- 1. Wallimann P, Marti T, Furer A, Diederich F: Steroids in Molecular Recognition. *Chemical Reviews* 1997, 97:1567-1608.
- 2. Davidson L, Hayes W: Molecular imprinting of biologically active steroidal systems. *Current Organic Chemistry* 2002, **6**:265-281.
- 3. Sobande PO, Kercsmar CM: Inhaled corticosteroids in asthma management. *Respiratory Care* 2008, **53:**625.
- 4. Kompella UB, Bandi N, Ayalasomayajula SP: **Poly(lactic acid) nanoparticles for sustained release of budesonide.** *Drug Delivery Technology* 2001, **1**:29-35.
- 5. Olejnik O, Hughes P: **Drug delivery strategies to treat age-related macular degeneration.** *Advanced Drug Delivery Reviews* 2005, **57:**1991-1993.
- 6. Salvador JP, Estevez MC, Marco MP, Sanchez-Baeza F: A new methodology for the rational design of molecularly imprinted polymers. *Analytical Letters* 2007, **40**:1294-1306.
- 7. Zhu Q, Wang L, Wu S, Joseph W, Gu X, Tang J: Selectivity of molecularly imprinted solid phase extraction for sterol compounds. *Food Chemistry* 2009, **113**:608.
- 8. Zhongbo Z, Hu J: Selective removal of estrogenic compounds by molecular imprinted polymer (MIP). *Water Research* 2008, **42:**4101.
- 9. Wilson IA, Arevalo JH, Stura EA, Taussig MJ: *J Mol Biol* 1993, **231**:103-118.
- 10. Sandor G: Advances in the analysis of steroid hormone drugs in pharmaceuticals and environmental samples (2004-2010). *Journal of Pharmaceutical and Biomedical Analysis* 2011, **55**:728-743.
- 11. Bystrom SE, Borje A, Akemark A: Selective reduction of steroid 3-ketones and 17-ketones using LiAlH₄ activated template polymers *J Am Chem Soc* 1993, 115:2081-2083.
- 12. Whitcombe MJ, Rodriguez ME, Villar P, Vulfson EN: A New Method for the Introduction of Recognition Site Functionality into Polymers Prepared by Molecular Imprinting: Synthesis and Characterization of Polymeric Receptors for Cholesterol. Journal of the American Chemical Society 1995, 117:7105-7111.
- 13. Ramstrom O, Ye L, Mosbach K: Artificial antibodies to corticosteroids prepared by molecular imprinting. *Chemistry & Biology* 1996, **3**:471-477.
- 14. Ciardelli G, Borrelli C, Silvestri D, Cristallini C, Barbani N, Giusti P: Supported imprinted nanospheres for the selective recognition of cholesterol. *Biosensors & Bioelectronics* 2006, **21**:2329-2338.
- 15. Lee WC, Hung CY, Hwang CC: Separation of cholesterol from other steroids using molecularly imprinted polymer prepared by seeded suspension polymerization. *Chemical And Biochemical Engineering Quarterly* 2008, 22:151-156.
- 16. Tong YJ, Guan HM, Wang SF, Xu JW, He CB: Syntheses of chitin-based imprinting polymers and their binding properties for cholesterol. *Carbohydrate Research* 2011, **346:**495-500.
- 17. Spizzirri UG, Peppas NA: Structural analysis and diffusional behavior of molecularly imprinted polymer networks for cholesterol recognition. *Chemistry of Materials* 2005, **17:**6719-6727.

- 18. Celiz MD, Aga DS, Colón LA: Evaluation of a molecularly imprinted polymer for the isolation/enrichment of [beta]-estradiol. *Microchemical Journal* 2009, **92:**174.
- 19. Shimon F, Bernard SG, Abraham JD: Selective 17-beta-estradiol molecular imprinting. *Journal of Polymer Science Part A: Polymer Chemistry* 2009, 47:5534-5542.
- 20. Ye L, Cormack PAG, Mosbach K: Molecularly imprinted monodisperse microspheres for competitive radioassay. *Analytical Communications* 1999, 36:35-38.
- 21. Wei ST, Molinelli A, Mizaikoff B: Molecularly imprinted micro and nanospheres for the selective recognition of 17 beta-estradiol. *Biosensors & Bioelectronics* 2006, **21**:1943-1951.
- 22. Noss KR, Vaughan AD, Byrne ME: Tailored binding and transport parameters of molecularly imprinted films via macromolecular structure: The rational design of recognitive polymers. Journal Of Applied Polymer Science 2008, 107:3435-3441.
- 23. Mirmandieh S, Mardihallaj A, Hashemian Z, Razavizadeh J, Ghaziaskar H, Khayamian T: Analysis of testosterone in human urine using molecularly imprinted solid-phase extraction and corona discharge ion mobility spectrometry. *Journal of Separation Science* 2011, **34**:107-112.
- 24. Bui BTS, Merlier F, Haupt K: Toward the Use of a Molecularly Imprinted Polymer in Doping Analysis: Selective Preconcentration and Analysis of Testosterone and Epitestosterone in Human Urine. *Analytical Chemistry* 2010, 82:4420-4427.
- 25. Baggiani C, Giraudi G, Trotta F, Giovannoli C, Vanni A: Chromatographic characterization of a molecular imprinted polymer binding cortisol. *Talanta* 2000, **51**:71-75.
- 26. Sreenivasan K: Application of molecularly imprinted polymer as a drug retaining matrix. *Angewandte makromolekulare Chemie* 1997, **246:**65-69.
- 27. Sreenivasan K: On the application of molecularly imprinted poly(HEMA) as a template responsive release system. *Journal of Applied Polymer Science* 1999, **71:**1819-1821.
- 28. Kitahara K-I, Yoshihama I, Hanada T, Kokuba H, Arai S: Synthesis of monodispersed molecularly imprinted polymer particles for highperformance liquid chromatographic separation of cholesterol using templating polymerization in porous silica gel bound with cholesterol molecules on its surface. *Journal of Chromatography A* 2010, **1217**:7249-7254.
- 29. Gadzala-Kopciuch R, Ricanyova J, Buszewskia B: Isolation and detection of steroids from human urine by molecularly imprinted solid-phase extraction and liquid chromatography. Journal Of Chromatography B-Analytical Technologies In The Biomedical And Life Sciences 2009, 877:1177-1184.
- 30. Lin Y, Shi Y, Jiang M, Jin Y, Peng Y, Lu B, Dai K: **Removal of phenolic** estrogen pollutants from different sources of water using molecularly imprinted polymeric microspheres. *Environmental Pollution* 2008, **153**:483.
- 31. Wang C, Javadi A, Ghaffari M, Gong S: A pH-sensitive molecularly imprinted nanospheres/hydrogel composite as a coating for implantable biosensors. *Biomaterials* 2010, **31:**4944-4951.
- 32. Ye L, Weiss R, Mosbach K: Synthesis and characterization of molecularly imprinted microspheres. *Macromolecules* 2000, 33:8239-8245.

- 33. Wei S, Mizaikoff B: **Binding site characteristics of 17[beta]-estradiol imprinted polymers.** *Biosensors and Bioelectronics* 2007, **23:**201.
- 34. Naka Y, Kaetsu I, Yamamoto Y, Hayashi K: Preparation of Microspheres by Radiation-Induced Polymerization .1. Mechanism for the Formation of Monodisperse Poly(Diethylene Glycol Dimethacrylate) Microspheres. Journal of Polymer Science Part a-Polymer Chemistry 1991, 29:1197-1202.
- 35. Naka Y, Yamamoto Y: Preparation of Microspheres by Radiation-Induced Polymerization .2. Mechanism of Microsphere Growth. Journal of Polymer Science Part a-Polymer Chemistry 1992, 30:1287-1298.
- Li K, Stover HDH: Synthesis of Monodisperse Poly(Divinylbenzene) Microspheres. Journal of Polymer Science Part a-Polymer Chemistry 1993, 31:3257-3263.
- 37. Li WH, Stover HDH: **Porous monodisperse poly(divinylbenzene) microspheres by precipitation polymerization.** Journal of Polymer Science Part a-Polymer Chemistry 1998, **36:**1543-1551.
- Downey JS, Frank RS, Li WH, Stover HDH: Growth mechanism of poly(divinylbenzene) microspheres in precipitation polymerization. *Macromolecules* 1999, 32:2838-2844.
- 39. Qi PP, Wang JC, Wang LD, Li Y, Jin J, Su F, Tian YZ, Chen JP: Molecularly imprinted polymers synthesized via semi-covalent imprinting with sacrificial spacer for imprinting phenols. *Polymer* 2010, **51**:5417-5423.
- 40. Liu Y, Hoshina K, Haginaka J: Monodispersed, molecularly imprinted polymers for cinchonidine by precipitation polymerization. *Talanta* 2010, **80:**1713-1718.
- Cirillo G, Iemma F, Puoci F, Parisi OI, Curcio M, Spizzirri UG, Picci N: Imprinted hydrophilic nanospheres as drug delivery systems for 5fluorouracil sustained release. *Journal of Drug Targeting* 2009, 17:72 - 77.
- 42. Pérez-Moral N, Mayes AG: Comparative study of imprinted polymer particles prepared by different polymerisation methods. *Analytica Chimica Acta* 2004, **504:**15.
- 43. Mohajeri SA, Karimi G, Aghamohammadian J, Khansari MR: Clozapine Recognition via Molecularly Imprinted Polymers; Bulk Polymerization Versus Precipitation Method. *Journal of Applied Polymer Science* 2011, 121:3590-3595.
- 44. Puoci F, Iemma E, Muzzalupo R, Spizzirri UG, Trombino S, Cassano R, Picci N: Spherical molecularly imprinted polymers (SMIPs) via a novel precipitation polymerization in the controlled delivery of sulfasalazine. *Macromolecular Bioscience* 2004, **4**:22-26.
- 45. Oh JK, Drumright R, Siegwart DJ, Matyjaszewski K: **The development of microgels/nanogels for drug delivery applications.** *Progress In Polymer Science* 2008, **33**:448-477.
- 46. Robinson DN, Peppas NA: **Preparation and Characterization of pH-Responsive Poly(methacrylic acid-g-ethylene glycol) Nanospheres.** *Macromolecules* 2002, **35:**3668-3674.
- 47. Faouzi MA, Dine T, Luyckx M, Brunet C, Gressier B, Cazin M, Wallaert B, Cazin JC: High-performance liquid chromatographic method for the determination of budesonide in bronchoalveolar lavage of asthmatic patients. *Journal of Chromatography B: Biomedical Sciences and Applications* 1995, 664:463.

- 48. Puoci F, Iemma F, Cirillo G, Trombino S, Cassano R, Picci N: Molecularly imprinted polymers for selective adsorption of cholesterol from aqueous environment. *E-Polymers* 2007:-.
- 49. Wang JF, Cormack PAG, Sherrington DC, Khoshdel E: Synthesis and characterization of micrometer-sized molecularly imprinted spherical polymer particulates prepared via precipitation polymerization. *Pure and Applied Chemistry* 2007, **79:**1505-1519.
- 50. Bravo JC, Garcinuno RM, Fernandez P, Durand JS: Selective solid-phase extraction of ethynylestradiol from river water by molecularly imprinted polymer microcolumns. *Analytical And Bioanalytical Chemistry* 2009, 393:1763-1768.
- 51. Sherrington DC: Preparation, structure and morphology of polymer supports. *Chemical Communications* 1998:2275-2286.
- 52. Esfandyari-Manesh M, Javanbakht M, Atyabi F, Badiei A, Dinarvand R: Effect of Porogenic Solvent on the Morphology, Recognition and Release Properties of Carbamazepine-Molecularly Imprinted Polymer Nanospheres. Journal of Applied Polymer Science 2011, 121:1118-1126.
- 53. Rachkov AE, Cheong SH, El'skaya AV, Yano K, Karube I: Molecularly imprinted polymers as artificial steroid receptors. *Polymers For Advanced Technologies* 1998, **9**:511-519.
- 54. Sellergren B, Shea KJ: Influence Of Polymer Morphology On The Ability Of Imprinted Network Polymers To Resolve Enantiomers. *Journal Of Chromatography* 1993, 635:31-49.
- 55. Wang JF, Cormack PAG, Sherrington DC, Khoshdel E: Monodisperse, molecularly imprinted polymer microspheres prepared by precipitation polymerization for affinity separation applications. *Angewandte Chemie-International Edition* 2003, **42**:5336-5338.
- 56. Yoshimatsu K, Reimhult K, Krozer A, Mosbach K, Sode K, Ye L: Uniform molecularly imprinted microspheres and nanoparticles prepared by precipitation polymerization: The control of particle size suitable for different analytical applications. *Analytica Chimica Acta* 2007, **584**:112.
- 57. Lai J-P, Yang M-L, Niessner R, Knopp D: Molecularly imprinted microspheres and nanospheres for di(2-ethylhexyl)phthalate prepared by precipitation polymerization. Analytical and Bioanalytical Chemistry 2007, 389:405-412.
- 58. Cummins W, Duggan P, McLoughlin P: Thermal desorption characterisation of molecularly imprinted polymers. Part I: a novel study using direct-probe GC-MS analysis. Analytical And Bioanalytical Chemistry 2008, 391:1237-1244.
- 59. Tunc Y, Ulubayram K: Production of Highly Crosslinked Microspheres by the Precipitation Polymerization of 2-(Diethylamino)ethyl Methacrylate with Two or Three Functional Crosslinkers. Journal of Applied Polymer Science 2009, 112:532-540.
- 60. Spivak DA, Shea KJ: Investigation into the scope and limitations of molecular imprinting with DNA molecules. *Analytica Chimica Acta* 2001, 435:65-74.
- 61. Sellergren B, Hall AJ: Chapter 2. In Molecularly Imprinted Polymers: Man Made Mimics of Antibodies and their Applications in analytical chemistry. Volume 23. Edited by Sellergren B. Amsterdam: Elsevier; 2001: Techniques and Instrumentation in Analytical Chemistry].

- 62. Li P, Rong F, Yuan CW: Morphologies and binding characteristics of molecularly imprinted polymers prepared by precipitation polymerization. *Polymer International* 2003, **52:**1799-1806.
- 63. Yang M, Gu W, Sun L, Zhang F, Ling Y, Chu X, Wang D: Study on the molecularly imprinted polymers with methyl-testosterone as the template. *Talanta* 2010, **81:**156-161.
- 64. Connors KA: Binding Constants. New York: Wiley; 1987.
- 65. Fielding L: Determination of Association Constants (Ka) from Solution NMR Data. *Tetrahedron* 2000, **56:**6151.
- 66. Atwood JL: *Comprehensive Supramolecular Chemistry*. 1st edn. Amsterdam: Elsevier; 1996.
- 67. Huang CY: Determination Of Binding Stoichiometry By The Continuous Variation Method The Job Plot. *Methods In Enzymology* 1982, 87:509-525.
- 68. Andersson HS, Nicholls IA: Spectroscopic Evaluation of Molecular Imprinting Polymerization Systems. *Bioorganic Chemistry* 1997, 25:203.
- 69. Nicholls IA, Adbo K, Andersson HS, Andersson PO, Ankarloo J, Hedin-Dahlström J, Jokela P, Karlsson JG, Olofsson L, Rosengren J, et al: Can we rationally design molecularly imprinted polymers? *Analytica Chimica Acta* 2001, 435:9.
- 70. Alexander C, Andersson HS, Andersson LI, Ansell RJ, Kirsch N, Nicholls IA, O'Mahony J, Whitcombe MJ: Molecular imprinting science and technology: a survey of the literature for the years up to and including 2003. *Journal Of Molecular Recognition* 2006, **19**:106-180.
- 71. Osmani Q, Hughes H, Flavin K, Hedin-Dahlstrom J, Allender CJ, Frisby J, McLoughlin P: The use of FTIR and NMR spectroscopies to study prepolymerisation interactions in nitrogen heterocycles. *Analytical and Bioanalytical Chemistry* 2008, **391**:1229-1236.
- 72. Martini S, Bonechi C, Casolaro M, Corbini G, Rossi C: Drug-protein recognition processes investigated by NMR relaxation data: A study on corticosteroid-albumin interactions. *Biochemical Pharmacology* 2006, 71:858.
- 73. Arbez-Gindre C, Berl V, Lepoittevin J-P: Air oxidation of 17hydroxycorticosteroids catalyzed by cupric acetate: formation of hemiacetal dimers. *Steroids* 2003, 68:361.
- 74. Rachwal S, Pop E, Brewster ME: Structural studies of loteprednol etabonate and other analogs of prednisolone using NMR techniques. *Steroids* 1996, 61:524.
- 75. Liguori A, Perri F, Siciliano C: **D-homoannulation of 17α,21-dihydroxy-20**keto steroids (corticosteroids). *Steroids* 2006, **71:**1091-1096.
- 76. Ansell RJ, Wang DY: Imprinted polymers for chiral resolution of (+/-)ephedrine. Part 3: NMR predictions and HPLC results with alternative functional monomers. *Analyst* 2009, **134**:564-576.
- 77. Wulff G, Knorr K: Stoichiometric noncovalent interaction in molecular imprinting. *Bioseparation* 2001, **10**:257-276.
- 78. Quaglia M, Chenon K, Hall AJ, De Lorenzi E, Sellergren B: Target Analogue Imprinted Polymers with Affinity for Folic Acid and Related Compounds. J Am Chem Soc 2001, 123:2146-2154.
- 79. Idziak I, Benrebouh A, Deschamps F: Simple NMR experiments as a means to predict the performance of an anti-17[alpha]-ethynylestradiol molecularly imprinted polymer. *Analytica Chimica Acta* 2001, **435**:137.

- 80. Svenson J, Karlsson JG, Nicholls IA: **1H Nuclear magnetic resonance study of the molecular imprinting of (-)-nicotine: template self-association, a molecular basis for cooperative ligand binding.** *Journal of Chromatography A* 2004, **1024:**39.
- 81. O'Mahony J, Molinelli A, Nolan K, Smyth MR, Mizaikoff B: Towards the rational development of molecularly imprinted polymers: 1H NMR studies on hydrophobicity and ion-pair interactions as driving forces for selectivity. *Biosensors and Bioelectronics* 2005, 20:1884-1893.
- 82. Urraca JL, Carbajo MC, Torralvo MJ, González-Vázquez J, Orellana G, Moreno-Bondi MC: Effect of the template and functional monomer on the textural properties of molecularly imprinted polymers. *Biosensors and Bioelectronics* 2008, 24:155.
- 83. Ansell RJ, Kuah KL: Imprinted polymers for chiral resolution of (+/-)ephedrine: understanding the pre-polymerisation equilibrium and the action of different mobile phase modifiers. *Analyst* 2005, **130**:179-187.
- 84. Long YY, Philip JYN, Schillen K, Liu F, Ye L: Insight into molecular imprinting in precipitation polymerization systems using solution NMR and dynamic light scattering. *Journal Of Molecular Recognition* 2011, 24:619-630.
- 85. Kareuhanon W, Lee VS, Nimmanpipug P, Tayapiwatana C, Pattarawarapan M: Synthesis of Molecularly Imprinted Polymers for Nevirapine by Dummy Template Imprinting Approach. *Chromatographia* 2009, **70**:1531-1537.
- 86. Matsui J, Fujiwara K, Takeuchi T: Atrazine-Selective Polymers Prepared by Molecular Imprinting of Trialkylmelamines as Dummy Template Species of Atrazine. Analytical Chemistry 2000, 72:1810.

Chapter 3

The synthesis of a molecularly imprinted polymer using the target analogue approach

3. The synthesis of a molecularly imprinted polymer using the target analogue approach

3.1 Introduction

3.1.1 Target analogue imprinting

Since Andersson *et al.* first utilised a target analogue as the template for the synthesis of a sameridine selective polymer [1], the use of the so-called 'dummy template' or 'template analogue' approach has become widespread in molecular imprinting [2-4]. (As the preceding sentence suggests, this approach can be described using a number of names. In this chapter, these names will be interchanged but as the template used for all polymers is an analogue of the primary target, 'target analogue' is primarily used). As with the first occurrence of this method, the vast majority of work employing this strategy involves solid phase extraction, where the presence of even trace quantities of template 'bleed' will negatively effect the ultimate quantitation of the analytes of interest, often only present themselves at part per billion or part per trillion levels [5-7]. This technique is used, however, for other applications, such as the synthesis of a molecularly imprinted membrane by Wang *et al.* for lovastatin acid using lovastatin as a target analogue [8] or the preparation of explosive sensing technologies, using carboxylic acid analogues of the targets by Riskin *et al.* [9].

As stated above, where trace quantities of analytes are to be detected, any template that remains in the polymer matrix, commonly quoted as greater than 1% [7, 10], has the potential to leach from the matrix and interfere with quantitation, thus necessitating the use of a target analogue as the template species. Despite this being the predominant reason for their use, target analogues may also be employed for a variety of other reasons; Khorrami *et al.*, for example, used oxindole as a cost effective target analogue for the toxin patulin [11], while Kubo *et al.* cite toxicity or rarity of targets as suitable reasons for utilising a target analogue [12]. Whatever the reason for its use, the process of molecular imprinting dictates that there are certain criteria that the target analogue must fulfil to make it suitable for use, including functionality, as well as molecular size and shape. An example of this can be seen in the aforementioned work by Andersson *et*

al., where the template analogue differed from the target by only the substitution of a methyl group for an ethyl group [1].

While many template analogues are commercially available, special requirements may necessitate the synthesis of a particular compound, such as the work of Kubo *et al.* where the intended target was a cyanobacterium toxin [12]. The toxin contained opposing ionic groups at a particular interval and the target analogue was synthesised to have the same interval between ionic groups, thereby organising the binding sites in such a way that the target could be recognised based on this "interval immobilisation". As discussed in Section 1.6.1, Baggiani *et al.* have also synthesised a "template mimic", a chlorinated azo-dye, for targeting Sudan dyes in food samples [13]. Interestingly, this imparted selectivity to the polymer not just for the main target, Sudan I, but also group selectivity for several of its analogues. Group selectivity was also imparted upon polymers prepared by Baggiani *et al.* for the recognition of corticosteroids, where hydrocortisone, or cortisol, was used as a template and a vast range of polymers were studied for their selectivity towards fourteen related corticosteroids [14].

3.1.2 MIP Chromatography

Chromatography, as a method for evaluating the success of the templating process for MIPs, has been in use for almost as long as the modern era of molecular imprinting itself. Since Wulff *et al.* first applied affinity chromatography to a ground, sieved and packed imprinted polymer to evaluate the racemic separation of D- and L-4-nitrophenyl mannopyranoside [15, 16], this technique has been consistently employed for investigations into the factors contributing to the recognition capabilities of imprinted polymers [17-22].

Despite the usefulness of this technique in comparing MIP to NIP and investigating the fidelity of binding recognition, MIP chromatography as an analytical method in its own right has a number of problems. The previously mentioned work by Wulff *et al.* described the presence of "broad peaks with strong tailing" and attributed this to the mechanisms by which binding takes place in these covalent polymers and the resulting exchange equilibria [15]. While early work by Sellergren *et al.* suggested that these problems could be overcome by using non-covalent interactions [17], these problems

have continued to plague the use of molecularly imprinted polymers as successful stationary phases and, over a decade and a half later, the same author cited ten problems preventing their use for commercial applications, including binding site heterogeneity, slow mass transfer and swelling/shrinkage problems [23].

In a bid to overcome some of these problems, a range of techniques have been employed with considerable focus on particle morphology and size distribution, factors which the use of ground and sieved particles exacerbate. To this end, polymerisation techniques including those discussed in detail in Section 1.5.2, i.e. suspension, precipitation and core-shell methods, have been investigated for use as polymeric stationary phases. Some notable examples include the separation of racemic ketoprofen on surface imprinted core-shell beads by Jiang *et al.* [24] and the separation of cholesterol from a number of other steroids, on beads prepared by seeded suspension by Lee *et al.* [25]. The separations achieved by these groups are exemplified by Figure 3.1.



Figure 3.1: Separations achieved on molecularly imprinted stationary phases for (a) racemic nonsteroidal anti-inflammatory drugs [24] and (b) for a range of steroids [25].

For applications where a molecularly imprinted stationary phase is not the ultimate aim, MIP chromatography has been ubiquitously used throughout the field as a reliable method of investigating polymeric properties [22, 26, 27]. Sellergren *et al.* have used chromatographic evaluation for imprinted polymers prepared for phosphorous esters [28] and riboflavin [29], while Baggiani *et al.* have employed MIP chromatography for evaluation of polymers prepared for a number of templates, including fungicides [30],

polycyclic aromatic hydrocarbons [31] and the steroid, hydrocortisone [32]. Indeed, the effectiveness of the imprinting process for a range of steroids has been studied using MIP chromatography, including cholesterol [33], estradiol [27] and testosterone and its derivatives [34]. The latter example succeeded in separating testosterone and epitestosterone on a MIP based column.

The retention behaviour of an analyte on different polymer columns can be used to draw conclusions about the molecular recognition properties of the materials. It can often be necessary, however, to carry out physical characterisation of the materials to ensure that any differences observed between the polymers are due to differences in the manner and extent of chemical interactions between the polymers, rather than simply physical differences [35]. A number of physical characterisation strategies are outlined in Section 1.6, including scanning electron microscopy (SEM), binding site characterisation and nitrogen sorption porosimetry. To supplement the brief introduction to the latter technique, given in Section 1.6, a more detailed discussion is presented in Section 3.1.3.

3.1.3 Nitrogen sorption porosimetry

The physisorption or adsorption of gases, typically nitrogen, is a general method for the physical characterisation of materials, capable of providing information on the surface area and porosity of the materials analysed [36]. As such, this method has been widely applied to molecularly imprinted polymers [37-39]. This technique has been used to analyse polymer morphologies and in so doing to gain a more detailed understanding of the physical factors that can affect molecular recognition. This technique involves the construction of adsorption-desorption isotherms by exposing a fixed mass of dry material to a gas at a constant temperature and monitoring the equilibrium pressures as a series of known quantities of gas are added to the sealed chamber [36, 40, 41]. The isotherms can provide information on specific surface area, pore diameter and volume, as well as pore size distribution. Regarding porous materials the following classifications have been made by the IUPAC relating to pore sizes [36]:

- Micropores: pores with widths not exceeding about 2 nm
- Mesopores: pores with widths between 2 nm and 50 nm
- Macropores: pores with widths exceeding about 50 nm

The isotherms obtained from gas sorption studies follow six basic types, as shown in Figure 3.2 (a).



Figure 3.2: (a) Six basic types of physisorption isotherms and (b) four possible types of hysteresis loops obtained [36].

The reversible isotherms are defined by the IUPAC as follows [36]:

- Type I isotherms are observed for microporous solids that have relatively small external surfaces.
- Type II isotherms are the typical form of the isotherm obtained for a non-porous or macroporous adsorbent. This isotherm represents unrestricted monolayer-multilayer adsorption.
- Type III isotherms are uncommon and represent materials where adsorbateadsorbate interactions are important.
- Type IV isotherms are characterised by their hysteresis loop, a characteristic related to capillary condensation taking place in mesopores, and as such, this type of isotherm indicates a mesoporous material.
- Type V isotherms are related to Type III and are again uncommon.
- Type VI isotherms represent stepwise multilayer adsorption on non-porous surfaces.

The different shapes of the hysteresis loops obtained, as shown in Figure 3.2 (a), while not fully understood, can be used to identify specific pore structures [36], for example H1 type hysteresis can be associated with porous materials consisting of agglomerates while porous adsorbents with poorly defined size and shape distribution can give H2 type loops.

In order to determine the surface area of the materials analysed by such methods, the Brunauer-Emmett-Teller (BET) method [42] has been widely used and has been applied to molecularly imprinted materials [36, 37, 43, 44]. The BET method employs the following equation in the linear form:

$$\frac{p}{n^{a}(p^{0}-p)} = \frac{1}{n^{a}.c} + \frac{(c-1)p}{n^{a}_{m}.c p^{0}}$$
 Equation 3.1

Where:

P is the equilibrium pressure;

 P_o is the nitrogen saturation pressure at 77 K;

 n^{a} is the amount adsorbed at the relative pressure P/P_o;

 n_m^a is the monolayer capacity;

C is a constant dependent on the shape of the isotherm.

To calculate the surface area, the BET plot is constructed using the linear relationship between $p/n^a(p^0-p)$ and p/p^0 with the range of linearity restricted to the linear part of the isotherm, typically between 0.05 an 0.30 [36].

The surface area is then determined using the following equation:

$$a_s = n_m^a L a_m$$
 Equation 3.2

Where:

 a_s is the surface area;

L is Avogadro constant;

 a_m is the molecular cross-sectional area of the adsorptive.

It is necessary to note that Sing states that the BET method, while generally suitable for Type II and Type IV isotherms as long as the C value is neither too high nor too low, it is not considered likely that it will yield reliable values for the actual surface area of Type I or Type III isotherms [36]. It is also worth mentioning the limitations of nitrogen sorption data relating to MIPs, in that the information gleaned from such experiments relates to polymers in the dry state. While this information is important and it allows us to draw comparisons between the MIP, the NIP and other polymers, it must be remembered that the polymers are typically used in conjunction with a solvent. As such, conclusions are drawn on information acquired from physisorption experiments in the absence of solvent and these conclusions are used to rationalise polymer behaviour in the presence of solvent.

3.2 Research aims and objectives

The aims and objectives of this chapter were as follows:

- To prepare molecularly imprinted polymers for corticosteroids with a particular focus on budesonide by use of the target analogue imprinting approach.
- To test the prepared polymers using MIP chromatography and so ascertain the binding characteristics of the polymers and the factors driving recognition.
- To carry out a physical characterisation on the polymers prepared in order to gain a full understanding of the morphological effects of different polymerisation conditions and to investigate the effect of such differences on molecular recognition.

3.3 Experimental

3.3.1 Materials

Hydrocortisone-17-butyrate (HCB), hydrocortisone (HYD), hydrocortisone-21-acetate (HAC) and prednisolone-21-acetate (PAC) (all \geq 97% purity or better) and trimethylolpropane trimethacrylate (TRIM) and pentaerythritol triacrylate (PETRA) (both technical grade) were all purchased from Sigma-Aldrich, Wicklow, Ireland. All other materials and reagents used were as per Table 2.3 and treated as per Section 2.3.1.

3.3.2 Instrumentation

Instrument	Model	Supplier
Photochemical	RMR-600	Branford, CT, USA
mini-reactor (8 W		
lamps)		
Column packer	Preparatory pump	JVA analytical, Ireland
Nitrogen sorption	Micromeritics	Particular sciences,
porosimeter	Gemini VI	Ireland
Scanning electron	Hitachi S-2460N	Oxford instruments,
microscope		England
Shaking incubator	C24	New Brunswick
		Scientific, USA
HPLC	1200	Agilent Technologies,
		Ireland
Particle size	Mastersizer 2000S	Malvern, Ireland
analyser		

Table 3.1: Instrumentation used for this study.

3.3.3 Preparation of polymers

Polymers were prepared as per Table 3.2

Polymer	Template	Functional	Crosslinking	Molar	Solvent	Polymer
		monomer	monomer	ratio ^a		format ^b
MB3	HCB	MAA	EGDMA	1:10:30	MeCN	Bulk
MB4	HCB	MAA	DVB	1:10:30	MeCN/	Bulk
				TOL		
					3:1	
					(v/v)	
MB5	HCB	MAA	TRIM	1:10:15	MeCN	Bulk
MB6	HCB	MAA	PETRA	1:10:15	MeCN	Bulk
MB8	HCB		EGDMA	1:0:30	MeCN	Bulk
MB10	HCB	HEMA	EGDMA	1:10:30	MeCN	Bulk
MB12	HCB	MAA	EGDMA	1:10:30	CHCl ₃	Bulk
MB13	HCB	MAA	TRIM	1:10:15 CHCl ₃		Bulk
M13	HCB	MAA	EGDMA	1:10:30	MeCN	PPT
M14	HCB	MAA	DVB	1:10:30	MeCN/	PPT
					TOL	
					3:1	
					(v/v)	
M23	HCB	MAA	TRIM	2:4:2	DCM	PPT

Table 3.2: Polymer compositions and conditions used for this study

^a Ratio of template:functional monomer:crosslinking monomer; ^b PPT: precipitation

Polymers MB3-MB13 were prepared as bulk monolith polymers at 60 °C for 24 hours at a monomer concentration ranging between 75 and 85% (w/v). These polymers were crushed and roughly sieved before Soxhlet extraction in MeOH for 24 hours before final grinding and sieving to size ranges of either 25-75 μ m or 38-53 μ m. Fine particles were sedimented in MeOH-H₂O 80/20 (v/v). Polymers M13 and M14 were prepared as per Section 2.3.2 by thermal initiation and at a monomer concentration of 4% (w/v). Precipitated beads, M23, were prepared using a method based on that employed by Ye *et al.* [45]. Dichloromethane was used as the porogen and the polymerisation was carried out at 4 °C and photochemically initiated at 350 nm. Polymerisation was carried out in a round-bottom flask in the UV reactor while being rotated around its long axis at

a speed of 20 RPM. The monomer was 2% (w/v) and precipitated beads were cleaned on-line by washing the column with acidified MeOH with 5% AcOH at 0.2 mL/min until a stable baseline was achieved.

3.3.4 Chromatographic polymer evaluation

3.3.4.1 Column packing

 50×4.6 mm i.d. columns were used in all cases for polymer packing. Polymers were first suspended and sonicated in the packing solvent, which was composed of either aqueous solutions of MeOH and MeCN or MeCN alone. This slurry was then added to the packing chamber and the solvent was pumped through the column, at flow rates which varied between 0.2 and 0.5 mL/min for precipitated beads and 18 mL/min for ground bulk particles, until three chamber volumes had been pumped through. The polymers were tested for packing uniformity by equilibrating the column with acetonitrile at the required analytical flow rate and injecting 5 μ L of a 20 μ L/mL solution of acetone in the mobile phase and monitoring the retention time and shape of the resulting peak.

3.3.4.2 Zonal chromatography

Suitably packed columns were equilibrated with the mobile phase, either MeCN or MeCN/H₂O, at a flow rate of 0.5 mL/min for bulk particles or 0.2-0.5 mL/min for precipitated beads, until a stable baseline was achieved. Stock steroid solutions were prepared at a concentration of 2.5 mM (ca. 1 mg/mL for each steroid) in acetonitrile and stored at -20 °C. Steroid solutions for injection onto the imprinted and blank columns were prepared by carrying out a 1 in 50 dilution on the stock solution, giving a final concentration of 5×10^{-5} M for each analyte. 5 µL of the steroid solutions were injected on to the column and eluted at the flow rates listed above, while monitoring the absorbance at 240 nm. Injections were carried out in duplicate, or triplicate where necessary, to ensure repeatability. The capacity factor, *k'*, was calculated using Equation 3.3.

$$k' = \frac{(t-t_0)}{t_0}$$
 Equation 3.3

Where k' is the capacity factor or retention factor, t, is the retention time for the analyte and t_0 is the retention time for the column void marker, obtained by injecting 5 µL of a 20 µL/mL acetone solution. Equation 3.4 was used to calculate the imprinting factor (IF) for each steroid.

$$IF = \frac{k'_{MIP}}{k'_{NIP}}$$
 Equation 3.4

where k'_{MIP} and k'_{NIP} are the capacity factors obtained on the MIP and NIP column, respectively.

3.3.4.3 Frontal chromatography

Staircase frontal chromatography was carried out on all polymers as described by Hall *et al.* [46]. This technique acquires analyte adsorption data over a wide range of concentrations, typically covering at least three orders of magnitude. Here, three stock solutions of HCB were prepared in acetonitrile at concentrations of 0.001, 0.01 and 0.1 mM. These solutions were used as the mobile phase in a step gradient using acetonitrile as the co-solvent. Beginning with 100% acetonitrile and the lowest concentration template solution, the composition of the mobile phase was changed in a stepwise fashion by increasing the percentage of the template solution until 100% template solution was flowing through the column. Without washing the column, this was continued with the higher concentration template solutions, resulting in three separate staircase-type chromatograms, each comprised of ten steps.

The results obtained by frontal chromatography are plotted in binding isotherms that were subsequently fitted to the Freundlich adsorption model using Prism, GraphPad software 4.0 and thermodynamic constants were derived using affinity distribution analysis (FIAD), as described by Rampey *et al.* [47]. The Freundlich isotherm is shown in Equation 3.5.

$$q = aC^m$$
 Equation 3.5

where q is the amount of analyte bound on the stationary phase and C is the concentration of free steroid, while a and m are fitting constants. The concentration of

analyte bound to the stationary phase for each step, q^* , is calculated using the following equation:

$$q^* = \Delta C_m (V_{1/2} - V_0)$$
 Equation 3.6

where ΔC_m is the change in concentration of analyte in the mobile phase at each step, $V_{1/2}$ is the retention volume at half-height of the corresponding step and V_0 is the void volume of the system including the column. V_0 was measured by either using the halfheight of an un-retained analyte, i.e. acetone, or by measuring the volume of the system and the column separately, using acetone as a void marker.

3.3.5 Nitrogen sorption porosimetry

All polymers were dried under vacuum at 50 °C and degassed overnight at 50 °C under a constant flow of nitrogen. A known amount of polymer, between 50-200 mg, was placed in the analysis tube and nitrogen physisorption isotherms were measured at 77 K. The BET method, as described in Section 3.1.3, was used to calculate the specific surface area while the Barret-Joyner-Halenda (BJH) method was applied to physisorption isotherms to determine pore size distribution as carried out by Urraca *et al.* [38].

3.3.6 Particle size and scanning electron microscopy analysis

All particle size and SEM analyses were carried out as per sections 2.3.3 and 2.3.5.

3.4 Results and discussion

3.4.1 Target analogue selection

Initial efforts towards the use of a target analogue to prepare molecularly imprinted polymers selective for corticosteroids, with a particular focus on budesonide, employed prednisolone as the template for use as a target analogue. While the imprinting factors obtained were relatively low, i.e. 1.15-1.46, they suggested that the use of a cost effective target analogue, coupled with a means of enhancing the weak template-monomer interactions in the pre-polymerisation solution, would be a viable approach for effecting corticosteroid selectivity within imprinted polymers. In order to choose the most suitable target analogue, a number of suitably priced and structurally analogous steroids were investigated. Five corticosteroids, bearing similar functionalities and ranging in price from ten to forty times less than budesonide, were selected as possible target analogues. These compounds were also used for an investigation of the factors driving recognition and selectivity. The steroids used in this study are presented in Figure 3.3.



Figure 3.3: Corticosteroids used in this study: I Budesonide (BDN); II Hydorcortisone-17-butyrate (HCB); III prednisolone-21-acetate (PAC); IV hydrocortisone-21-acetate (HAC); V prednisolone (PRD); VI hydrocortisone (HYD)

As discussed in Section 3.1, the target analogue must fulfil certain criteria for it to be suitable for use. As such, the target analogue must have similar size, shape and functionality to the target. As can be seen from Figure 3.3, all of the steroids have similar structure and functionality on rings A-C. In the case of I budesonide (BDN), III prednisolone-21-acetate (PAC) and V prednisolone (PRD), there are two double bonds in the A-ring of the steroid. This is opposed to only one for steroids II hydrocortisone (HYD), IV hydrocortisone-21-acetate (HAC) and VI hydrocortisone-17-butyrate (HCB). The second double bond being positioned between C-1 and C-2 on the former group of steroids. The effect of this structural difference has been discussed by Baggiani *et al.* where the planar configuration of the A ring, resulting from the second double bond, was seen to have a considerable affect on polymer recognition and determined to be "one of the main discriminating structural differences of a steric nature" for MIPs with group selectivity for corticosteroids [14].

Figure 3.3 shows that it is on the D-ring of the steroid that a number of differences exist between the principal target, budesonide, and the structural analogues. BDN possesses a hydroxyl group on the C-21 position, as do steroids, **II**, **V** and **VI**, while **III** and **IV** have bulky acetate groupings at this position. It is worth noting that evidence of interaction at OH-21 was observed using NMR investigations, as reported in Section 2.4.4 and disruption of this interaction with an acetate group, would be expected to have a deleterious effect on recognition for the target. While steroids **III-VI** have another hydroxyl group on the 17 α position, BDN possesses an acetal grouping on positions 16 and 17, forming an extra ring structure, from which a propyl chain extends. The presence of this bulky substituent in **I** and the lack of any such substituents on **III-VI** would be expected to play a major role in the steric recognition for the steroid and it was postulated that the creation of an imprinted site without the proper steric requirements might inhibit recognition. With this in mind, the presence of a bulky hydrophobic group in this position, as with **II** above, would be considered to be advantageous.

With a view to predominantly aqueous biological applications for the corticosteroid selective polymers and preliminary results showing that hydrophobic interactions would be important, it was deemed pertinent to also take the hydrophobicity of the steroids into consideration. The predicted log P values were calculated using ChemBioDraw

Ultra 11.0, giving a value of 2.73 for BDN and 2.86 for **II**, while steroids **III-VI** all had much lower predicted log P values than BDN, with values ranging from 1.3-1.7. This was considered to be an appropriate method for log P predictions as the predicted value for prednisolone was 1.71, which was in close agreement to the literature value of 1.62 [48]. The combination of these factors suggested hydrocortisone-17-butyrate (HCB) as a likely target analogue wit NMR studies carried out to further investigate its suitability.

3.4.1.1 NMR studies using HCB

Job plot experiments were carried out with HCB and deuterated acetic acid in deuterated chloroform in a manner analogous to that described in Section 2.4. While many of the shift changes that occurred could not be accurately mapped due to overlapping signals, the shift change for the H4 proton was readily tracked and the Job plot obtained suggested a possible 1:1 interaction between template and monomer, as shown in Figure 3.4.



Figure 3.4: Job plot obtained for interaction between HCB and AcOD in $CDCl_3$ with a maximum at χ HCB 0.5 suggesting a possible 1:1 interaction taking place at the C-3 carbonyl.

While the Job plot obtained suggested the likelihood of a 1:1 interaction between HCB and AcOD, when a titration was carried out the points of interaction between HCB and deuterated acetic acid were similar to those observed for BDN and acetic acid under similar conditions (as outlined in Section 2.4.4). It must be noted that there was potential evidence of 1:2 interaction, with the possible presence of another maximum at 0.7, however, further investigation would be necessary to determine this.

As with BDN, evidence was observed that suggested interaction at the C-3 carbonyl, OH11 and OH21. As H11 was a complex multiplet it was not possible to map the change in shift but the changes in chemical shifts for H4 and H21_{α,β} were readily observed for titration experiments and the plots presented in Figure 3.5 were obtained.



Figure 3.5: Plots obtained for chemical shift changes observed during NMR titrations of HCB and AcOD in CDCl₃ for (a) H4 and (b) H21.

The interaction observed at the C-3 carbonyl, Figure 3.5 (a), appeared to become saturated at a functional monomer:template ratio of 1:20 and the dissociation constant, k_{diss} , obtained was considerably lower than any for BDN, at 0.058 M. This suggested that the degree of interaction was stronger at this point than that of BDN and AcOD (k_{diss} 0.774 M) and this may have been due to the shape of the A ring of the steroid. The interaction observed at the OH21, however, appeared to be more analogous to that of BDN, as even at FM/T ratios of 100:1, saturation was not achieved and the k_{diss} value

was 0.767 M. Interactions observed at similar points on HCB as BDN indicated that HCB would be a suitable target analogue based on NMR investigations.

The combination of the bulky hydrophobic substituent on the 17α position, the hydroxyl group in the 21 position, the similar hydrophobicity of **II**, hydrocortisone-17-butyrate, to BDN and the analogous interactions observed by NMR studies, suggested its use as a target analogue. Further, it was postulated that these factors could possibly overcome any problems posed by the difference in structure of ring-A of the two steroids. As such, HCB was chosen as the template for use as a target analogue for the synthesis of the polymers discussed in the following sections.

3.4.2 Preparation and analysis of HCB imprinted polymers

In order to investigate the variety of factors contributing to selective imprinting, the polymers outlined in Table 3.2 were prepared. These polymers were prepared to study the effect of a range of parameters including the type of crosslinking and functional monomers, the porogen and the polymer morphology.

Initial results are summarised in Table 3.3. This table also summarises the imprinting factors (*IF*) obtained for the polymers prepared during this study in both 100% acetonitrile and acetonitrile-water 30/70 (v/v). These solvents were chosen as the mobile phase for chromatographic investigation into the imprinting effect for a number of reasons; 100% acetonitrile was chosen because this solvent was used as either the porogen or co-porogen for the majority of cases and is widely used in analysis. The aqueous solution of acetonitrile allowed for a study of the behaviour of the polymers in predominantly aqueous systems but had a sufficient quantity of acetonitrile present to prevent the hydrophobic analytes from being retained to an impractical extent on the majority of the polymeric columns.

Polymer	Functional	Crosslinking	Solvent	Polymer ¹	\mathbf{IF}^2	\mathbf{IF}^2
	monomer	monomer		morphology	MeCN	MeCN/
					100%	H_2O
						(30/70)
MB3	MAA	EGDMA	MeCN	GP	7.5	2.0
MB4	MAA	DVB	MeCN/	GP	3.0	2.2
			TOL			
			3:1 (v/v)			
MB5	MAA	TRIM	MeCN	GP	2.9	1.7
MB6	MAA	PETRA	MeCN	GP	N.E.	NE
MB8		EGDMA	MeCN	GP	1.7	1.3
MB10	HEMA	EGDMA	MeCN	GP	N.E.	N.E.
MB12	MAA	EGDMA	CHCl ₃	GP	1.9	1.5*
MB13	MAA	TRIM	CHCl ₃	GP	2.8	2.0
M13	MAA	EGDMA	MeCN	MS	N.E.	1.5
M14	MAA	DVB	MeCN/	MS	1.5	N.E.**
			TOL			
			3:1 (v/v)			
M23	MAA	TRIM	DCM	MP	1.4/1.7	N.E.**
					2	

Table 3.3: Result summary for HCB imprinted polymers

¹GP: ground bulk particles; MS: microspheres; MP: microparticles; ² IF: imprinting factor as calculated as per Equation 3.4

N.E.: no imprinting effect observed

* result obtained from single sample run; ** analytes retained for more than three hours on both MIP and NIP columns

Flow rates: 0.5 mL/min for ground particles; 0.2 mL/min for microspherical beads/particles except for M23 where first value was at 0.5 mL/min and second value at 0.2 mL/min.

Table 3.3 shows a wide range of behaviour in the two solvents for all polymers and to obtain a clearer understanding of the factors driving these behaviours, the affect of the different variables were examined. Polymer composition, solvent and polymerisation strategy all play important roles in determining the success of molecular imprinting and while detailed results are presented in Table 3.4, the results will be discussed in such a manner as to isolate the discussion of each variable as much as possible.

Polymer	Surface area	Pore ¹	Pore ¹	\mathbf{m}^2	\mathbf{K}^{3}	\mathbf{N}^4	k' ⁵	IF	k'	IF
	$(m^2.g^{-1})$	volume	diameter		(L.mol ⁻¹)	(µmol.g ⁻¹)	MeCN	MeCN	MeCN-H ₂ O	MeCN-H ₂ O
		(cm ³ .g ⁻¹)	(nm)							
MB3	265	0.529	11	0.8769	9.37×10^{4}	34.8	1.42	7.5	7.22	2.0
NB3	251	0.457	10	0.9684	7.49×10^{4}	6.31	0.19		3.65	
MB4	466	0.425	6.2	0.7618	1.34×10^{5}	5.06	0.75	3.0	14.87	2.2
NB4	468	0.418	6.3	0.7326	1.47×10^{5}	1.79	0.25		6.68	
MB5	331	0.604	9.8	0.5891	2.30×10^{5}	1.1×10^{-2}	2.11	2.9	9.34	1.7
NB5	320	0.606	8.6	0.8529	8.70×10^{4}	1.2×10^{-2}	0.73		5.43	
MB12	277	0.480	9.6	n.d.	n.d.	n.d.	0.70	1.9	n.d.	n.d.
NB12	240	0.336	7.5	n.d.	n.d.	n.d.	0.36		n.d.	
MB13	246	0.242	5.1	0.8822	9.43×10^{4}	1.8×10^{-2}	2.47	2.8	25.07	2.0
NB13	211	0.280	6.8	0.6842	1.72×10^{5}	4.8×10^{-3}	0.87		12.67	

Table 3.4: Comparative result data for ground monolith particles prepared using EGDMA (MB3 and MB12), DVB (MB4) and TRIM (MB5 and MB13) as crosslinking monomers from nitrogen physisorption, chromatographic binding site characterisation and chromatographic evaluation of imprinting factors.

¹ values obtained from the adsorption curve; ² heterogeneity index obtained from Freundlich isotherm (FI); ³ apparent weighted average affinity constant of binding sites from FI; ⁴apparent number of sites from FI; ⁵capacity factor of the analytes from Equation 3.3. n.d.: no repeatable data

Frontal chromatography carried out in MeCN and all chromatography carried out at flow rate of 0.5 mL.min⁻¹

3.4.3 Polymers prepared by bulk methods

3.4.3.1 The effect of the crosslinking monomer on ground bulk particles

Four crosslinking monomers, divinyl benzene (DVB), ethylene glycol dimethacrylate (EGDMA), trimethylolpropane trimethacrylate (TRIM) and pentaerythritol triacrylate (PETRA), were chosen as the crosslinking monomers for use in this study, the structures of which are given in Figure 3.6.



Figure 3.6: Crosslinking monomers used in this study: I DVB; II EGDMA; III TRIM; IV PETRA.

With the exception of PETRA, MIPs prepared using the above crosslinking monomers exhibited promising imprinting effects. It was postulated that the use of the hydrophilic crosslinker PETRA [29] would have given the polymer a degree of hydrophilicity that repelled the predominantly hydrophobic steroidal compounds and as such PETRA polymers were not investigated further. Using the remaining three crosslinking monomers, polymers were prepared using both precipitation and bulk methods.

3.4.3.2 Zonal Chromatographic evaluation of polymer binding properties

Zonal chromatography was employed to investigate the difference in retention behaviour of the analytes on the imprinted and non-imprinted polymers. Chromatograms typical of the type observed can be seen in Figure 3.7.



Figure 3.7: Representative chromatograms of the template, HCB, when injected on the MIP column (purple trace) and the NIP column (black trace). Example shown: MB3 in MeCN.

The chromatograms presented in Figure 3.7, exhibited the broad peaks and tailing that are typical of those observed in MIP chromatography [25, 49]. This broadness and tailing can be due to the factors mentioned in Section 3.1.2, including binding equilibria and binding site heterogeneity [15, 23]. Since polymers MB3 and MB5 were prepared in acetonitrile and MB4 in acetonitrile-toluene, the differences due to crosslinking monomer alone are most readily discussed by an initial comparison of these three polymers and the retention behaviour in the two solvents are presented in Figure 3.8 (a) and (b).



Figure 3.8: Effect of crosslinker on: (a) imprinting factors and (b) capacity factors for the three polymers, MB3 (EGDMA), MB4 (DVB) and MB5 (TRIM), prepared with acetonitrile/acetonitrile mixtures as porogen in two mobile phase compositions.

As can be seen from Figure 3.8 (a), in the case of these polymers the imprinting effect was most pronounced when MeCN was used as a the mobile phase. This was not unexpected, as enhanced specific recognition in the porogen due to the so-called 'solvent memory effect' has been well documented, and Spivak *et al.* have suggested that this is due to the solvent swelling the polymer to a state where the arrangement and proximity of the functional groups of the binding sites achieve the dimensions created during the imprinting process [19, 20, 50]. The much larger k' values obtained in the aqueous based mobile phase for all polymers (Figure 3.5 (b)) suggest that the hydrophobic nature of the steroids facilitated enhanced hydrophobic interactions between the polymer and the analyte in all cases. Given the monomers used, this would be expected to be largely non-specific and indeed, the larger k' values for the MIP were accompanied by larger k' values for the NIP in all cases, resulting in a corresponding decrease in *IF* values. Despite the decrease in *IF* values specific recognition, which was presumably due mainly to hydrophobic interaction, was observed in aqueous based systems for the three polymers.

The difference between the two solvent compositions was most noticeable for MB3, which had the highest *IF* in acetonitrile but had an *IF* almost four times lower in the aqueous mobile phase. As expected, for MB4, the polymer based on DVB, the difference between *IF* for the different mobile phases was less pronounced, with a decrease of only 26% observed in the aqueous system. The increase in the k' value was also the largest in this case and was believed to be due to the hydrophobic nature of the vinyl benzene crosslinking monomer. MB5, the TRIM based polymer appeared to give intermediate results in all cases. While the k' values in acetonitrile were the largest for both MIP and NIP, the corresponding *IF* value was the lowest by a small margin. It is here postulated that this was due to an inherent affinity of the polymer for the template HCB, resulting in a non-imprinted polymer with a good deal of attraction for the template and a correspondingly low imprinting factor.

Spivak discusses the need to compare binding results with results determined using the same technique, as chromatographic and batch methods do not "correspond quantitatively" with one another [51]. There is, however, a limited number of publications dealing with corticosteroid imprinting and an even more limited number using chromatographic determination of *IF* values. In the aforementioned work by

Baggiani *et al.* where hydrocortisone was imprinted [32], chromatographic evaluation demonstrated an imprinting factor of 9.71 for the template. The values obtained in this study compared favourably with those results, with the highest value for the template reaching 7.5.

3.4.3.3 Evaluation of binding site characteristics by frontal chromatographic evaluation To elucidate fully the behaviour of polymers MB3, MB4 and MB5 in zonal chromatographic experiments, frontal chromatography was carried out as outlined in Section 3.4. As discussed in Section 1.6.3, quantitation of the number (N) and affinity (K) of binding sites is typically achieved by acquiring analyte adsorption data for a polymer over a specified concentration range, where concentrations of bound (B) and free (F) analyte are measured. This data is then treated using a suitable mathematical model, typically Scatchard analysis, Langmuir-Freundlich (LFI) or Freundlich (FI) isotherms. Rampey *et al.* state that while the LFI is more universally applicable, it is not usually necessary and the FI, which is more easily applied, can be used [52]. This is because sub-saturation concentrations are typically used and the FI is used for subsaturation conditions. The FI was chosen for this study and Equation 3.6 was employed to fit the adsorption data obtained from multistep staircase analyses to the FI as it has been shown to be suitable for use for MIPs due to the heterogeneous nature of the binding sites within such polymers [53].

The experimental data were fit to the power function of the FI using GraphPad Prism software and curves with a degree of correlation (R^2) >0.99 were obtained for all three MIPs and NIPs, demonstrating the suitability of this model for the experimental data acquired. A representative example of the experimental and 'fit' adsorption isotherms is presented in Figure 3.9.



Figure 3.9: Representative adsorption curves obtained for fitting the experimental data (dots) with FI function fit data (solid lines) for MIP and NIP. The example shown is for MB5 and NB5 (TRIM polymers).

Fitting of this data in the manner demonstrated in Figure 3.9 provides values for a and m (Equation 3.5), with the m value being of particular interest, as this is the heterogeneity index and indicates the degree of heterogeneity present within the binding sites of the polymer, while a is related to the median binding affinity [53]. It can be seen in Table 3.3, that for MB3 and MB5, the m values for the corresponding non-imprinted polymers, NB3 and NB5, respectively, are larger. This suggests a greater degree of heterogeneity for the imprinted polymers. The opposite appeared to be the case for the MB4 polymers with the non-imprinted polymer having a lower m value. Of the polymers prepared from the three crosslinking monomers, the difference in m value for MIP and NIP was greatest for polymer MB5, suggesting that the presence of the template in the pre-polymerisation complex affected the heterogeneity of the sites to the greatest extent. MB3 had the highest heterogeneity index for all three imprinted polymers, suggesting that the template had a positive directing step for the binding sites in this polymer, as the absence of the template would lead to a more homogenous polymer.

When the experimental data and fit parameters are used in conjunction with the Freundlich isotherm affinity distribution (FIAD), information can also be provided on the number and strength of the binding sites within the polymer matrix [46, 47, 54]. The FIAD results in the calculation of the apparent number of sites, N, and the apparent

weighted average affinity constant, K and the affinity distribution can be represented in a linear log *N*-log *K* plot. Figure 3.10 illustrates the linear nature of these plots for polymers MB3 (EGDMA), MB4 (DVB) and MB5 (TRIM), which demonstrates the suitability of the use of the FI for these polymeric systems [47].



Figure 3.10: Affinity distributions for MB3-MB5 using log N vs. log K format

Of the three polymers, MB3 had the highest *N* value at 34.8 µmol.g⁻¹, while its corresponding NIP had only 6.31 µmol.g⁻¹. Since there was only a marginal difference observed for the affinity constants at 9.37×10^4 and 7.49×10^4 L.mol⁻¹ for MIP and NIP, respectively, it would appear that it was the difference in the number of the binding sites that resulted in the high imprinting factor obtained for MB3. MB4 and NB4 had average affinity constants an order of magnitude higher than MB3 at 1.34×10^5 and 1.47×10^5 L.mol⁻¹, respectively, but the number of binding sites were much lower for both MIP and NIP in these cases. It must also be taken into consideration that the pore diameter for DVB polymers was also much smaller than that of the EGDMA polymers, with values of 6 and 11 nm, respectively and this may have had an effect on access to sites. What was interesting here was that *K* was slightly higher for NB4 than MB4 but due to the lower number of binding sites, $5.06 \,\mu\text{mol.g}^{-1}$ for the MIP and $1.79 \,\mu\text{mol.g}^{-1}$ for the NIP, the template was still retained more strongly on MB4. MB5 had a marginally lower number of binding sites, $1.1 \times 10^{-2} \,\mu\text{mol.g}^{-1}$, compared to NB5, $1.2 \times 10^{-2} \,\mu\text{mol.g}^{-1}$, but due its far higher *K* value, $2.3 \times 10^5 \,\text{L.mol}^{-1}$ as opposed to 8.7×10^4

L.mol⁻¹ for NB5, the template was again retained more specifically on the MIP. These results, while varying greatly from each other, nevertheless provide physical data explaining the different retention behaviour for MIP vs. NIP in each. Interestingly, the polymer which had the greatest difference in the number of binding sites between MIP and NIP, MB3, rather than that which had the largest difference between the average affinity constants, MB5, resulted in the polymer with the highest imprinting factor.

3.4.3.4 Nitrogen sorption porosimetry

To probe whether the retention behaviour of the imprinted polymers compared to that of the non-imprinted polymers was due to molecular interaction rather than mere physical and morphological factors, nitrogen sorption porosimetry was carried out, the results of which are presented in Table 3.3. The first point to note was that there were no relevant differences for the surface area of any of these MIP and NIP pairs, i.e. 265 and 251 m².g⁻¹ for MB3 and NB3, respectively, 331 and 320 m².g⁻¹ for MB5 and NB5, and 466 and 468 m².g⁻¹ for MB4 and NB4, respectively. While MB3 polymers had the lowest surface area, they had the largest pore diameter and conversely, MB4 polymers had the highest surface area but the lowest pore diameter. As with the binding characteristics, MB5 had intermediate surface area and pore diameters compared to the other two. In all cases, the nitrogen physisorption isotherms obtained were Type IV isotherms and had hysteresis resembling H4 type loops. This isotherm is typically related to meso-macroporous materials [38]. The isotherms for MB3, MB4 and MB5 are shown Figure 3.11 and the corresponding non-imprinted polymers displayed analogous isotherms.



Figure 3.11: Representative physisorption isotherms displaying isotherms for MB3-MB5.

Hysteresis, which Sing states is associated with capillary condensation within mesopores [36] and is due to differences between the adsorption and desorption behaviours, is present even at low pressure ranges. This behaviour was also observed by Urraca *et al.* who suggested that this was related to the presence of narrow pores that were more accessible at higher gas pressures but from which the adsorbate was not completely removed [38] and was also reported by Sellergren and Shea to be related to entrapment of nitrogen within the polymer matrix [18]. The pore distributions obtained from adsorption and desorption segments of the physisorption isotherms using the BJH method are presented in Figure 3.12 (a) and (b).



Figure 3.12: Pore size distributions attained using the BJH method for (a) adsorption and (b) desorption.

Figure 3.12 (a) demonstrates that the pore size distribution was largely heterogeneous and in the mesoporous 2-50 nm range. The desorption profile in Figure 3.12 (b) suggests that there is evidence of a sharp peak in the low nm range in the desorption

curve for MB3 and MB5 with a broad peak in higher ranges but Sing states that the desorption profile can be unreliable if pore blocking effects occur [36]. The significance of these results in this case is predominantly that the crosslinking monomer does indeed affect the morphological characteristics of the resulting polymers but for individual MIP/NIP pairs, the presence of the template appeared to have very little effect on the morphology of these bulk particles. This would suggest that the differences in retention behaviour were indeed due to the imprinting process and both frontal and zonal chromatographic experiments corroborate this.

3.4.3.5 The effect of the functional monomer

To better probe the role of the functional monomer in the recognition process it was deemed pertinent to prepare a polymer using crosslinking monomer alone. As Table 3.3 demonstrates, the highest imprinting effect was observed for MB3, where EGDMA was used as crosslinker and this polymer was used as the basis for functional monomer studies.

Table 3.1 shows that polymer MB8 was synthesised using only the template, HCB, and the crosslinker, EGDMA, in a 1:30 ratio in acetonitrile and Table 3.3 shows that *IF* values of 1.7 and 1.3 were obtained in MeCN and MeCN-H₂O (30/70), respectively. This suggested that there was some degree of selectivity imparted upon the polymer but it must be noted that the k' values for MeCN were 0.19 and 0.11 for MIP and NIP, respectively, so retention on the polymers in MeCN was limited. When MAA was used as the functional monomer (MB3 polymers), the k' value increased to 1.14 for the MIP but only 0.19 for the NIP. This large increase in the MIP relative to that of the NIP suggested that it was not just an inherent increased affinity for the template due to the presence of MAA but rather the creation of imprinted sites, which caused an increase in the degree of interaction between the template and the polymer.

MB10 was prepared as a polymeric analogue to MB3, replacing MAA with HEMA as the functional monomer as NMR studies had suggested a degree of interaction of HEMA with steroidal templates (Section 2.4.). Despite the possible interaction demonstrated by NMR studies, this polymer displayed no imprinting effect and it was postulated that, as with PETRA, the slightly hydrophilic character of the monomer prevented any selective interaction between the hydrophobic steroid analytes and the polymer. Another factor could be interaction between the solvent and the sites, effectively forming a barrier between the steroid and the sites. For this reason, MAA was chosen as the functional monomer for the preparation of all further polymers.

3.4.3.6 The effect of the porogen on ground bulk polymers

The results shown in Table 3.3 and discussed in the preceding sections suggested that the most promising bulk polymers were those prepared with EGDMA and TRIM as the crosslinking monomers. Since MB3 and MB5 were prepared in acetonitrile, it was decided to prepare similar polymers but using chloroform as the porogen as a route to understanding the role of the porogen. The polymers so prepared were MB12 and MB13 for EGDMA and TRIM, respectively. Table 3.3 demonstrates that while an imprinting factor of 1.9 was achieved for MB12 in acetonitrile, this was over three times lower than the corresponding *IF* value obtained for MB3 in acetonitrile. It was also not possible to obtain a repeatable retention time for any of the injections carried out in the MeCN-H₂O mobile phase and for this reason no further chromatographic studies were carried out. It was interesting to note, however, that nitrogen sorption analysis suggested that there was very little difference between the surface areas, pore volumes or pore diameters of the polymers prepared in the different porogens and isotherms obtained were of the same type, despite the fact that polymers prepared in chloroform are typically non-porous [50].

This similarity in morphology was not repeated for the TRIM polymers prepared in chloroform. In this case, where MB5 (prepared in MeCN) polymers produced Type IV isotherms with H4 hysteresis loops, MB13 (prepared in CHCl₃) produced again Type IV isotherms but appeared to produce a H2 hysteresis loop for the MIP and a H4 hysterisis loop for the NIP, as demonstrated in Figure 3.13 (a).


Figure 3.13: Comparison of physisorption isotherms obtained for (a) MB13 and NB13 and (b) MB5 and MB13.

While H2 hysteresis loops are considered difficult to accurately interpret [36], the differences exhibited for the templated polymer prepared in CHCl₃ suggest that the template plays a major role in the type of polymer morphology obtained when TRIM and MAA are used as the monomers with chloroform as a solvent. The increased morphological affect of the template was postulated because H4 hysteresis loops were obtained in all other cases for bulk polymers, for both imprinted and non-imprinted polymers. Table 3.3 shows that the pore volumes and pore diameters were also reduced in the polymers prepared in chloroform using TRIM as the crosslinking monomer.

Frontal analysis on these MB13 polymers also uncovered different binding properties than the acetonitrile prepared polymers. The average affinity constant for NB13 was actually higher, at 1.72×10^5 L.mol⁻¹, than the value of 9.43×10^4 L.mol⁻¹ for MB13, while the opposite was observed for MB5. The number of binding sites, *N*, for MB13 was higher by an order of magnitude than NB13, with values of 1.8×10^{-2} and 4.8×10^{-3}

 μ mol.g⁻¹ for MIP and NIP, respectively. The latter value was the lowest number of binding sites for all of the TRIM based polymers. The heterogeneity index was also higher for MB13 than NB13 suggesting that the presence of the template in the prepolymerisation mixture had an effect on the creation of the binding sites when chloroform was used as the porogen.

Despite the differences in polymer morphology and binding characteristics observed for frontal analysis, the results obtained for zonal chromatography were remarkably similar with an imprinting factor of 2.8 obtained in acetonitrile for MB13, compared to the value of 2.9 for MB5. Using MeCN-H₂O (30/70) as the mobile phase, the *IF* value was actually higher for MB13 than MB5, with values of 2 and 1.7, respectively. These results suggested that while the porogen had pronounced effects on the morphology and individual binding characteristics of the imprinted polymer, the overall effect on the selective binding of the analyte was negligible and the solvent memory effect was negligible. This was contrasted by the EGDMA based polymer, MB12, where the solvent memory effect deleteriously affected binding selectivity. These results suggest that the use of the tri-functional crosslinker TRIM, as opposed to the di-functional EGDMA, while reported to increase load capacity and improve resolution of imprinted polymers [55], may also contribute to a more rigidly crosslinked polymeric matrix in which the binding sites created are less susceptible to swelling effects by the various solvents used. This factor could prove to be advantageous in that polymers could be prepared in non-polar solvents, thereby promoting hydrogen bonding interactions but could be utilised in aqueous based or polar solvents where the rigid binding sites would remain intact for rebinding.

3.4.3.7 Selectivity studies

To study the selectivity of the polymers and explore the factors driving recognition in the polymers, zonal chromatography was carried out on the most successful bulk polymers using all six steroidal analogues (Figure 3.2). The results for MeCN and MeCN-H₂O (30/70) are given in Table 3.5 (a) and (b) and all data is the average of at least two repetitions.

Table 3.5: Summary for the retention behaviour of ground bulk particles, MB3 (EGDMA), MB4 (DVB), MB5 (TRIM with MeCN as porogen) and MB13 (TRIM with CHCl₃ as porogen), using all six steroidal analogues in two mobile phases: (a) acetonitrile and (b) acetonitrile-water (30/70). (k' values quoted are MIP only).

	(a)							
Analyte	M	B3	MB4		MB5		MB13	
	k'	IF	<i>k</i> '	IF	<i>k</i> '	IF	<i>k</i> '	IF
НСВ	1.42	7.5	0.75	3.0	2.11	2.9	2.47	2.8
BDN	1.57	4.9	0.88	2.2	2.13	1.8	2.92	1.9
HYD	2.73	3.9	0.99	1.7	4.90	1.6	4.95	1.8
PRD	3.15	9.3	1.10	1.5	6.16	1.7	6.00	1.7
HAC	0.68	3.1	0.28	2.9	0.56	1.6	1.38	1.3
PAC	0.91	2.7	0.38	2.5	0.77	1.4	1.03	1.3

(b)

Analyte	MB3		MB4		MB5		MB13	
	<i>k</i> '	IF						
НСВ	7.22	2.0	14.87	2.2	9.34	1.7	25.07	2.0
BDN	6.66	1.7	14.37	2.0	9.29	1.3	28.86	1.7
HYD	1.58	1.7	1.16	2.0	1.59	1.3	5.19	1.8
PRD	3.84	1.7	1.06	2.0	1.48	1.2	5.03	1.8
HAC	6.33	1.6	8.08	1.9	6.69	1.2	22.75	1.8
PAC	6.09	1.6	7.13	1.8	7.00	1.2	23.51	1.8

These results for the imprinting factors in the two solvent compositions are also presented graphically in Figure 3.14.



Figure 3.14: Imprinting factors obtained for ground bulk particles, MB3, MB4, MB5 and MB13, in MeCN and MeCN-H₂O (30/70).

HYD

PAC

HAC

6.0 *IF*

4.0

2.0

0.0

HCB

BDN

PRD

While analysis of the data presented in Table 3.5 and Figure 3.14 demonstrate that there are a number of differences between the imprinting effects observed in MeCN and in MeCN-H₂O, it can also be seen that there are a number of discernible patterns. Firstly, with the exception of one anomalous result, the template, HCB, had the highest *IF* value for all polymers in both solvents. This would be expected to be the case as the binding sites created within the imprinted polymers should most closely match the shape, size and functionality of the template. The exception to this was in the case of prednisolone in MB3 (EGDMA) using acetonitrile as the solvent.

MB4 MB13

MB3

MB5

In 100% MeCN, MB3 appears to display shape and size selectivity as, apart from the result for PRD, the highest *IF*, 7.5, was obtained for the template, HCB, followed by BDN at 4.9. The structure of BDN was used to choose the template and as such, the cavity formed by the template should match the shape and size of BDN to a greater extent than the other analytes. PRD and HYD possess no acetate, acetal or butyrate groupings on the C-17 or C-21 positions and, in theory, should 'fit' into the templated cavity with relative ease. The high *IF* obtained for PRD shows this to be the case while HYD had an *IF* of 3.9, next to BDN. HAC and PAC, both possessing acetate groupings on the C-21 position had the lowest *IF* values at 3.1 and 2.7, respectively. This suggested that the shape of these molecules obstructed rebinding to the same extent as the more sterically suitable steroids. A somewhat similar pattern was observed for MB13, although the *IF* values were lower than those of MB3 in all cases. That being said, shape and size selectivity still appeared to play an important role in analyte recognition.

The TRIM polymer prepared in acetonitrile, MB5, had an *IF* of 2.9 for the template and again, BDN had the closest imprinting factor but in this case the distinction between the remaining analytes was less clear. Here, the difference between the *IF* values for HCB and BDN was 1.1 but the difference between BDN and the lowest, PAC, was only 0.4. This suggested that MB5 had a high degree of fidelity for the template but rebound the remaining analytes to a more or less similar extent. It would be expected that TRIM, due to its trifunctionality, should give a more rigid structure with better shape and functional group selectivity for the template.

MB4, the DVB polymer, while having relatively low k' values for the majority of analytes still managed to achieve the highest imprinting factor for HCB. For this polymer, hydrophobicity of the analytes appeared to play a more significant role as HYD and PRD, the analytes with the lowest predicted log P values, had the lowest *IF* values. This suggested that while these analytes could fit in the binding sites quite readily, they were repelled by their extra hydroxyl groups from the relatively hydrophobic polymer backbone, comprised mainly of DVB.

For analyses carried out using the predominantly aqueous mobile phase, i.e. MeCN-H₂O (30/70), Table 3.5 (b) and Figure 3.14 (b) show that imprinting factors in all cases were

reduced and in the case of MB3, reduced significantly. The results also show that this reduction in imprinting factors coincided with an increase in the k' values in the majority of cases. PRD and HYD were exceptional in this trend and this may have been due to their extra hydroxyl group at the C-17 position, imparting greater affinity towards the aqueous mobile phase.

Despite the reduction in *IF* values it was seen that a number of patterns still emerged, such as the highest imprinting factor in all cases for the template. MB5 still displayed a degree of fidelity for the template, with an *IF* of 1.7, while showing little or no discrimination between the other analytes with *IF* values ranging from 1.2-1.3. MB13, MB4 and MB3 did display marginally higher degrees of specificity for the template with *IF* values of 2.2, 2 and 2, respectively, but the difference between template values and other analyte values was no greater than 0.4 in any case. These results suggest that the hydrophobic interactions taking place in the predominantly aqueous solutions cause the analytes to be retained to a greater extent on both MIP and NIP in all cases. This still allowed for the analytes to interact with the specific binding sites on the MIP to a greater extent, thus demonstrating an imprinting effect in all cases.

Overall, these results show a high degree of recognition in these bulk polymers prepared with different crosslinking monomers. The highest degree of shape and size selectivity was demonstrated by MB3, the polymer prepared with EGDMA, in acetonitrile. This was not unexpected as the combination of the "reactive and prochiral methacrylate ester with a short spacer" allows for a large degree of conformational possibilities to facilitate imprinting and a highly rigid polymer matrix [56]. While the affinity of the polymers for the analytes was increased by the use of the tri-functional TRIM, where both MIP and NIP had a high k' values, the shape and size differentiation was somewhat diminished in the case of MB5, although fidelity for the template was relatively high. What the combination of these results suggest is that the 'best' polymer would not be chosen based on a comparison of the individual characteristics but rather on the ultimate application for the material. For example sensing applications, where sensing of closely related structural analogues was necessary, MB3 with the obvious shape and size selectivity may best be utilised. Whereas for an application that required the slow release of an analyte, such as controlled release applications, the material with the highest affinity for the drug, such as MB5 or MB13 may best be employed.

3.4.4 Polymers prepared by precipitation

The preparation of bulk polymers prepared with HCB as the template suggested the suitability of this template as a target analogue for corticosteroid recognition and as such, was used as the template for the synthesis of imprinted polymers prepared by precipitation methods. The information gleaned from the work discussed in Chapter 2 allowed for the rapid synthesis of polymeric microspheres and microparticles with a high degree of control over particle size. The polymers that were synthesised in this way (Table 3.1) all used MAA as the functional monomer and employed EGDMA (M13), DVB (M14) and TRIM (M23) as crosslinking monomers.

M13 and M14 were prepared in acetonitrile and acetonitrile-toluene 3:1 (v/v), respectively, using thermal initiation while M23 was prepared in dichloromethane using UV irradiation to initiate polymerisation. While M13 and M14 were prepared based on information acquired in Chapter 2 it was deemed suitable to use a non-polar solvent to synthesise M23 based on the results given in Section 3.4.3. These results demonstrated the similar behaviour of TRIM polymers prepared in polar and non-polar solvents, with slightly improved imprinting effects observed for those prepared in non-polar porogens. The use of dichloromethane, however, necessitated the use of a low temperature polymerisation strategy and a method was developed based on that employed by Ye *et al.* for the preparation of TRIM based microspheres with estradiol selectivity. SEM micrographs of the precipitated polymers investigated in this study can be seen in Figure 3.15



Figure 3.15: Scanning electron micrographs of: (a) M13; (b) N13; (c) M14; (d) N14; (e) M23; (f) N23.

The SEM images in Figure 3.15 (a)-(f) show that polymers with a microspherical morphology were obtained for M13 and M14 formulations but the M23 formulation, while producing precipitated microparticles, did not possess the smooth spherical shape of the other formulations. Instead, a mixture comprising of a small percentage of microspheres but predominantly rough agglomerated particles in the low micron range were obtained. The particle sizes for the three polymer formulations are presented in Table 3.6.

Table 3.6: Particle size data for polymer formulations M13, M14 and M23

Polymer	Μ	13	M14		M23	
formulation	MIP	NIP	MIP	NIP	MIP	NIP
Particle size (µm)	1.1-3.2	1.0-2.8	1.5-3.7	1.7-7.3	2.3-8.1	2.4-12.4

Figure 3.15 and Table 3.6 demonstrate that a high degree of control over particle size was attained. For M13, the EGDMA based polymer, the template appeared to have little effect on the particle size. The imprinted beads were slightly larger than the non-imprinted beads, which was the reverse effect of the use of budesonide as the template. The bead size was similar to those prepared using the same formulation, in Chapter 2. M14, the DVB polymer had a similar size distribution as the budesonide imprinted

polymer and the same effect of the template, i.e. smaller particles for the MIP was observed. Overall, the effect of changing templates for EGDMA and DVB based beads was relatively minor.

M23, the TRIM based polymer, had not been prepared for budesonide. Interestingly, as stated above, the smooth spherical shape was absent for these polymers. This has been observed in the work of Yoshimatsu *et al.* where varying amounts of TRIM were added to DVB formulations and in some cases the particles lost their spherical shape [57]. Precipitated particles prepared with TRIM by Yong Jiang *et al.* also lacked this spherical shape [58]. While spherical TRIM particles have also been prepared [59], it appears to be the case that for certain formulations, this crosslinker forms irregular shaped particles rather than spheres. This may be related to the trifunctionality of the monomer and the rapid nature of the growing network, which is ultimately prevented from forming a space filling gel due to dilution.

The primary focus of this study, however, was to investigate the applicability of the target analogue approach for imprinting polymeric microspheres and to determine if MIP chromatography might be a suitable probe for any selective recognition created within the polymers during the molecular imprinting process. To this end, the results obtained for these polymers are presented in Table 3.7.

Table 3.7: Comparative result data for polymers prepared by a precipitation polymerisation approach using three different crosslinking monomers: EGDMA (M13), DVB (M14) and TRIM (M23) from nitrogen physisorption, chromatographic binding site characterisation and chromatographic evaluation of imprinting factors.

Polymer	Surface	Pore ¹	Pore ¹	\mathbf{m}^2	\mathbf{K}^2	\mathbf{N}^3	k'	IF	k'	IF
	area	volume	diameter		(L.mol ⁻¹)	(µmol.g ⁻¹)	MeCN	MeCN	MeCN-H ₂ O	MeCN-H ₂ O
	$(m^2.g^{-1})$	(cm ³ .g ⁻¹)	(nm)							
M13	5.54	0.012	14	n.d.	n.d.	n.d.	n.r.	n.d.	5.84	1.5
N13	4.28	0.015	17	n.d.	n.d.	n.d.	n.r.		3.94	
M14	596	0.142	2.9	0.8556	1.02×10^{5}	18.1	1.85	15	n.d.	n.d.
N14	631	0.154	2.8	0.8267	1.11×10^{5}	16.4	1.26	1.5	n.d.	
M23	5.47	0.011	8.9	0.9554	7.74×10^{4}	7.3×10^{-2}	4.13	17	n.d.	n d
N23	21.24	0.025	7.7	0.9627	7.60×10^{4}	3.1×10^{-2}	2.4	1.7	n.d.	n.u.

¹ values obtained from the adsorption curve; ² heterogeneity index obtained from Freundlich isotherm (FI); ³ apparent weighted average affinity constant of binding sites from FI; ⁴apparent number of sites from FI;

n.d.: no repeatable data; n.r.: not retained

k' and IF values obtained at 0.2 mL.min⁻¹ and frontal chromatography carried out at 0.5 mL.min⁻¹ in MeCN

The results presented in Table 3.7 demonstrate that the effect of imprinting on the precipitated particles was not as clear-cut as in the case of the ground bulk particles. This was to be expected given the results obtained for precipitated polymers in Chapter 2 and the NMR studies showing weak interaction between the template in those studies. This fact notwithstanding, there were still a number of positive results obtained for the precipitated particles. While no polymer yielded imprinting factors in both solvent compositions, each polymer displayed imprinting effects in one solvent system.

For analysis where acetonitrile was used as the mobile phase, HCB was not retained past the void marker on M13 or N13. Some clue to the reasons for this can be seen in the low surface areas observed for these polymers, 5.5 m².g⁻¹ and 4.28 m².g⁻¹, for MIP and NIP, respectively. The isotherms (Figure 3.16) demonstrate Type II behaviour.



Figure 3.16: Physisorption data obtained for M13 and N13 displaying Type II isotherms.

The Type II physisorption isotherms obtained for M13 and N13, coupled with the low BET surface area suggest that these polymers were non-porous adsorbents. While relatively large average pore diameters were calculated, these appear to be somewhat misleading as the pore volumes were low, with values of 0.012 cm³.g⁻¹ for M13 and 0.015 cm³.g⁻¹ for N13. The low surface area and poor porosity of these polymers would suggest that acetonitrile was a relatively 'bad' thermodynamic solvent for the growing polymer, causing it to phase separate at an early stage resulting in non-porous microspheres with low surface areas [60]. The poor porosity and low surface area of the polymers coupled with a greater affinity of the analyte for the solvent may have

prevented the analytes from being retained on the polymeric column, even in imprinted sites.

The non-retention of the analyte prevented further investigation of the binding site characteristics by frontal chromatography. It did not prevent, however, the analysis of these polymers in the acetonitrile-water mobile phase and it can be seen from Table 3.7 that an imprinting factor of 1.5 was achieved for this analysis. Again, the k' values were increased considerably due to presumably hydrophobic interactions and this facilitated interaction with specific binding sites present on the surface of these polymers.

Precipitated polymers prepared using DVB as the crosslinking monomer, M14 and N14, displayed quite different behaviour in a number of cases. Here acetonitrile-toluene in a 3:1 (v/v) ratio was used, as this had been shown in Chapter 2 to prepare the largest and most spherical particles for polymers based on this crosslinker. Coupled with this, toluene is a thermodynamically 'good' solvent for polymers prepared with DVB, as its addition delays phase separation of the growing polymer allowing for the creation of pores [60, 61]. The BET surface areas for these polymers would therefore be expected to be high and this was shown to be the case, with values of 596 m^2 .g⁻¹ and 631 m^2 .g⁻¹ for MIP and NIP, respectively. The pore volumes, 0.142 cm³.g⁻¹ for the MIP and 0.154 cm³.g⁻¹ for the NIP were also considerably larger than for M13 or M23. The physisorption isotherms for M14 and N14 are shown in Figure 3.17 and the presence of hysteresis appears to suggest Type IV behaviour similar to that seen with the ground bulk particles and typical for meso-macroporous materials. Again, the gas was not fully desorbed at low P/P⁰ values and as with the ground bulk particles, this was probably due to the presence of narrow pores from which it was difficult for the gas to be removed [38].



Figure 3.17: Physisorption isotherms obtained for M14 and N14.

As well has having different morphological characteristics, the M14 polymers also had different retention behaviours for the template. While the large surface areas and strong hydrophobic nature of the polymer retained the analytes on the columns for over three hours in the case of both MIP and NIP in highly aqueous mobile phases, an imprinting factor of 1.5 was achieved in MeCN.

Frontal chromatography showed that there was very little difference between the heterogeneity indexes for the MIP and NIP. While they both displayed a strong degree of affinity for the template, with values of 1.02×10^5 L.mol⁻¹ and 1.11×10^5 L.mol⁻¹ for MIP and NIP, respectively, the MIP had a marginally higher *N* value at 18.1 µmol.g⁻¹ compared to 16.4 µmol.g⁻¹ for the NIP.

These results suggested that while imprinting did indeed take place, the number of specific binding sites created within the polymer matrix was minimal. The $\log N$ vs. $\log K$ plot shown in Figure 3.18 demonstrated the almost negligible differences for the two polymers.



Figure 3.18: Log N vs. log K plot obtained for M14 and N14 showing the marginal differences between the MIP and NIP.

M23 exhibited similar behaviour to M14 in that an *IF* value of 1.7 was achieved for HCB in MeCN. This was the highest *IF* observed for the HCB for the precipitated polymers. This was despite that fact that the MIP had morphological characteristics similar to those of M13, with a surface area of only 5.47 m².g⁻¹ and displaying a Type II isotherm. Interestingly, N23 had a considerably higher surface area (though by no means a high surface area) with a value of 21.24 m².g⁻¹. It was noteworthy that N23 had a Type IV isotherm with H4 hysteresis, similar to all of the ground bulk particles and M14. The differences between the two isotherms is depicted in Figure 3.19.



Figure 3.19: Physisorption isotherms obtained for M23 (Type II) and N23 (Type IV).

With the data for M23 suggesting a non-porous material and that for N23 suggesting a meso-macroporous material, it would appear that the presence of the template in the pre-polymerisation complex had a profound effect on the morphological characteristics

of the resultant polymer. While the SEM images in Figure 3.15 (e) and (f) show that microparticles rather than microspheres were formed in both cases, there did appear to be some differences in the morphologies of the polymers in that the NIP appeared to have slightly larger agglomerates than did the NIP. This would suggest, as would the gas sorption data, that the addition of template to the complex mixture reduced its thermodynamic favourability for the polymer and caused the precipitation of the MIP at an earlier stage and resulting in non-porous microparticles.

While gas sorption data showed differences between the morphologies of the polymers, frontal chromatographic analysis demonstrated that differences also existed for the binding characteristics of the two polymers. The heterogeneity index, *m*, was slightly lower for M23 at 0.9554 compared to 0.9627 for N23, suggesting the presence of a marginally greater degree of heterogeneity. *K* values were 7.74×10^4 L.mol⁻¹ and 7.60×10^4 L.mol⁻¹ for MIP and NIP, respectively while the *N* value was twice as high at 7.3×10^{-2} µmol.g⁻¹ for M23 and 3.1×10^{-2} µmol.g⁻¹ for N23. These results are represented graphically in Figure 3.20.



Figure 3.20: Log *N* vs. log *K* plot for M23 showing differences in binding strength and the number of binding sites.

While these differences were relatively marginal, they were sufficient to impart a degree of specific binding to the MIP. As was the case for M14 the steroid was retained for over three hours on both MIP and NIP when the aqueous mobile phase was used, presumably due to strong hydrophobic interactions and no repeatable or reliable data could be obtained for acetonitrile-water mixtures.

3.4.4.1 Selectivity studies for precipitation polymers

The six steroidal analogues were again injected onto the imprinted and control polymers to attempt to probe the factors driving recognition within the precipitated polymers. These studies were only carried out for each polymer in the solvent in which imprinting effects could be measured, therefore all three polymers could not be directly compared. For this reason the results in Table 3.8 and Figure 3.21 are presented for analysis of M13 in acetonitrile-water and for M14 and M23 in acetonitrile. All results given are an average of at least two repetitions.

Analyte	M	M13		M14		M23	
	(MeCN	(MeCN-H ₂ O)		(MeCN)		CN)	
	<i>k</i> '	IF	k'	IF	<i>k</i> '	IF	
HCB	5.84	1.5	1.85	1.5	4.13	1.7	
BDN	8.29	1.4	2.41	1.1	7.70	1.7	
HYD	1.89	1.5	3.10	1.7	10.73	1.7	
PRD	1.84	1.5	3.61	1.6	14.47	1.7	
HAC	7.92	1.5	1.41	1.3	4.07	1.7	
PAC	7.32	1.4	1.75	1.1	5.96	1.6	

Table 3.8: Result data for selectivity studies on the six structural analogues for precipitated polymers M13, M14 and M23.



Figure 3.21: Imprinting factors observed for the six structural analogues for (a) M13 in MeCN-H₂O (30/70) and (b) M14 and M23 in MeCN.

As Table 3.8 and Figure 3.21 demonstrate, there was almost no difference present for the different analytes on polymers M13 or M23, with differences no greater than 0.1 for the imprinting factors of any of the analytes. M14 displayed some small differences in

IF values, with low values for BDN, HAC and PAC, ranging from 1.1 to 1.3. The imprinting effect was more pronounced for the template, with a value of 1.5 but the highest values were for PRD and HYD at 1.6 and 1.7, respectively. While the higher *IF* for the template would have been expected, the higher values for HYD and PRD were accompanied by increased retention and while this may have been due to slight incompatibilities with the solvent, it allowed for a greater degree of interaction between these analytes and the specific binding sites on the imprinted polymers.

While the results obtained for all of the precipitated beads were, as previously stated, less clear-cut and without the discernible patterns of the ground bulk particles, they nonetheless demonstrated the first instance of corticosteroid specific molecularly imprinted polymers prepared *via* precipitation methods, using a target analogue approach. While many of the differences were small, it must be noted that these results were highly repeatable and averages of at least two values are quoted. The lower imprinting effects observed in the precipitated beads than the bulk particles corroborated the findings of Chapter 2, which suggested that a means of promoting the interaction between the template and monomers in the pre-polymerisation solution for dilute precipitated methodologies would be necessary. It was shown here that even the use of a non-polar solvent, DCM, and UV polymerisation at 4 °C could not promote the interaction enough to significantly increase the number of binding sites within the polymer matrix.

3.5 Conclusion

Molecularly imprinted polymers with group selectivity for corticosteroids have been prepared using a ground, bulk particle format by the target analogue approach. Hydrocortisone-17-butyrate was chosen as the template that most closely matched one of the primary targets, budesonide and the success of this strategy was demonstrated by the high degree of affinity observed for budesonide over analytes other than the template in the majority of cases. This suggested the suitability of this approach for the preparation of polymers for use in solid-phase extraction applications where HCB imprinted polymers could specifically retain BDN. This is discussed in Chapter 4.

Novel corticosteroid imprinted microspheres and microparticles were prepared using three crosslinking monomers. A high degree of control over bead size and morphology was demonstrated for these beads. Binding site characterisation and chromatographic evaluation suggested that the molecular imprinting effect was less pronounced than in the case of the bulk particle strategy. This corroborated findings in Chapter 2, which suggested that this was related to weak interaction between the template and monomer in the pre-polymerisation mixture for such dilute solutions and efforts to enhance these interactions will be discussed in Chapter 5.

References

- 1. Andersson LI, Paprica A, Arvidsson T: A highly selective solid phase extraction sorbent for pre-concentration of sameridine made by molecular imprinting. *Chromatographia* 1997, **46:**57-62.
- 2. Wang S, Li Y, Wu X, Ding M, Yuan L, Wang R, Wen T, Zhang J, Chen L, Zhou X, Li F: Construction of uniformly sized pseudo template imprinted polymers coupled with HPLC-UV for the selective extraction and determination of trace estrogens in chicken tissue samples. *Journal of Hazardous Materials* 2011, **186**:1513-1519.
- 3. Zhao W, Sheng N, Zhu R, Wei F, Cai Z, Zhai M, Du S, Hu Q: **Preparation of** dummy template imprinted polymers at surface of silica microparticles for the selective extraction of trace bisphenol A from water samples. *Journal of Hazardous Materials* 2011, **179:**223-229.
- 4. Li M, Zhang L, Meng Z, Wang Z, Wu H: Molecularly-imprinted microspheres for selective extraction and determination of melamine in milk and feed using gas chromatography-mass spectrometry. *Journal of Chromatography B* 2011, **878**:2333-2338.
- 5. Chapuis-Hugon F, Cruz-Vera M, Savane R, Ali WH, Valcarcel M, Deveaux M, Pichon V: Selective sample pretreatment by molecularly imprinted polymer for the determination of LSD in biological fluids. *Journal of Separation Science* 2009, **32**:3301-3309.
- 6. Kareuhanon W, Lee VS, Nimmanpipug P, Tayapiwatana C, Pattarawarapan M: Synthesis of Molecularly Imprinted Polymers for Nevirapine by Dummy Template Imprinting Approach. *Chromatographia* 2009, **70:**1531-1537.
- 7. Theodoridis G, Kantifes A, Manesiotis P, Raikos N, Tsoukali-Papadopoulou H: Preparation of a molecularly imprinted polymer for the solid-phase extraction of scopolamine with hyoscyamine as a dummy template molecule. *Journal of Chromatography A* 2003, **987**:103-109.
- 8. Wang X-J, Xu Z-L, Feng J-L, Bing N-C, Yang Z-G: Molecularly imprinted membranes for the recognition of lovastatin acid in aqueous medium by a template analogue imprinting strategy. *Journal of Membrane Science* 2008, 313:97-105.
- Riskin M, Ben-Amram Y, Tel-Vered R, Chegel V, Almog J, Willner I: Molecularly Imprinted Au Nanoparticles Composites on Au Surfaces for the Surface Plasmon Resonance Detection of Pentaerythritol Tetranitrate, Nitroglycerin, and Ethylene Glycol Dinitrate. Analytical Chemistry 2011, 83:3082-3088.
- 10. Lanza F, Sellergren B: **The application of molecular imprinting technology to solid phase extraction.** *Chromatographia* 2001, **53:**599-611.
- 11. Khorrami AR, Taherkhani M: Synthesis and Evaluation of a Molecularly Imprinted Polymer for Pre-concentration of Patulin from Apple Juice. *Chromatographia* 2011, **73**:S151-S156.
- 12. Kubo T, Hosoya K, Watabe Y, Tanaka N, Takagi H, Sano T, Kaya K: Interval immobilization technique for recognition toward a highly hydrophilic cyanobacterium toxin. *Journal of Chromatography B* 2004, **806**:229-235.
- 13. Baggiani C, Anfossi L, Baravalle P, Giovannoli C, Giraudi G, Barolo C, Viscardi G: Determination of banned Sudan dyes in food samples by molecularly imprinted solid phase extraction-high performance liquid chromatography. Journal of Separation Science 2009, 32:3292-3300.

- 14. Baggiani C, Baravalle P, Giovannoli C, Anfossi L, Giraudi G: Molecularly imprinted polymers for corticosteroids: Analysis of binding selectivity. *Biosensors and Bioelectronics* 2010, **26**:590-595.
- 15. Wulff G, Vesper W: ENZYME-ANALOGUE BUILT POLYMERS .8. PREPARATION OF CHROMATOGRAPHIC SORBENTS WITH CHIRAL CAVITIES FOR RACEMIC-RESOLUTION. Journal Of Chromatography 1978, 167:171-186.
- 16. Wulff G, Vesper W, Grobeeinsler R, Sarhan A: ENZYME-ANALOGUE BUILT POLYMERS, .4. SYNTHESIS OF POLYMERS CONTAINING CHIRAL CAVITIES AND THEIR USE FOR RESOLUTION OF RACEMATES. Makromolekulare Chemie-Macromolecular Chemistry And Physics 1977, 178:2799-2816.
- 17. Sellergren B, Ekberg B, Mosbach K: MOLECULAR IMPRINTING OF AMINO-ACID DERIVATIVES IN MACROPOROUS POLYMERS -DEMONSTRATION OF SUBSTRATE-SELECTIVITY AND ENANTIO-SELECTIVITY BY CHROMATOGRAPHIC RESOLUTION OF RACEMIC MIXTURES OF AMINO-ACID DERIVATIVES. Journal Of Chromatography 1985, 347:1-10.
- Sellergren B, Shea KJ: Influence Of Polymer Morphology On The Ability Of Imprinted Network Polymers To Resolve Enantiomers. *Journal Of Chromatography* 1993, 635:31-49.
- 19. Spivak D, Gilmore MA, Shea KJ: Evaluation of binding and origins of specificity of 9-ethyladenine imprinted polymers. *Journal Of the American Chemical Society* 1997, 119:4388-4393.
- 20. Yu C, Mosbach K: Influence of mobile phase composition and cross-linking density on the enantiomeric recognition properties of molecularly imprinted polymers. *Journal of Chromatography A* 2000, **888**:63-72.
- 21. Jiang M, Shi Y, Zhang R-l, Shi C-h, Peng Y, Huang Z, Lu B: Selective molecularly imprinted stationary phases for Bisphenol A analysis prepared by modified precipitation polymerization. *Journal of Separation Science* 2009, **32**:3265-3273A.
- 22. Liu Y, Hoshina K, Haginaka J: Monodispersed, molecularly imprinted polymers for cinchonidine by precipitation polymerization. *Talanta* 2010, **80**:1713-1718.
- 23. Sellergren B: Imprinted chiral stationary phases in high-performance liquid chromatography. *Journal of Chromatography A* 2001, **906**:227-252.
- 24. Jiang JB, Song KS, Chen Z, Zhou Q, Tang YW, Gu FL, Zuo XJ, Xu ZG: Novel molecularly imprinted microsphere using a single chiral monomer and chirality-matching (S)-ketoprofen template. *Journal of Chromatography A* 2011, **1218**:3763-3770.
- 25. Lee WC, Hung CY, Hwang CC: Separation of cholesterol from other steroids using molecularly imprinted polymer prepared by seeded suspension polymerization. *Chemical And Biochemical Engineering Quarterly* 2008, 22:151-156.
- 26. Yang M, Chu X: Molecularly Imprinted Polymers with Mandelic Acid as Template. *Analytical Letters* 2010, **43**:2390 2398.
- 27. Shi Y, Peng D-D, Shi C-H, Zhang X, Xie Y-T, Lu B: Selective determination of trace 17[beta]-estradiol in dairy and meat samples by molecularly imprinted solid-phase extraction and HPLC. *Food Chemistry* 2011, 126:1916-1925.
- 28. Hall AJ, Lanza-Sellergren F, Manesiotis P, Sellergren Br: Non-covalent imprinting of phosphorous esters. *Analytica Chimica Acta* 2005, **538**:9-14.

- 29. Manesiotis P, Borrelli C, Aureliano CSA, Svensson C, Sellergren B: Watercompatible imprinted polymers for selective depletion of riboflavine from beverages. *Journal of Materials Chemistry* 2009, **19:**6185-6193.
- 30. Anfossi L, Baggiani C, Baravalle P, Giovannoli C, Guzzella L, Pozzoni F: Molecular Recognition of the Fungicide Carbendazim by a Molecular Imprinted Polymer Obtained through a Mimic Template Approach. Analytical Letters 2009, 42:807-820.
- 31. Baggiani C, Anfossi L, Baravalle P, Giovannoli C, Giraudi G: Molecular recognition of polycyclic aromatic hydrocarbons by pyrene-imprinted microspheres. *Analytical and Bioanalytical Chemistry* 2007, **389:**413-422.
- 32. Baggiani C, Giraudi G, Trotta F, Giovannoli C, Vanni A: Chromatographic characterization of a molecular imprinted polymer binding cortisol. *Talanta* 2000, **51**:71-75.
- 33. Kitahara K-I, Yoshihama I, Hanada T, Kokuba H, Arai S: Synthesis of monodispersed molecularly imprinted polymer particles for highperformance liquid chromatographic separation of cholesterol using templating polymerization in porous silica gel bound with cholesterol molecules on its surface. *Journal of Chromatography A* 2010, **1217**:7249-7254.
- 34. Gavrilovic I, Mitchell K, Brailsford AD, Cowan DA, Kicman AT, Ansell RJ: A molecularly imprinted receptor for separation of testosterone and epitestosterone, based on a steroidal cross-linker. *Steroids* 2011, **76**:478-483.
- 35. O'Mahony J, Molinelli A, Nolan K, Smyth MR, Mizaikoff B: Anatomy of a successful imprint: Analysing the recognition mechanisms of a molecularly imprinted polymer for quercetin. *Biosensors and Bioelectronics* 2006, 21:1383-1392.
- 36. Sing KSW: **Reporting physisorption data for gas/solid systems with special reference to the determination of surface area and porosity** (**Recommendations 1984**). *Pure and Applied Chemistry* 1985, **57:**603-619.
- 37. Baggiani C, Baravalle P, Giovannoli C, Anfossi L, Passini C, Giraudi G: Binding behaviour of molecularly imprinted polymers prepared by a hierarchical approach in mesoporous silica beads of varying porosity. Journal of Chromatography A 2011, **1218**:1828-1834.
- 38. Urraca JL, Carbajo MC, Torralvo MJ, González-Vázquez J, Orellana G, Moreno-Bondi MC: Effect of the template and functional monomer on the textural properties of molecularly imprinted polymers. *Biosensors and Bioelectronics* 2008, 24:155.
- 39. Xiao P, Dudal Y, Corvini PFX, Shahgaldian P: Polymeric cyclodextrin-based nanoparticles: synthesis, characterization and sorption properties of three selected pharmaceutically active ingredients. *Polymer Chemistry* 2011, 2:120-125.
- 40. Sellergren B, Hall AJ: Chapter 2. In Molecularly Imprinted Polymers: Man Made Mimics of Antibodies and their Applications in analytical chemistry. Volume 23. Edited by Sellergren B. Amsterdam: Elsevier; 2001: Techniques and Instrumentation in Analytical Chemistry].
- 41. Cormack PAG, Elorza AZ: Molecularly imprinted polymers: synthesis and characterisation. *Journal of Chromatography B-Analytical Technologies in the Biomedical and Life Sciences* 2004, **804**:173-182.
- 42. Brunauer S, Emmett PH, Teller E: Adsorption of Gases in Multimolecular Layers. *Journal Of the American Chemical Society* 1938, **60**:309-319.
- 43. Xu L, Lv R, Wang Y, Gao J: Selective solid-phase extraction of naphazoline using imprinted polymers as matrix prepared by precipitation polymerization. *Journal of Applied Polymer Science* 2010, **118**:881-886.

- 44. Tunc Y, Ulubayram K: Production of Highly Crosslinked Microspheres by the Precipitation Polymerization of 2-(Diethylamino)ethyl Methacrylate with Two or Three Functional Crosslinkers. Journal of Applied Polymer Science 2009, 112:532-540.
- 45. Ye L, Cormack PAG, Mosbach K: Molecularly imprinted monodisperse microspheres for competitive radioassay. *Analytical Communications* 1999, 36:35-38.
- 46. Hall AJ, Quaglia M, Manesiotis P, De Lorenzi E, Sellergren Br: Polymeric Receptors for the Recognition of Folic Acid and Related Compounds via Substructure Imprinting. *Analytical Chemistry* 2006, **78**:8362-8367.
- 47. Rampey AM, Umpleby RJ, Rushton GT, Iseman JC, Shah RN, Shimizu KD: Characterization of the Imprint Effect and the Influence of Imprinting Conditions on Affinity, Capacity, and Heterogeneity in Molecularly Imprinted Polymers Using the Freundlich Isotherm-Affinity Distribution Analysis. Analytical Chemistry 2004, 76:1123-1133.
- 48. Zhao YH, Le J, Abraham MH, Hersey A, Eddershaw PJ, Luscombe CN, Boutina D, Beck G, Sherborne B, Cooper I, Platts JA: Evaluation of human intestinal absorption data and subsequent derivation of a quantitative structure-activity relationship (QSAR) with the Abraham descriptors. Journal of Pharmaceutical Sciences 2001, 90:749-784.
- 49. Ramstrom O, Ye L, Krook M, Mosbach K: Screening of a combinatorial steroid library using molecularly imprinted polymers. *Analytical Communications* 1998, **35:9**-11.
- Spivak DA, Shea KJ: Investigation into the scope and limitations of molecular imprinting with DNA molecules. Analytica Chimica Acta 2001, 435:65-74.
- 51. Spivak DA: **Optimization, evaluation, and characterization of molecularly imprinted polymers.** *Advanced Drug Delivery Reviews* 2005, **57:**1779.
- 52. Umpleby RJ, Baxter SC, Rampey AM, Rushton GT, Chen Y, Shimizu KD: Characterization of the heterogeneous binding site affinity distributions in molecularly imprinted polymers. *Journal of Chromatography B* 2004, 804:141-149.
- 53. Umpleby RJ, Baxter SC, Bode M, Berch JK, Shah RN, Shimizu KD: Application of the Freundlich adsorption isotherm in the characterization of molecularly imprinted polymers. *Analytica Chimica Acta* 2001, **435**:35-42.
- 54. Wei S, Molinelli A, Mizaikoff B: Molecularly imprinted micro and nanospheres for the selective recognition of 17[beta]-estradiol. *Biosensors and Bioelectronics* 2006, 21:1943.
- 55. Kempe M, Mosbach K: **Receptor binding mimetics: A novel molecularly imprinted polymer.** *Tetrahedron Letters* 1995, **36:**3563.
- 56. Mayes AG, Whitcombe MJ: Synthetic strategies for the generation of molecularly imprinted organic polymers. Advanced Drug Delivery Reviews 2005, 57:1742-1778.
- 57. Yoshimatsu K, Reimhult K, Krozer A, Mosbach K, Sode K, Ye L: Uniform molecularly imprinted microspheres and nanoparticles prepared by precipitation polymerization: The control of particle size suitable for different analytical applications. *Analytica Chimica Acta* 2007, **584**:112.
- 58. Yong Jiang A-JT: Synthesis of molecularly imprinted microspheres for recognition of <I>trans</I>-aconitic acid. Journal of Applied Polymer Science 2004, 94:542-547.
- 59. Ye L, Cormack PAG, Mosbach K: **Molecular imprinting on microgel spheres.** *Analytica Chimica Acta* 2001, **435:**187.

- 60. Li WH, Stover HDH: **Porous monodisperse poly(divinylbenzene) microspheres by precipitation polymerization.** Journal of Polymer Science Part a-Polymer Chemistry 1998, **36:**1543-1551.
- 61. Wang JF, Cormack PAG, Sherrington DC, Khoshdel E: Synthesis and characterization of micrometer-sized molecularly imprinted spherical polymer particulates prepared via precipitation polymerization. *Pure and Applied Chemistry* 2007, **79:**1505-1519.

Chapter 4

Applications of corticosteroid imprinted polymers

4. Applications of corticosteroid imprinted polymers

4.1 Introduction

A number of applications, for which molecularly imprinted polymers have been developed, have been discussed in detail in Section 1.6 and Section 3.1. These examples included solid phase extraction (SPE) [1, 2], catalysis [3, 4], drug delivery [5, 6] and chromatography [7, 8]. Many other applications, however, exist, including the use of MIPs in sensor platforms, such as that prepared by Zhou *et al.* [9]. This involved the imprinting of an analogue of domoic acid on a polydopamine film, which was coated onto a quartz crystal microbalance (QCM) sensor. The use of the target selective imprinted polymer as the sensing element for the highly sensitive gravimetric sensor enabled detection of domoic acid down to 5 ppb and could discriminate between the target and two close structural analogues.

QCM sensors incorporating imprinted polymers have also been prepared by, amongst others, Yang *et al.* for billirubin [10] and Lee *et al.* for salivary proteins [11]. Liang *et al.* [12] and Guerreiro *et al.* [13] have employed MIPs as the sensing element in ion selective electrodes for the pesticide chlropyrifos and the antimicrobial chlortetracycline, respectively, while Riskin *et al.* have used imprinted films for the detection of explosives using surface plasmon resonance [14].

Depending on the application, and indeed, the specific end use for each application, different characteristic behaviours of MIPs can be exploited. The use of imprinted polymers in SPE, for example, can employ MIPs capable of targeting a group of analogous molecules in a system, as was the case for the imprinted polymers prepared for the extraction of banned Sudan dyes in food by Baggiani *et al.* [1] and organic acids from a herb sample by Zhu *et al.* [15]. The ability of MIPs to selectively recognise the template molecule when in the presence of close structural analogues was a characteristic of this technology exploited by Jiang *et al.* [7] and Lee *et al.* [8] for the chromatographic separation of structurally similar non-steroidal anti-inflammatory drugs and closely related steroids, respectively.

The potential use of corticosteroid imprinted polymers for various applications has been investigated by a number of research groups, for example, Ramstrom *et al.* have

discussed their potential use in biotechnological industrial processes that utilise catalysts for steroid transformation [16]. The same group has studied the use of molecularly imprinted polymers for corticosterone in the screening of chemical combinatorial libraries [17] while Sreenivasan has prepared a hydrocortisone imprinted polymer capable of the switchable release of testosterone [18].

Baggiani *et al.* have synthesised hydrocortisone imprinted polymers for application as steroid selective adsorbents for solid phase extraction [19, 20]. In the first of these studies, the binding properties of the imprinted polymer were studied using liquid chromatography and while the imprinting factor was highest for the template, at 9.71, significant binding was observed for a number of the structural analogues. The need for adsorbents with group selectivity for corticosteroids was highlighted in the second of these studies, due to the need for regulation and monitoring of steroids in food products and the inherent complexity of such samples.

A 2011 review by Gorog, dealt with 213 articles and evaluated the analysis of steroidal drugs in pharmaceutical and environmental samples [21]. That review highlighted the relevance of developing analytical strategies for steroidal compounds as it discussed the reliance on non-selective methods of steroid analysis of bulk drug materials, quoting figures of 75% of the assays of the Unites States Pharmacopoeia relying on selective chromatography, while the European and Japanese Pharmacopoeias use HPLC for 19% and 54%, respectively. The author goes on to discuss the inherent lack of suitability of many of the non-selective methods for the accurate quantitation of many impurities and degradation products.

The preceding examples demonstrate the need for materials that are capable of group selectivity for corticosteroid compounds, as well as the ability to prepare materials that can separate and quantify corticosteroid compounds with almost identical structures. To this end, corticosteroid imprinted polymers discussed in the preceding chapters have been investigated for use as adsorbents for solid phase extraction using a target analogue approach and for highly selective chromatographic stationary phases capable of differentiating between nearly identical steroidal compounds.

4.2 Research aims and objectives

The aims and objectives of this study were to:

- Investigate the use of the corticosteroid imprinted polymers outlined in the preceding chapters for solid phase extraction using a target analogue approach
- Investigate the use of novel corticosteroid imprinted polymers prepared by precipitation polymerisation for the chromatographic separation of structurally similar corticosteroid analytes

4.3 Experimental

4.3.1 Materials

Solid phase extraction (SPE) cartridges and frits were obtained from Sigma-Aldrich, Ireland. All other materials were as described in Sections 2.3 and 3.3.

4.3.2 Molecularly imprinted solid phase extraction

Dry polymer (MB3 or MB5) (200 mg), prepared as per Section 3.3.2, was packed into 3 mL polypropylene SPE cartridges containing frits with a 20 μ m porosity. The columns were connected to a vacuum manifold and washed with 50 mL acetonitrile to ensure the polymer was evenly packed. Prior to use, packed SPE cartridges were conditioned with 2 × 1 mL acetonitrile followed by 2 × 1 mL H₂O and vacuum dried to remove all traces of solvent. The columns were reconditioned using 2 × 2 mL 10% acetic acid in acetonitrile followed by 2 × 2 mL acetonitrile.

A number of molecularly imprinted solid phase extraction (MISPE) protocols were used (protocol 1-5). Unless otherwise stated the loading step involved the loading of budesonide (BDN) onto conditioned columns in 1 mL H₂O at a concentration 10 μ g.mL⁻¹:

Protocol 1 (P-1):

- Molecular recognition/wash: 1 mL aliquots of water-acetonitrile solutions, composed of sequentially increasing acetonitrile content from 10% to 50%.
- Elution: 1 × 1 mL acetonitrile followed by 3 × 1 mL 10% acetic acid in acetonitrile.

Protocol 2 (P-2):

- Molecular recognition/wash: 1 × 1 mL water-acetonitrile (MB3 60% H₂O; MB5 – 50% H₂O)
- Elution: 1 × 1 mL acetonitrile followed by 3 × 1 mL 10% acetic acid in acetonitrile

Protocol 3 (P-3):

- Molecular recognition/wash: 1×2 mL water-acetonitrile (60% H₂O)
- Elution: 1×2 mL acetonitrile followed by 1×1 mL acetonitrile

Protocol 4 (P-4):

- Load: 1 mL 5 μ g.mL⁻¹ budesonide solution in H₂O
- Molecular recognition/wash: 1×1 mL water-acetonitrile (60% H₂O)
- Elution: 2×2 mL acetonitrile

Protocol 5 (P-5):

- Conditioning: 1 × 2 mL MeOH followed by 1 × 1 mL 0.1 M KOH in MeOH followed by 2 × 5 mL H₂O
- Molecular recognition/wash: 1 mL aliquots of water-acetonitrile solutions, composed of sequentially increasing acetonitrile content from 10% to 40%.
- Elution: 1×2 mL acetonitrile.

P-1 was repeated three times for MB3 and MB5 with stated results calculated as the average of the three individual results. P-2 was carried out in triplicate for MB3 (with results calculated as per P-1) and carried out once for MB5. P-3, P-4 and P-5 were performed once for MB3.

4.3.3 Preparation and evaluation of prednisolone imprinted polymers (M25-M27)

Polymers M25-M27 were prepared using a formulation based on M23 (as outlined in Section 3.3.2) and polymers prepared by Ye *et al.* [22]. Prednisolone was used as the template and a saturated template solution was prepared in dichloromethane (0.27-0.8 g depending on the volume of solvent). Methacrylic acid was then added, followed by trimethylolpropane trimethacrylate (TRIM). The ratio of functional to crosslinking 197

monomer was held constant at 8:4 while the monomer concentration was set at 2% (w/v) for M25 and M26, while a concentration of 1% (w/v) was used for M27. Azobisisobutyronitrile was used as the initiator at a concentration of 3 mol%. Polymerisation was carried out at 4 °C using UV initiation at 350 nm. The control polymer was prepared as per N23 (Section 3.3.2).

Polymers were washed on-line using MeOH acidified with 5% acetic acid until a stable baseline was achieved. Chromatographic evaluation, nitrogen sorption porosimetry and particle size analysis were carried out as per sections 3.3.3, 3.3.4 and 3.3.5, respectively.

4.4 Results and discussion

4.4.1 Molecularly imprinted solid phase extraction of budesonide

4.4.1.1 MISPE protocol P-1

Budesonide (BDN), one of the primary steroidal targets for this study, is a corticosteroid used for a range of medical conditions, including Crohn's disease [23], asthma [24] and has been investigated for use in combating lung cancer [25]. Thus, the ability to extract and quantify budesonide from complex samples could be beneficial for a range of analytical processes. Chapter 3 discusses the use of the target analogue approach to molecular imprinting where a close structural and functional analogue of budesonide, hydrocortisone-17-butyrate (HCB) was used to prepare imprinted polymers with group selectivity for a number of corticosteroids, with a particular emphasis on affinity for budesonide.

In Chapter 3, HCB imprinted polymers MB3 and MB5, prepared using ethylene glycol dimethacrylate (EGDMA) and trimethylolpropane trimethacrylate (TRIM), respectively, as crosslinking monomers and methacrylic acid (MAA) as the functional monomer, had the most suitable affinity for budesonide. Table 3.4 shows that in all but two cases the imprinting factor for budesonide was closest to that of the template, HCB, for these polymers. While the TRIM based polymer prepared in chloroform also gave a high imprinting factor in acetonitrile-water mixtures, the *k*' value was extremely high, which could pose problems during analyte elution. MB3 and MB5 were thus chosen for preliminary MISPE studies.

MISPE protocols P-1 to P-5 were investigated for solid phase extraction experiments, in an effort to observe the greatest degree of molecular recognition in the imprinted polymers while using the lowest number of sample treatment steps. To this end, P-1 employed a molecular recognition step that utilised water-acetonitrile washes increasing in acetonitrile concentration up to 50% (v/v). The elution profiles for P-1 are presented in Table 4.1 and Figure 4.1.





Figure 4.1: BDN elution profiles using MISPE P-1 for HCB imprinted polymers: (a) MB3 (EGDMA) and (b) MB5 (TRIM).

Solvent composition		% Budeso	onide eluted			
	MB3	NB3	MB5	NB5		
90% H ₂ O	0	0	0	0		
80% H ₂ O	0	0	0	0		
70% H ₂ O	0	0.13 (± 0.1)	0.4 (± 0.3)	$0.2 (\pm 0.2)$		
60% H ₂ O	17.2 (± 4.1)	45.9 (± 5.8)	9.0 (± 2.1)	9.1 (± 2.1)		
50% H ₂ O	56.4 (± 2.5)	36.7 (± 5.8)	29.5 (± 6.6)	43.4 (± 6.2)		
100% MeCN	8.5 (± 0.3)	$4.9 (\pm 0.4)$	32.1 (± 8.1)	30.1 (± 6.6)		
10% acetic acid -1	2.3 (± 1.0)	1.4 (± 0.7)	10.8 (± 2.2)	5.7 (± 0.9)		
10% acetic acid -2	1.2 (± 0.7)	0.7 (± 0.2)	2.3 (± 0.3)	1.1 (± 0.3)		
10% acetic acid -3	$0.4 (\pm 0.1)$	0.5 (± 0.1)	1.3 (± 0.1)	$0.5 (\pm 0.2)$		
Total recovered	86.1 (± 8.0)	90.1 (± 13.2)	85.3 (± 18.8)	90.2 (± 15.8)		

 Table 4.1: BDN elution profiles for HCB imprinted polymers: MB3 (EGDMA) and MB5 (TRIM) using MISPE protocol P-1.

Protocol P-1 used a gradient molecular recognition step to enhance the interactions between budesonide and the specific sites in the imprinted polymer with concomitant removal of the budesonide molecules non-specifically bound to both imprinted and control polymers. Figure 4.1 and Table 4.1 demonstrate that for this molecular recognition profile, no budesonide was detected during the 90% and 80% water steps for any of the polymers. This was expected due to the hydrophobic nature of the interaction between BDN and the polymers.

Figure 4.1 (a) shows that when the water content was reduced to 70% 0.13% of the budesonide was eluted from the NB3 column while no budesonide was detected for the MIP. As the MeCN content was further increased and the water reduced to 60%, 45.9% of the budesonide was eluted from NB3 while over 2.5 times less was eluted from MB3. This suggested that the gradient steps used did indeed drive the interaction between the analyte and the specific sites on the polymer. The material so adsorbed, however, was then removed during the next steps where 50% and 100% acetonitrile were used, due to increasing affinity of the solvent for the analyte.

These results suggested that specific sites created for HCB were capable of specifically retaining BDN. This was in agreement with the zonal chromatography on this polymer,

discussed in Section 3.4.2.4, which showed that imprinting factors of 4.9 and 1.7 were obtained for budesonide in MeCN and MeCN-H₂O (30/70), respectively. This was also in agreement with the results for frontal chromatography, which determined that while the difference between the affinity constants was marginal, the MIP had over five times more sites than the NIP (34.8 μ mol.g⁻¹ compared to 6.31 μ mol.g⁻¹).

Figure 4.1 (b) shows that again, the affinity of the solvent for the analyte played a significant role in elution. Similar behaviour was observed for the TRIM based polymers in that it was at 70% water where some budesonide began to elute from both columns. An increase in the acetonitrile content to 40% did not result in any difference between the MIP and the NIP, however, further increase to 50% acetonitrile resulted in 14% more budesonide being retained on MB5 than NB5. As with MB3, further increase to 100% acetonitrile and the use of acidified acetonitrile resulted in the elution of the specifically bound analyte. These results were again in agreement with the chromatographic evaluation of MB5 in Chapter 3, where imprinting factors of 1.8 and 1.3 were obtained for BDN in MeCN and MeCN-H₂O (30/70), respectively. The results from frontal chromatography on this polymer suggested that the number of binding sites was similar for MIP and NIP but that the affinity constant was an order of magnitude higher for the MIP at 2.3×10^5 L.mol⁻¹ compared to the NIP at 8.7×10^4 L.mol⁻¹.

A comparison of the two polymer suggested correlation between the results from the MISPE experiment using P-1 and the imprinting factors observed by chromatographic evaluation. The greatest imprinting factors obtained, i.e. for MB3, resulted in the greatest difference between material specifically retained on the SPE column for a single molecular recognition step. This was during the 60% H₂O step where 17.2% (\pm 4.1) was eluted from the MIP while 45.9% (\pm 5.8) was eluted from the NIP.

An interesting point to note is that for both the EGDMA and TRIM based polymers, the recovery from the NIP was greater by approximately 5%. This suggested that some material was more strongly bound to the MIPs in both cases, a factor that could be related to a greater number of binding sites and to stronger interaction between the analyte and these binding sites, as predicted by frontal chromatography.

Protocols P-2 to P-5 were investigated to determine if the results obtained in P-1 could be improved upon, in terms of molecular recognition or a reduction in the number of steps used. Only the most successful protocols were carried out in duplicate.

4.4.1.2 MISPE protocols P-2 – P-5

Protocol P-2 was an exploratory study carried out to ascertain whether the number of steps utilised in P-1 could be reduced, and if indeed, this would improve the retention of the analyte on the imprinted polymer.

P-2 investigated the possibility of not using the gradient mobile phase increments and instead using the wash step that demonstrated the highest elution of non-specifically bound budesonide in P-1. For MB3 this involved a wash step using 60% H₂O and continuing with the subsequent elution steps as P-1, while the molecular recognition step for MB5 necessitated 50% H₂O. P-2 for MB5 resulted in a difference of only 2% in the quantity of budesonide specifically retained on the imprinted polymer column and, since this polymer also displayed decreased performance in P-2, no further investigations were carried out for MB5. The averaged results for the P-2 protocol carried out in triplicate for MB3 are presented in Table 4.2 and represented graphically in Figure 4.2.

Solvent composition	% Budesonide eluted				
	MB3	NB3			
60% H ₂ O	2.3 (± 0.9)	16.3 (± 9.7)			
100% MeCN	82.5 (± 10)	71.3 (± 16.7)			
10% acetic acid -1	8.7 (± 3.1)	6.7 (± 1.7)			
10% acetic acid -2	2.6 (± 1.3)	1.8 (± 0.7)			
10% acetic acid -3	$0.6(\pm 0.3)$	0.43 (± 0.43)			
Total recovered	96.7 (± 16.2)	96.5 (± 19.1)			

 Table 4.2: BDN Elution profile for HCB imprinted polymer MB3 (EGDMA) using MISPE protocol

 P-2.



Figure 4.2: Elution profile for MB3 (EGDMA) using the P-2 protocol.

The results presented in Table 4.2 and Figure 4.2 show that the molecular recognition wash step induced specific retention of budesonide on the imprinted polymer. In this case, seven times more material was retained on the imprinted polymer than the non-imprinted polymer with 2.3% (\pm 0.9) eluted from the MIP and 16.3% (\pm 9.7) eluted from the NIP. As with P-1, the difference in the amount of material was balanced in the following elution steps. It is worth noting that a greater amount of analyte was recovered for this protocol than for P-1, with ca. 96% recovery for P-2. This may suggest that budesonide was indeed eluting from the columns during the initial molecular recognition steps for P-1 but that the concentration of BDN in these steps was such that it was not distinguishable from baseline noise.

P-3 doubled the volume of the 60% H₂O wash step to determine the effects on the specific retention of budesonide on the imprinted polymers. This resulted in the elution of 71.2% and 80.7% of the loaded analyte for the MIP and NIP, respectively. This suggested that the increased volume of acetonitrile reduced the ability of the polymer to specifically retain budesonide and to discriminate between specific and non-specific binding, as shown in Figure 4.3.



Figure 4.3: Elution profile for the P-3 protocol.

A decreased concentration of budesonide (5 μ g.mL⁻¹) was loaded onto the polymers to determine if this would allow for enhanced molecular recognition, in the event that the original concentration was saturating the specific binding sites. This approach yielded quite similar results to P-2, in that 4.4% and 18.8% budesonide was eluted from the MIP and the NIP, respectively. This is represented graphically in Figure 4.4.



Figure 4.4: Elution profile for MB3 using P-4.

Preliminary work on semi-covalent polymers suggested that treatment of polymers using a basic solution increased the retention of corticosteroid analytes. The P-5 protocol sought to exploit this characteristic of the polymers by pre-conditioning the column with 0.1 M KOH in MeOH. Only minimal differences, however, were observed between the MIP and the NIP during this investigation.

4.4.1.3 Summary of MISPE investigations

The preliminary solid phase extraction studies carried out in this work suggested that with appropriate optimisation of the protocols, the steroid specific binding sites on the hydrocortisone-17-butyrate imprinted polymers could be exploited to retain budesonide to a greater extent than on the non-imprinted polymers. For polymers where trace analysis using SPE is the final application, it is generally necessary to use such a target analogue approach, as template bleed could give false quantitation results.

4.4.2 MIP chromatography for the separation of closely related steroidal compounds

The MISPE experiments outlined in Section 4.4.1 exploited the potential for target analogue imprinting of the HCB imprinted polymers prepared in Chapter 3, allowing for the specific retention of BDN. Another important requirement for molecularly imprinted polymers is the ability to discriminate between closely related compounds. An important application of this would be the separation of structural analogues, for example impurities or related compounds in a system. This would have potential for use in process purification as well as chromatographic applications.

In Chapter 3 of this study, zonal chromatographic evaluation of the polymers prepared suggested that in the majority of cases, steroids with similar log P values and functional moieties appeared to be retained to similar extents on both imprinted and non-imprinted polymers. This resulted in similar retention profiles for the steroids in groups of two, for example, BDN and HCB often behaved similarly, as did hydrocortisone (HYD) and prednisolone (PRD). This was not unexpected, and in fact it was this very property of molecularly imprinted polymers that allowed for use of the target analogue approach for the preparation of the imprinted polymers prepared in Chapter 3.
Somewhat anomalous behaviour, however, was observed for hydrocortisone and prednisolone for the M23 polymers. These polymers were prepared as MAA-co-TRIM polymers, using a precipitation strategy with dichloromethane as the porogen. As stated, in the majority of cases the k' values for HYD and PRD differed by no more than 1.5. For M23, however, the difference was almost 4. The difference between k' values of these analytes on the NIP, N23, was higher than all of the other MIPs, with a value of 2. This suggested that MAA-co-TRIM polymers prepared in this fashion had an inherent affinity for prednisolone. A molecular imprinting strategy using PRD as the template was applied to these polymers, to enhance this inherent affinity and determine if the polymers so prepared could separate these closely related structural analogues.

NMR studies carried out in Chapter 2 and Chapter 3, coupled with the results obtained for precipitation polymers in Chapter 3, indicated that to maximise interaction in the pre-polymerisation complex the concentration of the template should be as high as possible. The solubility of PRD in dichloromethane was relatively low at less than 1 mg/mL. Three polymer formulations were tested for the preparation of PRD imprinted polymers: M25, M26 and M27.

These polymers differed only in the concentration of the template and the monomer concentration. M25 was prepared by saturating the porogen with the template followed by addition of the remaining components to a monomer concentration of 2% (w/v). M26 and M27 used the combined solvating powers of the monomers and the solvent to allow for increased template concentration by first adding the monomers to the porogen, followed by addition of the template. M26 used a 2% (w/v) monomer concentration and achieved an increase of 50% in the concentration of the template. M27 reduced the monomer concentration to 1% (w/v) and in doing so the concentration of the template in the solution was double that of M26 and triple that of M25.

The different polymerisation conditions resulted in particle size ranges of 0.8-7.4 μ m for M25, 0.7-4.8 μ m for M26 and 0.5-4.2 μ m for M27. These results suggest that the particle size range became smaller with increased template or with decreased monomer concentration. The effect on particle morphology can be seen in the SEM images presented in Figure 4.5, showing that predominantly discrete and roughly spherical microparticles were prepared for M25-M27.





Figure 4.5: SEM images for (a) M25; (b) M26; (c) M27.

The primary objective of this study was to achieve the separation of prednisolone and hydrocortisone using, prednisolone imprinted polymers. This was achieved to varying degrees for all three polymers prepared utilising a precipitation polymerisation strategy and 5 cm columns. The separation achieved for M26 at 20 °C at a flow rate of 0.3 mL.min⁻¹ is shown in Figure 4.6.



Figure 4.6: Chromatographic separation of HYD and PRD on PRD imprinted polymers in acetonitrile at 0.3 mL.min⁻¹.

Figure 4.6 shows that the non-imprinted polymers, while displaying different analyte retention times when injected separately, the polymer did not have the capability to resolve the peaks of the two analytes to any significant extent when injected together. As discussed, this suggested that the polymer had an inherent affinity for PRD. Molecular imprinting, using PRD as the template, enhanced this affinity and resulted in the separation of HYD and PRD shown in Figure 4.6. This separation for compounds that differ only in the presence of an extra double bond in the A-ring of PRD, demonstrates the ability of the imprinting process to enhance the inherent affinity of a polymer for a compound and so achieve a high degree of molecular recognition. The selectivity factor, α , for this separation was relatively high, at a value of 1.37, however, the resolution was not optimal because of the typically broad peaks observed with molecularly imprinted polymers. The separation temperature was varied to study the effects on the chromatographic separation and the results are presented in Table 4.3.

Temperature	k' _{HYD}	k' _{PRD}	α
(°C)			
2	6.41	8.99	1.40
10	5.57	7.74	1.39
20	4.85	6.63	1.37
30	4.71	6.17	1.31
50	3.40	4.49	1.32
60	2.96	3.90	1.32

Table 4.3: The effects of temperature on the chromatographic separation of HYD and PRD.



Figure 4.7: Temperature study for the chromatographic separation of HYD and PRD on the PRD imprinted M26 polymer. Acetonitrile was the mobile phase and the flow rate was 0.3 mL.min⁻¹.

Table 4.3 and Figure 4.7 show that while the α value increased as the temperature was decreased, this was accompanied by a concomitant increase in retention time and loss of resolution. While an increase in temperature brought a decrease in retention times, the resolution was again negatively affected. The optimum temperature was seen to be 20 °C as this had an average retention time with the most resolved peaks.

In an effort to improve the resolution, a higher concentration template solution was used in conjunction with a more dilute monomer loading. This resulted in a higher ratio of template to monomer. Investigations in Chapter 2 into the factors affecting particle formation suggested that at lower monomer loading, smaller, more discrete particles should be formed. To this end, M27 was formed. Figure 4.5 (a)-(c) shows that this polymer had a more discrete, microspherical morphology than M25 or M26, with a smaller particle size range of 0.5-4.2 μ m. The separation achieved on this polymer is shown in Figure 4.8.



Figure 4.8: Chromatographic separation of HYD and PRD on M27 in MeCN at 0.3 mL.min⁻¹.

The α value for this separation was similar, though slightly higher, to that for M26, at 1.33. The peaks in this case, however, were resolved to below half the peak height. This allowed for calculation of the resolution (R_s), which gave a value of 0.75. This increase in resolution may have been due to a number of factors, such as the increase in the concentration of the template in the polymerisation solution leading to a possible increase in the number of binding sites or the decrease in particle size leading to increase in combination of the two. The decrease in 209

particle size, however, was accompanied by an increase in back pressure, as can be seen from the spikes observed in Figure 4.8, as well as the shoulder on the hydrocortisone peak. This is most likely due to the fact that the 10% of the particles in the solution were below 0.5 μ m and as such, blockages of the 0.5 μ m frits used could occur.

4.5 Conclusions

The investigation into the use of molecularly imprinted polymers for SPE and chromatography demonstrate that different characteristics of MIP behaviour may be exploited to achieve the specific requirements for the chosen application. The ground bulk particles used for SPE demonstrated the success of the target analogue approach for use in applications where template bleed would interfere with accurate quantitation in trace analysis. Correlations were shown between chromatographic evaluation of the EGDMA and TRIM based polymers used and their selective retention behaviours in SPE. This demonstrated the power of chromatographic evaluation of imprinted polymers during development of polymers for a chosen application.

Novel microspherical prednisolone imprinted polymers were prepared by precipitation strategies and these polymers achieved the successful chromatographic separation of the structural analogues, HYD and PRD, on 5 cm columns. This demonstrated the ability to identify a polymer with inherent affinity for an analyte and to greatly enhance this affinity by molecular imprinting strategies, which increased the number of selective binding sites within the polymer matrix. Knowledge gained from the studies in Chapter 2, on the development of precipitated polymers allowed for an improvement of chromatographic separation and resolution of the target analytes based on altering the morphology of the polymers used. These polymers demonstrated the potential for use in applications such as process purification as well as their suitability for use in conjunction with chromatographic instrumentation capable of dealing with small particle size and high back pressures, such as ultra high performance liquid chromatography (UHPLC).

References

- 1. Baggiani C, Anfossi L, Baravalle P, Giovannoli C, Giraudi G, Barolo C, Viscardi G: Determination of banned Sudan dyes in food samples by molecularly imprinted solid phase extraction-high performance liquid chromatography. *Journal of Separation Science* 2009, **32**:3292-3300.
- 2. Theodoridis G, Kantifes A, Manesiotis P, Raikos N, Tsoukali-Papadopoulou H: Preparation of a molecularly imprinted polymer for the solid-phase extraction of scopolamine with hyoscyamine as a dummy template molecule. *Journal of Chromatography A* 2003, **987**:103-109.
- 3. Liu JQ, Wulff G: Molecularly imprinted polymers with strong carboxypeptidase A-like activity: Combination of an amidinium function with a zinc-ion binding site in transition-state imprinted cavities. *Angewandte Chemie-International Edition* 2004, **43**:1287-1290.
- 4. Huang X, Yin Y, Liu Y, Bai X, Zhang Z, Xu J, Shen J, Liu J: **Incorporation of glutathione peroxidase active site into polymer based on imprinting strategy.** *Biosensors and Bioelectronics* 2009, **25:**657-660.
- 5. Wang C, Javadi A, Ghaffari M, Gong S: A pH-sensitive molecularly imprinted nanospheres/hydrogel composite as a coating for implantable biosensors. *Biomaterials* 2010, **31**:4944-4951.
- 6. Vaughan AD, Zhang JB, Byrne ME: Enhancing Therapeutic Loading and Delaying Transport via Molecular Imprinting and Living/Controlled Polymerization. *Aiche Journal* 2010, **56**:268-279.
- 7. Jiang JB, Song KS, Chen Z, Zhou Q, Tang YW, Gu FL, Zuo XJ, Xu ZG: Novel molecularly imprinted microsphere using a single chiral monomer and chirality-matching (S)-ketoprofen template. *Journal of Chromatography A* 2011, **1218**:3763-3770.
- 8. Lee WC, Hung CY, Hwang CC: Separation of cholesterol from other steroids using molecularly imprinted polymer prepared by seeded suspension polymerization. *Chemical And Biochemical Engineering Quarterly* 2008, 22:151-156.
- 9. Zhou W-H, Tang S-F, Yao Q-H, Chen F-R, Yang H-H, Wang X-R: A quartz crystal microbalance sensor based on mussel-inspired molecularly imprinted polymer. *Biosensors and Bioelectronics* 2010, **26**:585-589.
- 10. Yang Z-p, Yan J-l, Zhang C-j, Luo S-q: Enhanced removal of bilirubin on molecularly imprinted titania film. *Colloids and Surfaces B: Biointerfaces* 2011, 87:187-191.
- 11. Lee M-H, Thomas JL, Tseng H-Y, Lin W-C, Liu B-D, Lin H-Y: Sensing of Digestive Proteins in Saliva with a Molecularly Imprinted Poly(ethylene-co-vinyl alcohol) Thin Film Coated Quartz Crystal Microbalance Sensor. ACS Applied Materials & Interfaces 2011, 3:3064-3071.
- 12. Liang R-N, Song D-A, Zhang R-M, Qin W: Potentiometric Sensing of Neutral Species Based on a Uniform-Sized Molecularly Imprinted Polymer as a Receptor. Angewandte Chemie International Edition 2010, 49:2556-2559.
- 13. Guerreiro JRL, Freitas V, Sales MGF: New sensing materials of molecularlyimprinted polymers for the selective recognition of Chlortetracycline. *Microchemical Journal* 2011, **97:**173-181.
- Riskin M, Ben-Amram Y, Tel-Vered R, Chegel V, Almog J, Willner I: Molecularly Imprinted Au Nanoparticles Composites on Au Surfaces for the Surface Plasmon Resonance Detection of Pentaerythritol Tetranitrate, Nitroglycerin, and Ethylene Glycol Dinitrate. Analytical Chemistry 2011, 83:3082-3088.

- 15. Zhu T, Li SN, Row KH: Molecularly Imprinted Monolithic Material for the Extraction of Three Organic Acids from Salicornia herbacea L. Journal of Applied Polymer Science 2011, 121:1691-1696.
- 16. Ramstrom O, Ye L, Mosbach K: Artificial antibodies to corticosteroids prepared by molecular imprinting. *Chemistry & Biology* 1996, **3**:471-477.
- 17. Ramstrom O, Ye L, Krook M, Mosbach K: Screening of a combinatorial steroid library using molecularly imprinted polymers. *Analytical Communications* 1998, **35**:9-11.
- 18. Sreenivasan K: On the application of molecularly imprinted poly(HEMA) as a template responsive release system. *Journal of Applied Polymer Science* 1999, **71:**1819-1821.
- 19. Baggiani C, Baravalle P, Giovannoli C, Anfossi L, Giraudi G: Molecularly imprinted polymers for corticosteroids: Analysis of binding selectivity. *Biosensors and Bioelectronics* 2010, **26**:590-595.
- 20. Baggiani C, Giraudi G, Trotta F, Giovannoli C, Vanni A: Chromatographic characterization of a molecular imprinted polymer binding cortisol. *Talanta* 2000, **51**:71-75.
- 21. Sandor G: Advances in the analysis of steroid hormone drugs in pharmaceuticals and environmental samples (2004-2010). *Journal of Pharmaceutical and Biomedical Analysis* 2011, **55**:728-743.
- 22. Ye L, Cormack PAG, Mosbach K: Molecularly imprinted monodisperse microspheres for competitive radioassay. *Analytical Communications* 1999, 36:35-38.
- 23. Rivkin A: Certolizumab pegol for the management of Crohn's disease in adults. *Clinical Therapeutics* 2009, **31:**1158.
- 24. Sobande PO, Kercsmar CM: Inhaled corticosteroids in asthma management. *Respiratory Care* 2008, **53:**625.
- 25. Yao R, Wang Y, Lemon WJ, Lubet RA, You M: Budesonide exerts its chemopreventive efficacy during mouse lung tumorigenesis by modulating gene expressions. *Oncogene* 2004, 23:7746.

Chapter 5

The synthesis of tailor-made functional monomers for corticosteroid molecular imprinting

5. The synthesis of tailor-made functional monomers for corticosteroid molecular imprinting

5.1 Introduction

The strategies employed to create binding sites within molecularly imprinted polymers, as discussed in Section 1.3, include covalent, non-covalent, semi-covalent and stoichiometric non-covalent methodologies. Of these routes to binding site creation, the non-covalent approach has been utilised to the greatest extent [1, 2]. Due to wide ranging applicability, the use of hydrogen bonding and ionic interactions appear to have predominated in this strategy, with monomers such as methacrylic acid [3] and 4-vinyl pyridine [4] typically employed for the former, while 2-(diethylamino) ethyl methacrylate [5] and 2-acryloylamido-2,2'-dimethylpropane sulfonic acid [6] have been utilised as cationic and anionic functional monomers, respectively.

However, for true biomimetic imprinting, templating techniques must exploit as many of the possible interactions, weak or otherwise, between the template and the interacting species. To this end, non-covalent molecular imprinting has also taken advantage of hydrophobic interactions, and, in a manner befitting biomimetic polymers, has been exemplified by the imprinting of the peptide, melittin, using the hydrophobic monomer N-*t*-butylmethacrylamide, by Hoshino *et al.* enabling *in vivo* recognition of the template by so-called 'plastic antibodies' [7, 8].

While some hydrophobic interactions could be considered relatively weak, there are compounds that can achieve high association constants based on predominantly hydrophobic interactions. One such set of compounds are the cyclodextrins, cyclic oligosaccharides composed of D-glucose units joined by α -1,4-glucosidic linkages [9]. Due to steric factors, cyclodextrins (CD) composed of less than six glucose units do not exist and the most common types of CD contain six, seven or eight glucose units and are named α -CD, β -CD and γ -CD, respectively [10], the chemical structures of which are presented in Figure 5.1.



Figure 5.1: Chemical structures of α -CD, β -CD and γ -CD.

Cyclodextrins have a three-dimensional structure that is generally considered to be a truncated cone, bearing the hydroxyl groups on the outer surface of the cone [10]. Primary and secondary hydroxyl groups are located on the narrow and wider side of this cone, respectively (Figure 5.2).



Figure 5.2: Three-dimensional structure of the cyclodextrin molecule represented as a truncated cone with n= 6, 7 and 8 for α -CD, β -CD and γ -CD, respectively [10].

The hydroxyl groups, on what is essentially the exterior rim of the CD molecule, makes them water-soluble while the interior is composed of C-C bonds and ether linkages, making it relatively hydrophobic. It is this hydrophobic cavity that allows for the formation of inclusion complexes with lipophilic compounds of appropriate size that display high association constants, often in the order of 10^3 or 10^4 M⁻¹ [10-13].

Inclusion complexes (driven primarily by hydrophobic and van der Waal's interactions [10]) may be represented schematically, as in Figure 5.3.



Figure 5.3: Schematic representation of the formation of an inclusion complex formed by a cyclodextrin and a lipophilic compound.

In light of the ability of cyclodextrins to form such inclusion complexes, it is little wonder that their use as functional monomers in molecular imprinting technologies has been investigated for a range of compounds, including pharmaceutical actives such as dextromethorphan [14], herbicides such as aciflourfen [15] and aromatic amino acids [16].

The relatively hydrophobic nature of the steroidal skeleton suggests these compounds as likely candidates for strong interaction with cyclodextrins and this has been confirmed by the wealth of literature investigating these interactions [17-20]. As such, a number of groups have employed these cyclic oligosaccharides as the complex-forming monomers for steroids, most notably that of Komiyama, where different steroids have been imprinted using cyclodextrins as functional monomers.

Komiyama *et al.* have used diisocyanate compounds to crosslink the cyclodextrin 'monomers' in a condensation reaction that forms a monolithic polymer and have achieved imprinting effects that resulted in fifteen times more material adsorbed to the cholesterol-imprinted polymer than the non-imprinted polymer [21]. Using cholesterol specific imprinted polymers, also prepared with toluene-diisocyanate and β -cyclodextrin, they achieved cholesterol retention times up to twenty four times longer on imprinted columns than non-imprinted columns [22].

Komiyama has also studied the use of 'tailor-made' vinylated cyclodextrin monomers which allowed the use of conventional crosslinking monomers, such as *N*,*N*'methylenebisacrylamide [23], for imprinting on vinylated silica particles [24]. In these studies, the vinyl monomers of 6-*O*- α -D-glucosyl- β -cyclodextrin and α -CD were synthesised by way of an ester exchange reaction with *m*-nitrophenyl acrylate. ¹H NMR and MALDI-TOFMS were used to characterise the vinylated cyclodextrin. It was shown that despite testing various reaction conditions, between one and three vinyl groups were invariably added to the CDs, with column chromatography unable to isolate individual compounds from these mixtures. ¹H NMR provided average numbers of 1.3 and 1.2 vinyl groups per CD for 6-*O*- α -D-glucosyl- β -cyclodextrin and α -CD, respectively.

Custom-made functional monomers based on modified cyclodextrins have also been employed by Piletsky *et al.*, in conjunction with an ionic monomer, for the imprinting of phenylalanine [6, 25]. As was the case for Komiyama *et al.*, the vinylated cyclodextrin was again synthesised as a mixture of differently substituted CDs, which ¹H NMR showed had an average of 1.9 vinyl groups per CD molecule.

Liu *et al.* employed a different approach in which spherical poly(glycidyl methacrylate) with β -cyclodextrin anchored to the surface, was used to prepare molecularly imprinted microspheres for use in the solid phase extraction of ursolic acid [26]. Ursolic acid was bound within the hydrophobic cavity of the cyclodextrin and was subsequently entrapped by crosslinking of surface bound 2-hydroxyethyl methacrylate (HEMA) and the crosslinker, ethylene glycol dimethacrylate. This resulted in molecularly imprinted microspheres where the capacity of the MIP for ursolic acid was 42.5 µmol.g⁻¹, while the capacity of the NIP was almost ten times less, at 4.9 µmol.g⁻¹.

Sreenivasan, however, in a study using cyclodextrin modified HEMA, adds a cautionary note to the use of cyclodextrin monomers for the imprinting of hydrophobic molecules as their ability to form inclusion complexes with any suitably sized lipophilic molecule, may significantly reduce the selectivity of the resultant polymer [27].

Another approach, which involves the use of custom-made monomers, is the semicovalent strategy. This method is discussed in detail in Section 1.3.3 and for this approach the monomers are not prepared to interact in a particular fashion with the template. Rather, a polymerisable analogue of the template is synthesised in an effort to construct binding sites with the fidelity achieved by covalently formed sites, while simultaneously allowing for the rapid rebinding kinetics of non-covalent interactions [28, 29].

As discussed in Section 1.3.3, this approach has been used for a range of templates including glucopyranoside [29], propazine [28] and bisphenol-A [30]. Where binding sites are created by relatively weak interactions, such as in the case of templates with limited functionality, it would be expected that a more directed structuring of the interacting functional moieties within the binding cavities could lead to a higher degree of specificity and selectivity. Given the weak interaction observed between corticosteroids and functional monomers throughout this study, the semi-covalent approach was considered a potentially useful strategy for the creation of corticosteroid molecularly imprinted polymers.

5.2 Research aims and objectives

The aims and objectives of this study were:

- To synthesise tailor-made functional monomers for use in the preparation of corticosteroid molecular imprinting.
- To study the interaction between cyclodextrins and budesonide and to assess the feasibility of cyclodextrin based functional monomers for corticosteroid imprinting.
- To carry out an investigation of a number of synthetic strategies for the preparation of polymerisable cyclodextrin monomers.

- To synthesise a polymerisable template-monomer based on a suitable corticosteroid compound for use in a semi-covalent imprinting strategy.
- To carry out preliminary investigations into the use of the prepared custom monomers for the synthesis of corticosteroid molecularly imprinted polymers.

5.3 Experimental

5.3.1 Materials

β-cyclodextrin (β-CD) (≥ 97%), γ-cyclodextrin (γ-CD) (≥ 99%), toluene-2,4diisocyanate (TDI) (≥ 98%), acryloyl chloride (AC) (≥ 97%), methacryloyl chloride (MAC) (≥ 97%), N-(hydroxymethyl) acrylamide (NHA) (48% (w/v) in H₂O), dimethyl sulfoxide-d₆ (99.9% atom D), deuterium oxide (99.99% atom D), anhydrous pyridine (≥ 99.8%), anhydrous dimethyl sulfoxide (≥ 99.9%), *N*, *N*'-methylene bisacrylamide (MBA) (≥ 98%), 4-(dimethylamino)pyridine (DMAP) (≥ 99%), triethylamine (TEA) (≥ 99%) and hydrochloric acid (37%) were all obtained from Sigma-Aldrich, Wicklow, Ireland. All other solvents and reagents used were as per Table 2.3 and Section 3.3.1.

5.3.2 Instrumentation

LC-MS analysis was carried out on using an Agilent 1200 series LC coupled to an XCT Ultra ion trap. All other instrumentation was as per Table 2.4 and Table 3.1.

5.3.3 Monomer synthesis

5.3.3.1 Cyclodextrin methacrylate

Cyclodextrin methacrylate (CDM) was prepared using a method adapted from that used by Xu *et al.* [15, 31]. β -cyclodextrin (1 mmol) was dissolved in 50 mL freshly distilled anhydrous pyridine and the temperature was reduced to 0° C. Methacryloyl chloride (2.2-5 equivalents) was dissolved in 10 mL sodium-dried ether at 0 °C. The methacryloyl chloride solution was added in a drop-wise manner to the cyclodextrin solution and the reaction was allowed to proceed for 24 hours. The pyridine was removed by evaporation *in vacuo* at 50 °C and the resultant solid was re-dissolved in ethanol. The ethanol was then removed by evaporation *in vacuo* at 50 °C and the resultant solid was stored at 4 °C until further use.

5.3.3.2 Acrylamidomethyl cyclodextrin

Acrylamidomethyl cyclodextrins (β -CDA and γ -CDA) were prepared in a fashion based on the method used by Lee *et al.* for the synthesis of polymerisable cyclodextrin monomers [32]. In the present study, β - or γ -cyclodextrin and N-(hydroxymethyl) acrylamide (NMA) were dissolved in water, using ratios ranging from 1:10 to 1:90, and the temperature was raised to 80 °C. Sufficient hydrochloric acid was added to form a 1% (v/v) solution and the reaction was allowed to proceed for between 5 and 45 minutes. The reaction mixture was then added to acetone (50 mL reaction mixture to 200 mL acetone) to stop the reaction and precipitate the product. The reaction so quenched was then stored overnight at 4 °C. The resultant precipitate was removed by filtration, washed repeatedly with acetone, and stored at 4 °C until further use.

5.3.3.3 Hydrocortisone methacrylate

Hydrocortisone (2.5 mmol), triethylamine (3 mmol) and a catalytic amount of 4-(dimethylamino)pyridine (11-12 mg) were dissolved in 50 mL THF and the solution was cooled to 0 °C. Methacryloyl chloride (3 mmol) was dissolved in ice-cooled THF and added to the magnetically stirred reaction mixture in a drop-wise fashion. The reaction was allowed to proceed for 24 hours and then filtered. Dichloromethane was added the solution was washed with water three times. After separation of the two layers the remaining dichloromethane was dried over magnesium sulphate and the solvent was evaporated *in vacuo* to give a solid, which was stored at 4 °C until further use.

5.3.4 Polymer synthesis

5.3.4.1 Cyclodextrin molecularly imprinted polymer 1 (CDMP1)

Cyclodextrin polymer 1 (CDMP1) was prepared using a method employed by Hishiya *et al.* [33], which used toluene-2,4-diisocyanate (TDI) as the crosslinking component. In the present study, the template (budesonide) (0.3 mmol) and β -cyclodextrin (0.88 mmol) were dissolved in anhydrous dimethyl sulfoxide (10 mL) and allowed to equilibrate for 24 hours at room temperature. TDI (5.6 mmol) was then added and the temperature was raised to 65 °C for two hours with magnetic stirring. The resultant polymer was then precipitated by adding the reaction mixture to acetone (200 mL). The irregular shaped microparticles obtained were washed repeatedly with hot water, 220

ethanol and tetrahydrofuran. Cyclodextrin non-imprinted polymer 1 (CDNP1) was prepared in a similar manner but in the absence of the template.

5.3.4.2 Cyclodextrin molecularly imprinted polymer 2 (CDMP2)

Cyclodextrin polymer 2 (CDMP2) was prepared using hydrocortisone-17-butyrate (HCB) as the template and an acrylamidomethyl cyclodextrin monomer (CDMA) as the functional monomer. CDMA and HCB were dissolved in dimethyl formamide (DMF) and allowed to equilibrate for 24 hours at room temperature. *N*, *N'*-methylenebisacrylamide was utilised as the crosslinking monomer and was added after 24 hours. The template:functional monomer: crosslinking monomer ratio was 1:3:20 and the monomer loading was 28% (w/v). Azobisisobutyronitrile was used as the initiator, at a concentration of 1 mol% and polymerisation was carried out at 60 °C for 24 hours. Soxhlet extraction was carried out in methanol and the polymer was ground and sieved as per Section 3.3.2. Cyclodextrin non-imprinted polymer 2 (CDNP2) was prepared in the same manner but without the template.

5.3.4.3 Cyclodextrin molecularly imprinted polymer 3 (CDMP3)

Cyclodextrin polymer 3 (CDMP3) was prepared by precipitation methods. HCB was used as the template, acrylamidomethyl γ CD (γ -CDA) and methacrylic acid (MAA) as co-functional monomers and ethylene glycol methacrylate (EGDMA) as the crosslinking monomer in a 1:2:4:30 ratio. Acetonitrile-water 60/40 (v/v) was used as the porogen and the monomer concentration was 1% (w/v).

5.3.4.4 Hydrocortisone methacrylate imprinted polymer 1 (HMAMP1)

Hydrocortisone methacrylate imprinted polymer 1 (HMAMP1) was prepared in bulk format using hydrocortisone methacrylate, methacrylic acid and divinylbenzene in a 1:2:20 ratio in acetonitrile-toluene 3:1 (v/v) at a monomer concentration of 70% (w/v). AIBN was used as the initiator at 1 mol% and polymerisation was carried out at 60 °C for 24 hours. The polymer was ground and sieved to 25-53 μ m size range. Basic hydrolysis was carried out by reflux in 1 M methanolic KOH for 6 hours followed by washing with dilute HCl using a method based on that employed by Cacho *et al.* [28]. The control polymer (HMANP1) was prepared without the template-monomer and to ensure the presence of a similar proportion of functional groups a MAA:DVB ratio of 4:20 was used.

5.3.4.5 Hydrocortisone methacrylate imprinted polymer 2 (HMAMP2)

Hydrocortisone methacrylate imprinted polymer 2 (HMAMP2) was prepared *via* a precipitation strategy based on that used for DVB in chapters 2 and 3. The same ratio of HMA:MAA:DVB was employed as for HMAMP1 but the monomer concentration was reduced to 4% (w/v). AIBN was used as the initiator at a concentration of 3 mol% and polymerisation was carried out at 60 °C for 24 hours with agitation at 20 RPM. The non-imprinted polymer, HMANP2, was prepared in the same manner with the component ratio as per HMANP1, again, in the absence of the template.

5.3.5 NMR

NMR analyses were carried out at 18.7 ± 0.2 °C. For routine identification, chemical shifts were referenced to the TMS peak. NMR binding studies were carried out by preparing a series of solutions where the concentration of budesonide was held constant (typically between 2-5 mM) and the concentration of the β -cyclodextrin was varied, giving molar ratios ranging from 0 to 20. The solutions were allowed to equilibrate for 24 hours before analysis and methanol was used as the internal reference for chemical shift determination.

5.3.5.1 NMR characterisation

The ¹H NMR peaks of β -CD in DMSO-*d*₆ were assigned as follows: H-1 = 4.68; H-2 = 3.29; H-3 = 3.64; H4 = 3.34; H-5 = 3.59; H-6 = 3.64; OH-2 = 5.52; OH-3 = 5.48; OH-6 = 4.26 [9]. Vinyl protons for β -CD methacrylate (CDM) were assigned to the multiplets from 5.5-6 ppm. Vinyl protons for acrylamidomethyl β -CD (β -CDA) were assigned to the multiplets between 5.5-6.3 ppm. The ¹H NMR peaks of γ -CD in D₂O were assigned as follows: H-1 = 5.11 ppm; H-2 = 3.65 ppm; H-3 = 3.93 ppm; H-4 = 3.59 ppm; H-5 = 3.86 ppm; H-6 = 3.87 ppm [34]. Vinyl protons were assigned to multiplets at 5.5-6.5 ppm. Vinylic carbons for all derivatised CD compounds were assigned to peaks at 120-130 ppm and carbonyl carbons were assigned to peaks at 170 ppm.

Vinyl protons for hydrocortisone methacrylate were assigned to multiplets at 5.3-6.0 ppm. Vinylic carbons for all derivatised compounds were assigned to peaks at 120-130 ppm and carbonyl carbons were assigned to peaks at 170 ppm.

5.3.6 LC-MS

LC-MS analyses employing acetonitrile-water mixtures were used as the mobile phase at a ratio of either 95:5 or 50:50 (v/v) and the flow rate was set at 0.5 mL.min⁻¹. An Agilent 150×4.6 mm 5 µm Eclipse XDB-C₁₈ column was used for all analyses.

5.3.6.1 LC-MS characterisation

MS assignment for β -CD: MNa⁺ = m/z 1157.5; CDM: monosubstituted: MNa⁺ = m/z 1225.5; disubstituted: MH⁺ = 1271.5; trisubstituted: MH⁺ = m/z 1339.5; β -CDA: monosubstituted: MH⁺ = m/z 1218.5; disubstituted: MH⁺ = m/z 1323.5; γ -CD: MH⁺ = m/z 1298.5; MNa⁺ = m/z 1319.5; γ -CDA: monosubstituted: MH⁺ = 1380.5; disubstituted: MHa⁺ = m/z 1485.5; trisubstituted: MH⁺ = m/z 1568.6; Hydrocortisone: dimer = m/z 725.6; hydrocortisone methacrylate: monosubstituted: MH⁺ = 431.6; dimer = m/z 861.5.

5.3.7 SEM and nitrogen sorption porosimetry

SEM and nitrogen sorption porosimetry were carried out as per Sections 2.3.5 and 3.3.4, respectively.

5.3.8 Equilibrium binding studies

Equilibrium binding studies were carried out as per Section 2.3.4. Briefly, dry polymer (50 mg) was placed in MeOH-H₂O (70/30) solution at budesonide concentrations of 2-6 μ g/mL. The suspended polymer solution was then placed in a C24 incubator shaker from New Brunswick Scientific (Edison, NJ, USA) at 20 °C at 300 RPM. After 24 hours the suspended polymer solution was centrifuged at 45000 RPM for 1 hour and the supernatant was filtered through 0.45 and 0.2 μ m syringe filters into clean HPLC vials. This was then analysed as outlined in Section 2.3.4.

5.3.9 Column packing and chromatographic evaluation

Column packing and chromatographic evaluation were carried out as per Section 3.3.3.

5.4 Results and discussion

Chapters 2 and 3 of this study outlined the preparation and characterisation of molecularly imprinted polymers capable of the selective recognition of corticosteroids. NMR studies were carried out with a view to achieving a rational understanding of the performance of these polymers (sections 2.4.4 and 3.4.1). These studies, coupled with the results observed for novel corticosteroid-imprinted precipitated microspheres and microparticles (Section 3.4.4), suggested that the weak interactions observed between template and functional monomer in the pre-polymerisation mixture resulted in the creation of a relatively low number of binding sites in the resultant polymers, particularly for polymers prepared in dilute solutions such as precipitation polymers.

This chapter focuses on the synthesis of tailor-made functional monomers capable of enhancing the number of corticosteroid specific binding sites within the polymeric matrix, using two separate strategies. The first strategy involved promotion of templatemonomer complex formation in the pre-polymerisation complex and the second involved the use of the synthesis of a polymerisable template-monomer for use in a semi-covalent imprinting strategy.

5.4.1 Cyclodextrin compounds as complex-forming agents for molecular imprinting

5.4.1.1 NMR studies on cyclodextrin-steroid interaction

The strong interaction between steroids and cyclodextrins discussed in the introduction, points to the applicability of cyclodextrins for use as complex-forming agents in molecular imprinting. To investigate their potential for interaction with the corticosteroids used in this study, BDN and HCB, binding studies were carried out to elucidate the strength and nature of the interactions taking place. α -cyclodextrin was not investigated, as a number of studies have suggested that prednisolone and hydrocortisone, i.e. structural analogues of BDN and HCB, can not enter the α -CD cavity [17, 18]. Due to the cost of γ -CD, preliminary studies were carried out using β -CD. The binding interactions taking place between BDN and β -CD were studied in DMSO using NMR titration investigation. The protons which were most responsive to complexation and produced the most notable complex induced shifts (CIS) were monitored [35]. The resonance signals for the H-1 proton on the A-ring of BDN and the

H-3 proton for β -CD were observed to be the most responsive and plots (presented in Figure 5.4) were constructed.



Figure 5.4: Plots constructed for $\Delta\delta$ vs. [β -CD]/[BDN] for (a) BDN H-1 proton and (b) β -CD H-3 proton obtained from NMR binding studies carried out in DMSO.

The CIS of the steroidal H-1 suggested that the interaction between the two compounds involved the A-ring of the steroid. The stereochemistry of cyclodextrins is such that the H-3 and H-5 protons point inwards to the cavity of the three-dimensional truncated cone represented in Figure 5.2, and as such, CIS observed for these protons are indicative of inclusion into the hydrophobic cavity of the CD [12, 17]. The shift observed for the CD H-3 proton is therefore evidence of the inclusion of BDN into the hydrophobic cavity [12].

The combination of the CIS observed for both locations, coupled with the absence of any significant CIS for the H-5 proton suggested the interaction represented schematically in Figure 5.5 where the A-ring of the steroid interacts with the hydrophobic cavity of β -CD



Figure 5.5: Proposed interaction between budesonide and β-CD.

Another factor, suggesting the interaction represented in Figure 5.5, was the almost identical dissociation constants observed for the two protons, having values of 2.2×10^{-3} M and 2.5×10^{-3} M, for the steroidal H-1 and the cyclodextrin H-3, respectively. It is worth noting that these dissociation constants suggested far greater interaction than had been observed for BDN and acetic acid or 2-(hydroxyethyl) methacrylate in Chapter 2 (k_{diss} 0.058-1.46 M).

This evidence suggesting strong hydrophobic interaction between BDN and β -CD and the formation of a complex in polar solutions, demonstrated the potential for using CD based functional monomers to enhance the formation of template-monomer complexes in the pre-polymerisation solution. Based on this, a number of strategies were investigated for their capability to exploit this enhancement of complex formation.

5.4.1.2 Synthesis and evaluation of cyclodextrin molecularly imprinted polymer 1 (CDMP1)

CDMP1 was prepared using a method analogous to that used by Komiyama *et al.* [21, 33, 36]. This method utilises toluene-2,4-diisocyanate (TDI) to crosslink native cyclodextrins *via* a condensation reaction of the OH groups of the cyclodextrin, as depicted in Figure 5.6.



Figure 5.6: Crosslinking of cyclodextrins with toluene-2,4-diisocyanate showing interaction at C-6 OH [21, 33, 36].

In the present study, BDN and β -CD were added to DMSO and allowed to equilibrate for 24 hours. TDI was then added and the temperature was raised to 65 °C for 2 hours. The resultant polymer did not need to be ground and sieved as irregular shaped particles in the micron size range were formed. It was decided to first test the capacity of these polymers using only the non-imprinted polymer in equilibrium binding studies described in Section 5.3.7. The concentration range tested was between 2 and 6 µg/mL and the results for non-specific binding on the non-imprinted polymer are shown in Figure 5.7.



Figure 5.7: Percentage of the template bound by the non-imprinted polymer CDNP1 over a range of template concentrations.

The results presented in Figure 5.7 show that the percentage of BDN bound to the nonimprinted polymer decreased as the concentration of the template increased. This suggested that the lowest concentration solution, $3 \mu g/mL$, should be used for binding studies using the MIP. This template solution concentration was applied during equilibrium binding studies on CDMP1 and CDNP1. During these studies it was observed that the amount of template bleed from the imprinted polymer was ten times the concentration of the rebinding solution. This prevented further use of these polymers as it appeared that not only was binding capacity relatively low in the first place, the amount of entrapped template that leeched from the polymer during binding studies was too high for any meaningful investigation.

The high degree of entrapped template may have been due to the nature of the crosslinking component, TDI, which would form a relatively short spacer between two cyclodextrin monomers (Figure 5.7). Hishiya *et al.* have carried out spectroscopic analysis of the complexes formed with TDI and have observed intra molecule linkages as well intermolecular crosslinkers involving more than one TDI molecule [37]. In the present study, the formation of such complexes may have played a role in the high degree of template bleed. There was also some evidence of solubilisation of the polymer in particular solvents, such as DMSO and solutions with high water contents. The combination of these factors suggested the necessity to prepare polymers using more conventional crosslinking monomers. This in turn, required the synthesis of tailor-made cyclodextrin monomers containing polymerisable vinyl groups.

5.4.1.3 Synthesis of polymerisable cyclodextrin monomers

The objective of this study was to prepare CD-based compounds, derivatised with polymerisable vinyl groups, for use in the preparation of corticosteroid molecularly imprinted polymers. A number of strategies were investigated for suitability. The aim was to successfully derivatise β -CD and γ -CD and to characterise them as fully as possible by NMR and LC-MS, given the limitations of both the synthetic and analytical methods described in Section 5.1, relating to previously published studies, i.e. the ability to provide an average number of substituted groups per CD molecule [6, 23, 25].

Cyclodextrin methacrylate (CDM) was prepared according to a method employed by Xu *et al.* (CDM) [15, 31]. This was a relatively straightforward synthetic strategy (Figure 5.8), where 2.2 equivalents of methacryloyl chloride (MAC), dissolved in ether, were added in a drop-wise fashion to a solution of β -CD in freshly distilled pyridine at 0 °C and reacted for twenty-four hours.



Figure 5.8: Synthesis of CDM showing reaction at C-6 OH.

The synthetic strategy for the derivatisation of β -CD using MAC in Figure 5.8 depicts derivatisation of the primary C-6 OH position of the CD. While this would be expected to react preferentially with the acid chloride, the secondary hydroxyl groups, C-2 OH and C-5 OH, could also react and the extent to which each group becomes derivatised would be difficult to accurately quantify using the methods outlined in this study. As stated in the introduction these monomers are typically prepared as a mixture of differently substituted CDs and an average figure for derivatisation is usually assessed.

The success of this synthetic strategy was monitored using both NMR and LC-MS to identify the obtained derivatised monomers and to quantify the average number of vinyl groups per cyclodextrin monomer. The MS, ¹H NMR, and ¹³C NMR spectra for the starting material, β -CD, are presented in Figure 5.9 (a)-(c). (Peak assignment of the ¹H NMR spectrum from reference [9]).







Figure 5.9: Spectra obtained for β -CD using (a) LC-MS, (b) ¹H NMR, (c) ¹³C NMR (NMR carried out in DMSO- d_6).

It is important to note the presence of sharply defined OH peaks in Figure 5.9 (a), as a reduction or broadening of these peaks, with concomitant appearance of vinyl proton signals would indicate the addition of the vinyl groups at these sites. The absence of any vinylic or carbonyl carbon resonance signals in Figure 5.9 (c) is also important, as the introduction of these groups would suggest success of the reaction.

A number of different ratios of β -CD to MAC were tested ranging from 1:2.2-1:5 and it was observed that the most successful ratio was 1:2.2. The spectra obtained for the CDM product are presented in Figure 5.10 (a)-(c).







Figure 5.10: Spectra obtained for the CDM product: (a) MS, (b) ¹H NMR and (c) ¹³C NMR (NMR carried out in DMSO- d_6).

Figure 5.10 (a)-(c) suggested that β -CD was successfully derivatised using methacryloyl chloride, with a number of different substituted vinyl derivatised CD monomers. Figure 5.10 (a) shows the mass spectrum of the product. The peak at m/z 1157.5, the sodium adduct of β -CD, suggests that some starting material remained. There was strong evidence for the presence of β -CD derivatised to various extents, with peaks at m/z 1225.5 signifying the sodium adduct of the monosubstituted product), m/z 1271.5 for the disubstituted product and m/z 1339.5 for the trisubstituted monomer. There was evidence of further substitution but these peaks were at a very low intensity.

¹H NMR was used to provide an average number of methacryloyl groups per cyclodextrin unit [6, 25]. Figure 5.10 (b) shows that the OH-6 peak at 4.3 ppm had all but disappeared, suggesting that this was the primary site of reaction. The OH-3 and OH-2 peaks changed from sharp peaks at 5.5 ppm to a broad signal, overlapping with the vinyl protons at 5.5-6.0 ppm. The integration for the CDM H-1 proton was normalised for a value of 7 and comparison of the integrals for the suspected vinyl protons gave a combined value of 1.6, indicating an average of 0.5 groups substituted per cyclodextrin molecule. Given the results obtained from LC-MS, this appeared to be

very low, which suggested that the presence of the broad signal overlapping the vinyl protons prevented accurate determination of the average number of groups per molecule.

There was evidence of the presence of vinylic carbons in the 13 C spectrum (Figure 5.6 (c)), at 130 and 150 ppm, suggesting the presence of vinyl groups of different types. A peak at 170 ppm indicated the presence of a carbonyl carbon.

The combination of evidence from MS and NMR suggested the successful derivatisation of β -CD, however, downfield peaks unrelated to the cyclodextrin monomer were observed in the ¹H NMR spectrum for CDM at 8-9 ppm. Analysis of the ¹H spectrum of pyridine suggested that these peaks were due to the presence of pyridine entrapped within the hydrophobic cavity. While the wash step with ethanol was carried out in a bid to reduce this problem, neither this step nor various vacuum drying steps could remove the pyridine. This was considered to be problematic as the pyridine molecules were likely to reside in the hydrophobic cavity intended for interaction with the steroidal template in the pre-polymerisation complex. To combat this, a synthetic strategy using a non-competitive solvent was investigated.

The approach developed by Lee *et al.* [32] and employed by Siemoneit *et al.* [34] for the synthesis of polymerisable β - and γ -cyclodextrin monomers, was studied. This method involved the synthesis of acrylamidomethyl cyclodextrin (β - or γ -CDMA) by preparation of an aqueous cyclodextrin solution followed by the addition of *N*-(hydroxymethyl)acrylamide (NMA). The reaction mixture was then heated to 80 °C and sufficient HCl was added to prepare a 1% (v/v) solution. The synthetic strategy employed is schematically depicted in Figure 5.11.



Figure 5.11: Synthesis of acrylamidomethyl cyclodextrins (CDMA) showing reaction at C-6 OH.

The derivatisation of CDs by this method had two distinct advantages over the preparation of CDM, namely, the short reaction time of five minutes and the use of a solvent that would be unlikely to compete with the steroid templates for interaction with the hydrophobic cavity. Preliminary investigations for β -CD used the 1:10 ratio of cyclodextrin to NMA reported in the literature by Siemoneit *et al.* for the derivatisation of γ -CD [34]. However, LC-MS analysis suggested that very little product was formed. This was possibly due to the difference in solubility between β -CD and γ -CD (the latter being over ten times more soluble in water). To promote interaction the concentration of NMA was increased three-fold and the following mass and NMR spectra (Figure 5.12) demonstrate the success of this reaction.





Figure 5.12: Spectra obtained for β -CDA: (a) MS, (b) ¹H NMR, (c) ¹³C NMR (NMR carried out in DMSO- d_6).

The major peak at m/z 1218.5 in the mass spectrum shown in Figure 5.12 (a) suggests that the monosubstituted compound was the major product as this was the sodium adduct. Evidence for the disubstituted product was also present with a peak at m/z 1323.6. The ¹H NMR spectrum shows evidence of vinyl protons at 5.5-6.3 ppm. Quantitation was again complicated by the presence of the broad overlapping peak in the region of the signals observed for the vinyl protons and a value could not be accurately obtained. The ¹³C NMR spectrum in Figure 5.12 (a) showed evidence of vinylic carbons at 120-130 ppm and a carbonyl group is indicated by the resonance signal observed at 170 ppm. These results again suggested the presence of a mixture of derivatised β -CD compounds and the low intensity of all signals may suggest that the monosubstituted product was the major component.

Due to the modest success of this reaction, this approach was applied to γ -CD for the preparation of γ -CDA. The mass and NMR spectra for the starting material, γ -CD, are presented in Figure 5.13 (¹H NMR peak assignment from reference [34]).





Figure 5.13: Spectra for γ -CD starting material: (a) MS, (b) ¹H NMR and (c) ¹³C NMR (NMR carried out in D₂O).

The preparation of the γ -CD based monomer initially attempted using the 1:10 CD:NMA ratio but as with β -CD, very little product was formed. In this case, the most successful ratio for CD:NMA was found to be 1:50 and the relevant spectra are presented in Figure 5.14.







Figure 5.14: Spectra for γ-CDA: (a) MS, (b) ¹H NMR and (c) ¹³C NMR (NMR carried out in D₂O).

The mass spectrum in Figure 5.14 (a) shows peaks for the monosubstituted γ -CDA (m/z 1380.5), disubstituted γ -CDA (m/z 1463.5) and trisubstituted γ -CDA (m/z 1568), as well as low intensity peaks for further substituted products. Using the H-1 CD proton in Figure 5.14 (b), the ¹H NMR spectrum, to normalise the integration of the three vinyl proton groups observed, the average number of acrylamide groups per CD was calculated to be 2.5. The ¹³C NMR spectrum (Figure 5.14 (c)) contains peaks at 170 ppm and 130 ppm, which are indicative of the presence of carbonyl carbons and vinylic carbons, respectively. These peaks were not present in the starting material, γ -CD, as is shown in Figure 5.13 (c). These results suggested the successful synthesis of derivatised γ -CD composed of a mixture of differently substituted compounds.

Although the purity and yield of the successfully derivatised CD compounds was not fully optimised at this stage, preliminary studies were carried out on the use of these monomers for the preparation of corticosteroid-imprinted polymers.

5.4.1.4 Preliminary studies on β -CDA and γ -CDA based molecularly imprinted polymers

Due to the high cost of γ -CD, the majority of method development was carried out using β -CD based monomers, i.e. β -CDA. This monomer was used in the preparation of cyclodextrin molecularly imprinted polymer 2 (CDMP2). The solubility characteristics of the polymeric components, necessitated the use of dimethyl formamide (DMF) as the solvent as, of a range of solvents tested, this was the only solvent in which the template (hydrocortisone-17-butyrate), the crosslinker (*N*,*N*-methylene bisacrylamide) and β -CDA were soluble. Ethylene glycol dimethacrylate (EGDMA) was tested for use as crosslinker and while this monomer was soluble in a range of solvents, it was not possible to simultaneously dissolve EGDMA and β -CDA in any of the solvents studied. For this reason *N*, *N*-methylene bisacrylamide was investigated for use and while solubility was still an issue to some extent, it was possible to dissolve all of the components simultaneously in DMF. The maximum monomer concentration, however, was still quite low with a limit of 28% (w/v).

CDMP2 was prepared as a bulk monolith and ground and sieved to a range of 23-58 μ m. This polymer fraction was then extracted using Soxhlet apparatus and was packed into HPLC columns as per Section 5.3.8. The polymer was tested and compared to the non-imprinted control polymer (CDNP2) using chromatographic evaluation to investigate the imprinting effect. Acetonitrile and acetonitrile-water compositions were used as the mobile phase in mixtures ranging from 10/90 to 90/10 (v/v). No difference was observed in the retention behaviour on the MIP and NIP column for either the template or any of the six steroidal analogues investigated in this study. A possible reason for this may have been the extremely low surface area of the polymer, which was less than 1 m².g⁻¹. Solvent competition for the CD cavity in the pre-polymerisation complex may also have had a negative effect on imprinting attempts. The lack of any observable imprinting effect, despite analysis in highly aqueous solutions where hydrophobic interactions should predominate, suggested that this polymer was not suitable for further investigations. Rather, other avenues for the exploitation of CD and steroid interaction in the pre-polymerisation solution were pursued.

Since a high monomer percentage is required for bulk monolith preparation, the aforementioned high cost of γ -CD prevented the cost effective synthesis of 'bulk' polymers using γ -CDA. In contrast, the low monomer percentage necessary for the 241
preparation of precipitated polymers facilitated preliminary investigations using γ -CDA as a co-monomer. Since NMR studies outlined in Section 5.4.1.1 showed strong interaction between β -CD and BDN, it was postulated that the interaction between γ -CD and steroidal templates (in this case HCB) would be of the same magnitude. In fact, literature reports have shown the interaction between γ -CD and steroidal analytes to be stronger than that of β -CD [12]. This enhanced interaction should allow for an improved imprinting effect for precipitated polymers over those outlined in Chapters 2 and 3.

To further drive reaction, the use of aqueous solutions for the preparation of such polymers would be expected to be advantageous, due to the hydrophobic nature of the interaction between steroids and CDs. While the inherently high monomer loading of bulk strategies prevented the use of water for this approach, the necessarily high solvent to monomer ratios required for precipitation strategies facilitated investigation of the use of water as the porogen for steroidal polymers. To this end, precipitated polymers using γ -CDA as the functional monomer and aqueous mixtures of acetonitrile as porogen were investigated.

It was decided to prepare precipitated polymers based on M13, as outlined in Chapter 3, which employed methacrylic acid (MAA) as the functional monomer, EGDMA as the crosslinking monomer and acetonitrile as the porogen. The amount of MAA could be reduced as the template could interact with the cyclodextrin monomer as well. The chosen ratio of template (HCB): γ -CDA:MAA:EGDMA was therefore selected as 1:2:4:30. Different ratios of acetonitrile to water were tested empirically to determine the composition in which all components could be dissolved and the highest monomer concentration that could be used. This resulted in the use of an acetonitrile-water 60/40 (v/v) mixture and a monomer concentration of 1% (w/v). This resulted in precipitated microparticles produced in the size range of 0.27 – 7.23 µm, of which an SEM image can be seen in Figure 5.15.



Figure 5.15: SEM image of microparticles formed by precipitation polymerisation using aqueous acetonitrile as the porogen and γ -CDA as the co-monomer.

Time constraints prevented any binding analysis or physical characterisation of this polymer, however, the ability to use CD based functional monomers in a highly aqueous solvent to prepare precipitated microparticles suggests the possibility of preparing molecularly imprinted polymers with an increased capability for selective corticosteroid rebinding.

5.4.2 The synthesis of a polymerisable template-monomer for use in semi-covalent corticosteroid MIPs

As discussed in detail in Section 1.3.3, the semi-covalent method of molecular imprinting attempts to utilise covalent linkages for the creation of binding sites with the homogeneity and fidelity synonymous with covalent imprinting, while achieving the fast rebinding kinetics inherent to non-covalent binding events. In an effort to achieve this for the corticosteroid templates focussed on in this study it was decided to use hydrocortisone as the starting material for derivatisation reactions. Hydrocortisone possesses three hydroxyl groups, at the C-11, C17 and C-21 positions, with the potential for derivatisation and subsequent addition of between one and three polymerisable vinyl groups. The method outline in Section 5.3.3.2 was used to synthesise hydrocortisone methacrylate (HMA) and the reaction scheme is shown in Figure 5.16.



Figure 5.16: Reaction scheme for the synthesis of HMA with all three potential vinyl groups shown.

Time constraints again dictated that the synthesis of this polymerisable templatemonomer was a preliminary feasibility study into the ease of preparation of the monomer and the possibility of its use for the preparation of corticosteroid imprinted polymers by semi-covalent means. As can be seen from the following discussion, LC-MS and NMR analysis were used to obtain evidence for the derivatisation of hydrocortisone and rather than carrying out a full characterisation of the prepared template-monomer, the discussion will centre on comparisons of the NMR spectra of hydrocortisone (HMA) and hydrocortisone methacrylate (HMA). To facilitate this, the mass spectra of both compounds can be seen in Figure 5.17 (a) and (b).





Figure 5.17: Mass spectra for (a) HYD and (b) HMA

The mass spectra shown in Figure 5.17 (a) and (b) suggest the synthesis of monosubstituted HMA based on peaks at m/z 431.6 and 861.5 for the MH⁺ and dimer ions, respectively. The NMR spectra for HYD and HMA is presented in Figure 5.18.





Figure 5.18: Comparative spectra highlighting the differences between HYD and HMA in (a) 1 H NMR and (b) 13 C NMR.

Figure 5.18 (a) demonstrates that while the main steroidal skeleton (chemical shifts between 0-4 ppm) remained unchanged, there was evidence to suggest the presence of a number of different vinyl protons at 5.6-6 ppm. Further evidence of the presence of different vinyl groups was observed on the ¹³C spectra; where extra peaks for HMA at 125-130 ppm suggested the presence of two types of vinyl protons while extra peaks at 167 ppm suggested the presence of two types of carbonyl groups. The evidence observed from MS and NMR analysis suggested that HYD had been successfully derivatised and while MS pointed to the monosubstituted compound, NMR analysis, where evidence of different vinyl protons was present, suggested the possibility of more than one type of monosubstituted compound. No evidence of starting material was observed in the MS spectrum, suggesting that the reaction went to completion. The monomer so prepared was utilised in preliminary investigations into corticosteroid imprinted polymers prepared by a semi-covalent strategy.

5.4.2.1 Towards the semi-covalent imprinting of corticosteroids

The semi-covalent approach to the molecular imprinting of corticosteroids was investigated using the polymerisable template-monomer, HMA, discussed in the

preceding section. The bulk imprinted polymer, hydrocortisone methacrylate molecularly imprinted polymer, HMAMP1, and non-imprinted polymer, HMANP1, were prepared (Section 5.3.4.2).

Prior to removal of the template-monomer by basic hydrolysis, the polymers were packed into stainless steel columns and the retention behaviours of the imprinted and control polymers for hydrocortisone were chromatographically evaluated. This involved using 100% acetonitrile as the mobile phase. Almost identical retention times of 2.3 minutes were obtained for both polymers.

The template-monomer was then removed by refluxing the polymer with 1 M KOH prepared in MeOH in a manner analogous to that used by Cacho *et al.* [28]. After washing with dilute HCl to neutralise the functional groups on the polymer, the ground and sieved particles were packed into stainless steel columns. The retention behaviours for polymers were drastically different at this stage, with retention times of 4.5 minutes obtained for both polymers. While no imprinting effect was observed, i.e. no difference between HMAMP1 and HMANP1, the basic treatment of the polymer appeared to have dramatically enhanced the affinity of the polymer for the steroidal template.

Physical characterisation by nitrogen sorption porosimetry showed that there was no significant differences between surface area (570 m².g⁻¹ ±10), pore volume (0.80 cm³.g⁻¹ ± 0.1) or pore diameter (8.5 nm ± 1.0) for the imprinted polymers before and after hydrolysis. No morphological differences were observed upon hydrolysis for the control polymer, though they had a slightly larger surface area at 620 m².g⁻¹. Given that differences in retention were not due to physical characteristics, it was postulated that it was due to chemical changes in the polymer upon hydrolysis.

This behaviour was also observed by Cheong *et al.* [38], in a study carried out for testosterone imprinted polymers. In that investigation an imprinting strategy was referred to, where testosterone methacrylate was prepared and used in the preparation of EGDMA based polymers. Removal of this template-monomer was carried out by reflux in methanolic NaOH. Cheong *et al.* observed no imprinting effect but did see drastically increased retention behaviour on both imprinted and control polymers. They attributed this increased retention to the presence of sodium carboxylate functionalities that interacted more strongly with the steroidal analytes. This may have been the case in the

present study, with potassium carboxylate functionalities being responsible for the increased affinity.

It is worth noting that Cheong *et al.* point to the work of Bystrom *et al.* [39], where it was observed that NaOH/MeOH and NaOMe/MeOH treatment of the polymer had no affect on the ester groups of the steroid-methacrylate template-monomers. Instead, Bystrom *et al.* required the use of LiAlH₄/THF to successfully remove the template-monomer. In the present study, no effort was made to quantify the amount of hydrocortisone removed from the imprinted polymer and these literature examples would suggest that the use of a stronger hydrolysis agent would be required for removal of the template.

The potential to use the prepared template-monomer, HMA, for precipitated microspheres was also demonstrated in the present study. The incorporation of the template-monomer complex into polymers prepared in dilute solution could increase the number of binding sites in the resultant polymer, despite being prepared in dilute solutions. This polymer, HMAMP2, was prepared as outlined in Section 5.3.4.2 and resulted in the synthesis of discrete microspherical particles in the 1.2-5.1 μ m size range. Again, time constraints prevented any analysis of these polymeric microspheres.

5.5 Conclusion

The primary focus of this study was to synthesise tailor-made functional monomers capable of increasing the number of corticosteroid selective binding sites within polymeric matrices by enhancing complex formation in the pre-polymerisation solution or incorporating polymerisable template-monomers into the growing polymer network. Regarding the former, enhanced interactions between β -CD and BDN were observed during NMR binding studies.

This chapter demonstrated that cyclodextrin compounds were successfully derivatised using two cyclodextrin compounds, β -CD and γ -CD. Two methods were investigated and it was demonstrated that the most successful strategy resulted in the addition of an average of 2.5 polymerisable acrylamide units per γ -CD molecule. The monomers prepared showed the capability of being incorporated into corticosteroid imprinted

polymers. While time constraints prevented a full investigation into the ability of these polymers to specifically bind corticosteroid targets, precipitated microparticles incorporating γ -CD based monomers were prepared in highly aqueous porogens, using HCB as a template. Bulk polymers were also prepared using β -CD based monomers, showing the versatility of these compounds for use in a range of polymer formats and solvents.

This study also demonstrated the successful synthesis of a polymerisable templatemonomer based on hydrocortisone. This template-monomer, hydrocortisone methacrylate, was prepared in what evidence suggested was a mixture of monosubstituted compounds with different positions of derivatisation. This polymerisable template was used in the preparation of bulk polymers and perhaps more interestingly, novel precipitated microspheres. This again showed the potential for the use of successfully derivatised compounds for a range of polymer formats.

Although it was not possible to fully investigate the polymers prepared from the synthesised functional monomers in either case, and so fully exploit the potential of these monomers, it was shown that polymerisable functional monomers with the potential for dramatically increasing the number of bindings sites within polymers for poorly functionalised templates were prepared.

References

- 1. Mayes AG, Whitcombe MJ: Synthetic strategies for the generation of molecularly imprinted organic polymers. Advanced Drug Delivery Reviews 2005, 57:1742-1778.
- 2. Chen L, Xu S, Li J: Recent advances in molecular imprinting technology: current status, challenges and highlighted applications. *Chemical Society Reviews* 2011, **40**:2922-2942.
- 3. Barahona F, Turiel E, Cormack PAG, Martin-Esteban A: Synthesis of coreshell molecularly imprinted polymer microspheres by precipitation polymerization for the inline molecularly imprinted solid-phase extraction of thiabendazole from citrus fruits and orange juice samples. *Journal of Separation Science* 2011, 34:217-224.
- 4. Zhou W-H, Guo X-C, Zhao H-Q, Wu S-X, Yang H-H, Wang X-R: Molecularly imprinted polymer for selective extraction of domoic acid from seafood coupled with high-performance liquid chromatographic determination. *Talanta* 2011, **84**:777-782.
- 5. Shi XZ, Song SQ, Qu GR, Zheng SL, Wu AB, Zhang DB: Water compatible molecularly imprinted polymer microspheres for extraction of ampicillin in foods. *Analytical Letters* 2010, **43**:757-767.
- 6. Piletsky SA, Andersson HS, Nicholls IA: Combined Hydrophobic and Electrostatic Interaction-Based Recognition in Molecularly Imprinted Polymers. *Macromolecules* 1999, **32:**633.
- Hoshino Y, Koide H, Urakami T, Kanazawa H, Kodama T, Oku N, Shea KJ: Recognition, Neutralization, and Clearance of Target Peptides in the Bloodstream of Living Mice by Molecularly Imprinted Polymer Nanoparticles: A Plastic Antibody. Journal Of the American Chemical Society 2010, 132:6644-+.
- 8. Hoshino Y, Shea KJ: **The evolution of plastic antibodies.** *Journal of Materials Chemistry* 2011, **21:**3517-3521.
- 9. Schneider H-J, Hacket F, Rudiger V, Ikeda H: **NMR Studies of Cyclodextrins** and Cyclodextrin Complexes. *Chemical Reviews* 1998, **98**:1755.
- 10. van de Manakker F, Vermonden T, van Nostrum CF, Hennink WE: Cyclodextrin-Based Polymeric Materials: Synthesis, Properties, and Pharmaceutical/Biomedical Applications. *Biomacromolecules* 2009, **10**:3157.
- 11. Tan ZJ, Zhu XX, Brown GR: Formation Of Inclusion Complexes Of Cyclodextrins With Bile-Salt Anions As Determined By Nmr Titration Studies. *Langmuir* 1994, **10**:1034-1039.
- 12. Cameron KS, Fletcher D, Fielding L: An NMR study of cyclodextrin complexes of the steroidal neuromuscular blocker drug Rocuronium Bromide. *Magnetic Resonance In Chemistry* 2002, **40**:251-260.
- Jansook P, Kurkov SV, Loftsson T: Cyclodextrins as solubilizers: Formation of complex aggregates. vol. 99. pp. 719-729: Wiley Subscription Services, Inc., A Wiley Company:719-729.
- 14. Roche PJR, Ng SM, Narayanaswamy R, Goddard N, Page KM: Multiple surface plasmon resonance quantification of dextromethorphan using a molecularly imprinted [beta]-cyclodextrin polymer: A potential probe for drug-drug interactions. *Sensors and Actuators B: Chemical* 2009, **139**:22.
- 15. Xu ZF, Kuang DZ, Feng YL, Zhang FX: Combination of hydrophobic effect and electrostatic interaction in imprinting for achieving efficient recognition in aqueous media. *Carbohydrate Polymers* 2010, **79:**642-647.

- 16. Tang SW, Kong L, Ou JJ, Liu YQ, Li X, Zou HF: **Application of cross-linked beta-cyclodextrin polymer for adsorption of aromatic amino acids.** *Journal Of Molecular Recognition* 2006, **19:3**9-48.
- 17. Larsen KL, Aachmann FL, Wimmer R, Stella VJ, Kjølner UM: **Phase solubility** and structure of the inclusion complexes of prednisolone and 6α-methyl prednisolone with various cyclodextrins. *Journal of Pharmaceutical Sciences* 2005, 94:507-515.
- 18. Zarzycki PK, Ohta H, Saito Y, Jinno K: Interaction of native alphacyclodextrin, beta-cyclodextrin and gamma-cyclodextrin and their hydroxypropyl derivatives with selected organic low molecular mass compounds at elevated and subambient temperature under RP-HPLC conditions. Analytical And Bioanalytical Chemistry 2008, **391**:2793-2801.
- 19. Forgo P, Vincze I, Kövér KE: Inclusion complexes of ketosteroids with [beta]-cyclodextrin. *Steroids* 2003, 68:321.
- 20. Forgo P, Gondos G: A study of beta-cyclodextrin inclusion complexes with progesterone and hydrocortisone using rotating frame Overhauser spectroscopy. *Monatshefte Fur Chemie* 2002, **133**:101-106.
- 21. Asanuma H, Kakazu M, Shibata M, Hishiya T, Komiyama M: Molecularly imprinted polymer of beta-cyclodextrin for the efficient recognition of cholesterol. *Chemical Communications* 1997:1971-1972.
- 22. Hishiya T, Asanuma H, Komiyama M: Molecularly imprinted cyclodextrin polymers as stationary phases of high performance liquid chromatography. *Polymer Journal* 2003, **35:**440-445.
- 23. Asanuma H, Akiyama T, Kajiya K, Hishiya T, Komiyama M: Molecular imprinting of cyclodextrin in water for the recognition of nanometer-scaled guests. *Analytica Chimica Acta* 2001, **435**:25.
- 24. Hishiya T, Akiyama T, Asanuma H, Komiyama M: Molecular imprinting of cyclodextrins leading to synthetic antibodies. *Journal Of Inclusion Phenomena And Macrocyclic Chemistry* 2002, **44**:365-367.
- 25. Piletsky SA, Andersson HS, Nicholls IA: The rational use of hydrophobic effect-based recognition in molecularly imprinted polymers. *Journal Of Molecular Recognition* 1998, **11**:94-97.
- 26. Liu H, Liu C, Yang X, Zeng S, Xiong Y, Xu W: Solid-phase extraction of ursolic acid from herb using β-cyclodextrin-based molecularly imprinted microspheres. *Journal of Separation Science* 2008, **31:**3573-3580.
- 27. Sreenivasan K: Synthesis and evaluation of a beta cyclodextrin-based molecularly imprinted copolymer. *Journal Of Applied Polymer Science* 1998, **70:**15-18.
- 28. Cacho C, Turiel E, Martín-Esteban A, Ayala D, Pérez-Conde C: Semi-covalent imprinted polymer using propazine methacrylate as template molecule for the clean-up of triazines in soil and vegetable samples. *Journal of Chromatography A* 2006, 1114:255.
- 29. Curcio P, Zandanel C, Wagner A, Mioskowski C, Baati R: Semi-Covalent Surface Molecular Imprinting of Polymers by One-Stage Mini-emulsion Polymerization: Glucopyranoside as a Model Analyte. *Macromolecular Bioscience* 2009, 9:596-604.
- 30. Alexiadou DK, Maragou NC, Thomaidis NS, Theodoridis GA, Koupparis MA: Molecularly imprinted polymers for bisphenol A for HPLC and SPE from water and milk. *Journal Of Separation Science* 2008, **31**:2272-2282.
- 31. Xu Z, Kuang D, Liu L, Deng Q: Selective adsorption of norfloxacin in aqueous media by an imprinted polymer based on hydrophobic and

electrostatic interactions. *Journal of Pharmaceutical and Biomedical Analysis* 2007, **45:**54.

- 32. Myung Hak L, Kee Jong Y, Sohk-Won K: Synthesis of a vinyl monomer containing beta-cyclodextrin and grafting onto cotton fiber. *Journal of Applied Polymer Science* 2001, **80:**438-446.
- 33. Hishiya T, Shibata M, Kakazu M, Asanuma H, Komiyama M: Molecularly imprinted cyclodextrins as selective receptors for steroids. *Macromolecules* 1999, **32:**2265-2269.
- 34. Siemoneit U, Schmitt C, Alvarez-Lorenzo C, Luzardo A, Otero-Espinar F, Concheiro A, Blanco-Méndez J: Acrylic/cyclodextrin hydrogels with enhanced drug loading and sustained release capability. *International Journal of Pharmaceutics* 2006, 312:66.
- 35. Fielding L: Determination of Association Constants (Ka) from Solution NMR Data. *Tetrahedron* 2000, 56:6151.
- 36. Asanuma H, Kakazu M, Shibata M, Hishiya T, Komiyama M: Synthesis of molecularly imprinted polymer of [beta]-cyclodextrin for the efficient recognition of cholesterol. *Supramolecular Science* 1998, **5**:417.
- 37. Hishiya T, Asanuma H, Komiyama M: Spectroscopic Anatomy of Molecular-Imprinting of Cyclodextrin. Evidence for Preferential Formation of Ordered Cyclodextrin Assemblies. Journal of the American Chemical Society 2002, 124:570.
- Cheong SH, McNiven S, Rachkov A, Levi R, Yano K, Karube I: Testosterone Receptor Binding Mimic Constructed Using Molecular Imprinting. Macromolecules 1997, 30:1317-1322.
- 39. Bystrom SE, Borje A, Akemark A: Selective reduction of steroid 3-ketones and 17-ketones using LiAlH₄ activated template polymers *J Am Chem Soc* 1993, 115:2081-2083.

Chapter 6

Conclusions and future work

6. Conclusions and Future work

Molecularly imprinted polymers with corticosteroid recognition capabilities have been prepared in this study. These polymers have been synthesised in a range of polymer formats and while some of the polymers have demonstrated group selectivity towards six steroidal analytes, others have demonstrated the ability to differentiate between closely related structural analogues.

Novel corticosteroid-imprinted polymers have been prepared by precipitation methods, with both microspherical and microparticular morphologies. The development of these precipitated polymers resulted in the compilation of detailed knowledge for the preparation of such polymers and the factors affecting polymerisation. It was shown that monomer concentration, monomer ratios, initiator type and concentration as well as temperature and initiation methods all played a role in the ability to control the resultant polymers. Polymers were prepared with a high degree of control over particle size and morphology, using three crosslinking monomers, two templates and a range of solvent compositions. The control of particle size and morphology is demonstrated by the microspherical EGDMA and DVB polymers shown in Figure 6.1.



Figure 6.1: Examples of the novel microspherical corticosteroid-imprinted polymers prepared in this study.

The novel microspheres and microparticles, while exhibiting both group and specific target selectivity, demonstrated that these polymers, prepared by precipitation strategies, had a relatively low number of specific binding sites. A range of NMR studies, including titration based binding studies and Job plot analyses, were successfully carried out. These studies revealed weak interactions between two of the templates, budesonide (BDN) and hydrocortisone-17-butyrate (HCB), and commercially available functional monomers or monomer analogues. This led to a rational understanding of the effects of the combination of these weak interactions and the high dilution conditions required for precipitation strategies.

This knowledge was applied to the preparation of bulk polymers in an effort to enhance the interaction between templates and functional monomers in the pre-polymerisation complex by the inherent increase in monomer loading for such strategies. The target analogue approach was employed, using HCB as the template for polymers synthesised to have a high degree of affinity for BDN. A range of bulk polymers were prepared using three crosslinking monomers, DVB, EGDMA and TRIM, and a number of different porogen compositions. Evaluation of these polymers displayed polymers with a number of different characteristics and showed that the crosslinking monomer had profound effects on the molecular recognition properties of the polymers, as well as impacting on the driving forces behind recognition.

These polymers were investigated for use in solid phase extraction applications and HCB imprinted polymers were shown to specifically recognise and retain the target, BDN, using a number of different protocols. Precipitated microparticles, using prednisolone (PRD) as the template, were shown to be capable of the chromatographic separation of PRD and the closely related, hydrocortisone, as shown in Figure 6.2.



Figure 6.2: Chromatographic separation using precipitated polymers to separate prednisolone and hydrocortisone

Separations of these compounds on 5 cm columns achieved α values ranging from 1.3-1.4 at different temperatures and using optimised polymers. Information obtained on the factors affecting precipitation polymerisation allowed for tailoring of particle morphology to achieve analyte resolution of 0.75. Future work for this polymer will focus on increased control over particle morphology and the use of chromatographic equipment suited to smaller particles and higher pressures, such as ultra pressure liquid chromatography (UPLC). In such instrumentation the high pressures needed for small particles can be achieved. Future investigations will also include studies on the effect of an increased amount of template in the porogen, to increase the number of selective binding sites or the use of a functional monomer capable of increasing the number of sites within the polymer.

These polymers suggested the potential for cheap and efficient bulk preparation of selective materials, capable of differentiating between close structural analogues. While the focus for MIP application has often been on their use for analytical separations such as in chromatography, realisation of their commercialisation has been slow. Some of the reasons for this, such as peak tailing and band broadening have been highlighted in this study, and while approaches such as the use of polymers with more defined and controllable morphology can be applied, they too have their limitations. Consideration must be given to where the focus of commercialisation hopes for MIPs should lie. They have definite and proven strengths, such as the cheap and inexpensive production of highly selective materials, which can be prepared for myriad targets. Solid-phase

extraction is an application that exploits such strengths and has achieved commercial success. This success could be built upon by the use of MIPs in process applications where they are employed in bulk for the selective purification of a product or process intermediate. This can be achieved by retention of either the product or retention of impurities that need to be removed from the process. The use of MIPs in membranes for such applications is a growing field and the ability to create strong, specific interaction for targets of all types is essential for this area to realise its full potential.

The need for strong, specific interaction was also highlighted during the course of this study, where the weak interaction between widely used functional monomers and the steroidal templates was seen to have a profound effect on the binding characteristics of the resultant polymers. This further demonstrated the need for tailor-made functional monomers with the ability to increase the number of selective binding sites within the polymeric matrix. To this end a number of tailor-made functional monomers were successfully synthesised using a range of approaches. One strategy involved the preparation of derivatised cyclodextrin (CD) compounds. In this study, CDs were shown to interact to a strong degree with BDN in polar solvents. Native β - and γ -CD were then successfully derivatised to form polymerisable functional monomers containing, for the most successful reaction, an average of 2.5 polymerisable vinyl groups per γ -CD. The successfully vinylated CDs were shown to have the capability to be incorporated into polymers prepared in either bulk or precipitation strategies in both organic and aqueous-organic porogens.

While time constraints prevented the full development of these polymers, the capability to prepare polymers using precipitation strategies in aqueous-based porogens has potential for the preparation of imprinted polymers with enhanced selectivity for poorly functionalised corticosteroids. Future studies for this strategy involve optimisation of the polymer morphology and a detailed binding study, evaluating the imprinting power for such polymers and corticosteroids.

The semi-covalent approach was investigated in this study and a polymerisable steroidal template-monomer was successfully prepared by derivatisation of hydrocortisone. The use of derived template-monomers negates the need for strong interaction in the pre-polymerisation stage and incorporates the template into the growing polymer by way of direct polymerisation. Hydrocortisone methacrylate was successfully prepared in a

mixture of monosubstituted compounds and this monomer was shown to be capable of incorporation into a precipitated microsphere with highly controllable morphology. Time constraints prevented the full development of this strategy and future work would involve a detailed investigation into the removal of the template from the polymer. This would allow for a detailed binding study on the polymer, allowing for quantitation of the effect on the strength and quantity of the binding sites present within the polymeric matrix. The potential to increase the number of binding sites on a polymer, for which the morphological properties can be accurately and readily adjusted, suggests the potential for use of such polymeric strategies in a wide range of applications.

This study has highlighted the urgent need in the field of molecular imprinting to focus on the development of tailor-made and custom monomers for use with challenging templates like corticosteroids. The ability to design and synthesis monomers capable of exploiting all of the possible molecular interactions, as well as having the potential for use in a range of highly controllable polymer formats, will widen the applicability of molecular imprinting, not just in terms of target choice but also in the realisation of their full commercial potential.